# Substitutions, Rauzy fractals and tilings

5

Valérie Berthé,

Anne Siegel,

Jörg Thuswaldner

## 5.1 Introduction

This chapter focuses on multiple tilings associated with substitutive dynamical systems. We recall that a substitutive dynamical system  $(X_{\sigma}, S)$ is a symbolic dynamical system where the shift S acts on the set  $X_{\sigma}$  of infinite words having the same language as a given infinite word which is generated by powers of a primitive substitution  $\sigma$ . We restrict to the case where the inflation factor of the substitution  $\sigma$  is a unit Pisot number. With such a substitution  $\sigma$ , we associate a multiple tiling composed of tiles which are given by the unique solution of a set equation expressed in terms of a graph associated with the substitution  $\sigma$ : these tiles are attractors of a graph-directed iterated function system (GIFS). They live in  $\mathbb{R}^{n-1}$ , where nstands for the cardinality of the alphabet of the substitution. Each of these tiles is compact, it is the closure of its interior, it has non-zero measure and it has a fractal boundary that is also an attractor of a GIFS. These tiles are called *central tiles* or *Rauzy fractals*, according to G. Rauzy who introduced them in (Rauzy 1982).

Central tiles were first introduced in (Rauzy 1982) for the case of the Tribonacci substitution  $(1 \mapsto 12, 2 \mapsto 13, 3 \mapsto 1)$ , and then in (Thurston 1989) for the case of the beta-numeration associated with the Tribonacci number (which is the positive root of  $X^3 - X^2 - X - 1$ ). One motivation for Rauzy's construction was to exhibit explicit factors of the substitutive dynamical system  $(X_{\sigma}, S)$  as translations on compact abelian groups, under the hypothesis that  $\sigma$  is a Pisot substitution.

By extending the seminal construction in (Rauzy 1982), it has been proved that central tiles can be associated with Pisot substitutions (see for instance (Arnoux and Ito 2001) or (Canterini and Siegel 2001b)) as well as

Combinatorics, Automata and Number Theory, ed. Valérie Berthé and Michel Rigo. Published by Cambridge University Press. ©Cambridge University Press 2010.

with beta-numeration with respect to Pisot numbers (cf. (Thurston 1989), (Akiyama 1999) and (Akiyama 2002)). They are conjectured to induce tilings in all these cases. The tiling property is known to be equivalent to the fact that the dynamical system  $(X_{\sigma}, S)$  has pure discrete spectrum (see (Pytheas Fogg 2002, Chapter 7) and (Barge and Kwapisz 2006)) when  $\sigma$  is a unit Pisot irreducible substitution.

We have chosen here to concentrate on tilings associated with substitutions for the sake of clarity. A similar study can be performed in the framework of beta-numeration, with both viewpoints being intimately connected through the notion of beta-substitution. Indeed, a beta-substitution can be associated with any Parry number  $\beta$  (for more details, see Exercise 5.1 and Section 5.11). In the case where  $\beta$  is a Pisot number, the associated substitution can be Pisot reducible as well as Pisot irreducible. The exposition of the theory of central tiles is much simpler when  $\sigma$  is assumed to be Pisot irreducible, even if it extends to the Pisot reducible case. Hence, we will restrict ourselves to the Pisot irreducible case.

There are several approaches for the definition of central tiles. We detail below a construction for unit Pisot substitutions based on a broken line which is defined in terms of the abelianisation of an infinite word generated by  $\sigma$ . Projecting the vertices of this broken line to the contractive subspace of the incidence matrix of  $\sigma$  along its expanding direction and taking the closure of this set yields the central tile. For more details on different approaches, see the surveys in (Pytheas Fogg 2002, Chapters 7 and 8) and (Berthé and Siegel 2005), as well as the discussion in (Barge and Kwapisz 2006) and (Ito and Rao 2006).

The aim of this chapter is to list a great variety of tiling conditions, by focusing on effectivity issues. These conditions rely on the use of various graphs associated with the substitution  $\sigma$ .

This chapter is organised as follows. Section 5.2 gathers all the introductory material. We assume that we are given a unit Pisot irreducible substitution  $\sigma$ . A suitable decomposition of the space  $\mathbb{R}^{n-1}$  is first introduced in Section 5.2.1 with respect to the eigenspaces of the incidence matrix  $\mathbf{M}_{\sigma}$ of  $\sigma$ . A definition of the central tile associated with  $\sigma$  as well as its decomposition into subtiles is then provided in Section 5.2.2. We discuss the graph-directed set equation satisfied by the subtiles in Section 5.2.3. Two (multiple) tilings associated with  $\sigma$  are then introduced in Section 5.3. The first one, introduced in Section 5.3.2, is called tiling of the expanding line. This tiling by intervals tiles the expanding line of the incidence matrix  $\mathbf{M}_{\sigma}$ of  $\sigma$ . The second one is *a priori* not a tiling, but a multiple tiling. It is defined on the contracting space of the incidence matrix  $\mathbf{M}_{\sigma}$ , and it is made of translated copies of the subtiles of the central tile. It is called the self-replicating multiple tiling. Note that it is conjectured to be a tiling. It will be the main objective of the present chapter to introduce various graphs that provide conditions for this multiple tiling to be a tiling.

The first series of tiling conditions is expressed in geometric terms directly related to properties of the self-replicating multiple tiling. We start in Section 5.4.1 with a sufficient tiling property inspired by the so-called finiteness property (F) (discussed in Section 2.3.2.2). This leads us to introduce successively several graphs in Section 5.4 and Section 5.5, yielding necessary and sufficient conditions. We then discuss in Section 5.6, 5.7 and 5.8 further formulations for the tiling property expressed in terms of the tiling of the expanding line. They can be considered as dual to the former set of conditions. In particular, a formulation in terms of the so-called *overlap coincidence condition* is provided in Section 5.7, as well as, in Section 5.8, a further effective condition based on the notion of *balanced pairs*.

#### 5.2 Basic definitions

We use the terminology of Section 1.4. Let  $\sigma$  be a substitution over the alphabet  $A = \{1, 2, ..., n\}$ . In all that follows  $\sigma$  is assumed to be a unit substitution that is Pisot irreducible. In particular,  $\sigma$  is primitive by Theorem 1.4.9. Let us recall that a primitive substitution always admits a power that is prolongable (see Definition 1.2.18 and (Queffélec 1987, Proposition V.1)), and which thus generates an infinite word. For the sake of simplicity, we assume that  $\sigma$  generates an infinite word according to Definition 1.2.18, that will be denoted as  $u = u_0 u_1 \cdots$ . We will see later that this causes no loss of generality (see Theorem 5.3.16 and Remark 5.3.17). Let us note that u is uniformly recurrent by Proposition 1.4.6, and that  $\sigma(u) = u$ , *i.e.*, u is a fixed point of  $\sigma$ .

## 5.2.1 Space decomposition

We want to give a geometric interpretation of the fixed point  $u = u_0 u_1 \cdots$ of the unit Pisot irreducible substitution  $\sigma$ . In the present section we first introduce some algebraic formalism in order to embed u in a subspace of  $\mathbb{R}^n$  spanned by the eigenvectors associated with the algebraic conjugates of the Perron–Frobenius eigenvalue of the incidence matrix of  $\sigma$  (see Theorem 1.4.2). Since  $\sigma$  is Pisot irreducible, this subspace turns out to be a hyperplane. We define a suitable projection of  $\mathbb{R}^n$  onto this hyperplane. The closure of the projections of the abelianised subwords  $\mathbf{P}(u_0 u_1 \cdots u_{N-1})$ , for  $N \in \mathbb{N}$ , will comprise the so-called *central tile* or *Rauzy fractal* that will be defined in Section 5.2.2.

**Eigenvectors and eigenvalues.** Let  $\sigma$  be a unit Pisot irreducible substitution. We want to decompose  $\mathbb{R}^n$  with respect to certain eigenspaces of the incidence matrix  $\mathbf{M}_{\sigma}$  of  $\sigma$ . Let  $\beta$  be the Perron–Frobenius eigenvalue of  $\mathbf{M}_{\sigma}$ . According to our assumptions  $\beta$  is a Pisot unit and n is the algebraic degree of  $\beta$ .

Let r-1 be the number of real conjugates of  $\beta$  (distinct from  $\beta$ ). They are denoted by  $\beta^{(2)}, \ldots, \beta^{(r)}$ . Each corresponding eigenspace has dimension one according to Perron–Frobenius' theorem (Theorem 1.4.2). Let 2s be the number of complex conjugates of  $\beta$ . They are denoted by  $\beta^{(r+1)}, \overline{\beta^{(r+1)}}, \ldots, \beta^{(r+s)}, \overline{\beta^{(r+s)}}$ . Each pair of a complex eigenvector together with its complex conjugate generates a two-dimensional plane. One has n = r + 2ssince  $\sigma$  is Pisot irreducible.

Let  $\mathbf{v}_{\beta}$  be a left eigenvector of  $\mathbf{M}_{\sigma}$  (*i.e.*, an eigenvector of  ${}^{t}\mathbf{M}_{\sigma}$ ) associated with the eigenvalue  $\beta$  having positive entries contained in  $\mathbb{Z}[\beta]$ . Such a vector exists by Perron–Frobenius' theorem. Let  $\mathbf{u}_{\beta}$  be the right eigenvector of  $\mathbf{M}_{\sigma}$  associated with  $\beta$ , and normalised by  $\langle \mathbf{v}_{\beta}, \mathbf{u}_{\beta} \rangle = 1$ . The eigenvector  $\mathbf{u}_{\beta}$  is well defined by the above conditions once  $\mathbf{v}_{\beta}$  is given. Again by Perron–Frobenius' theorem,  $\mathbf{u}_{\beta}$  has positive coordinates in  $\mathbb{Q}(\beta)$ . We obtain left eigenvectors  $\mathbf{v}_{\beta^{(i)}}$  for the algebraic conjugates  $\beta^{(i)}$  of  $\beta$  by replacing  $\beta$ by  $\beta^{(i)}$  in the coordinates of the vector  $\mathbf{v}_{\beta}$ . We similarly obtain the right eigenvectors  $\mathbf{u}_{\beta^{(i)}}$ . Furthermore, the coordinates of  $\mathbf{v}_{\beta}$  are easily seen to be linearly independent over  $\mathbb{Q}$ . The same holds for the coordinates of  $\mathbf{u}_{\beta}$ .

**Remark 5.2.1** Note that this normalisation convention for  $\mathbf{u}_{\beta}$  a priori does not correspond to the normalised Perron–Frobenius eigenvector of Theorem 1.4.5 and Proposition 10.4.2 whose coordinates give the frequencies of letters in u (in this latter case, the sum of coordinates equals 1). See also the discussion in Section 5.11.

The right and left eigenvectors are easily seen to satisfy the following relations, for  $k \ge 2, i \ge 2, k \ne i$ 

$$\langle \mathbf{v}_{\beta}, \mathbf{u}_{\beta^{(k)}} \rangle = 0, \ \langle \mathbf{v}_{\beta^{(i)}}, \mathbf{u}_{\beta^{(k)}} \rangle = 0, \ \langle \mathbf{v}_{\beta^{(k)}}, \mathbf{u}_{\beta^{(k)}} \rangle = 1.$$
 (5.1)

For more details see (Canterini and Siegel 2001b, Section 2), (Ei, Ito, and Rao 2006, Lemma 2.5), (Baker, Barge, and Kwapisz 2006) or (Siegel and Thuswaldner 2010).

A suitable decomposition of the space. Using the eigenvectors defined above we can decompose  $\mathbb{R}^n$  as follows. The *contracting space* of the matrix

 $\mathbf{M}_{\sigma}$  is the subspace  $\mathbb{H}_c$  generated by the eigenvectors  $\mathbf{u}_{\beta^{(i)}}$  associated with the n-1 conjugates of  $\beta$  (each of which has modulus less than one). The *expanding line* of  $\mathbf{M}_{\sigma}$  is the real line  $\mathbb{H}_e$  generated by the eigenvector  $\mathbf{u}_{\beta}$ . Note that the subscripts c and e stand here as abbreviations for *contracting* and *expanding*, respectively. The space  $\mathbb{H}_c$  has dimension r + 2s - 1 = n - 1so that  $\mathbb{H}_c \simeq \mathbb{R}^{n-1}$ . Moreover,  $\mathbb{H}_c$  is orthogonal to  $\mathbf{v}_{\beta}$ , according to (5.1).

We denote by  $h_{\sigma}$ :  $\mathbb{H}_{c} \to \mathbb{H}_{c}$  the restriction of  $\mathbf{M}_{\sigma}$  to  $\mathbb{H}_{c}$ . The mapping  $h_{\sigma}$  is a uniform contraction whose eigenvalues are the conjugates of  $\beta$ . Note that it scales down the (n-1)-dimensional Lebesgue measure by the factor  $|\beta^{(2)}\cdots\beta^{(r)}| |\beta^{(r+1)}|^2 \cdots |\beta^{(r+s)}|^2 = 1/\beta$ , since  $\beta$  is a unit. This contraction mapping will play a prominent role in the sequel.

In order to make the distinction between elements of the *n*-dimensional space  $\mathbb{R}^n$  and elements of the (n-1)-dimensional space  $\mathbb{H}_c$ , we restrict the use of bold symbols for the vectors and linear mappings of  $\mathbb{R}^n$ .

We denote by  $\mu_k$  the k-dimensional Lebesgue measure. In particular, we work with  $\mu_{n-1}$  on  $\mathbb{H}_c$ , and with  $\mu_1$  on  $\mathbb{H}_e$ .

**Projections on the eigenspaces.** Let  $\pi_c : \mathbb{R}^n \to \mathbb{H}_c$  be the projection of  $\mathbb{R}^n$  onto  $\mathbb{H}_c$  along  $\mathbb{H}_e$ , according to the natural decomposition  $\mathbb{R}^n = \mathbb{H}_c \oplus \mathbb{H}_e$ . We recall that  $\mathbf{P}$  denotes the abelianisation mapping defined in Section 1.4. The relation  $\mathbf{P}(\sigma(w)) = \mathbf{M}_{\sigma}\mathbf{P}(w)$  for all  $w \in A^*$  implies the commutation relation

$$\forall w \in A^*, \, \pi_c \circ \mathbf{P} \circ \sigma(w) = h_\sigma \circ \pi_c \circ \mathbf{P}(w).$$
(5.2)

Relation (5.2) reads as follows: when applying  $\sigma$  to a word w, the abelianisation  $\mathbf{P}(w)$  is mapped onto  $\mathbf{M}_{\sigma}\mathbf{P}(w)$ , which has a priori larger entries since  $\mathbf{M}_{\sigma}$  has non-negative entries, and thus 'moves away' from the origin. However, when considering the projection on  $\mathbb{H}_c$  of the abelianisations, the point  $\pi_c \circ \mathbf{P}(w)$ , which is mapped to the point  $h_\sigma \circ \pi_c \circ \mathbf{P}(w)$  when applying  $\sigma$ , gets closer to the origin since  $h_\sigma$  is a uniform contraction. The relation (5.2) will play a key role in the sequel.

We deduce from (5.1) that any element  $\mathbf{x} \in \mathbb{R}^d$  admits the decomposition

$$\mathbf{x} = \langle \mathbf{x}, \mathbf{v}_{\beta} \rangle \, \mathbf{u}_{\beta} + \sum_{i=2}^{r+2s} \langle \mathbf{x}, \mathbf{v}_{\beta^{(i)}} \rangle \mathbf{u}_{\beta^{(i)}}.$$
(5.3)

For more details, see (Canterini and Siegel 2001b, Section 2.1). This implies that the projection of  $\mathbf{x}$  onto  $\mathbb{H}_e$  along  $\mathbb{H}_c$  is equal to  $\langle \mathbf{x}, \mathbf{v}_\beta \rangle \mathbf{u}_\beta$ . We thus define

$$\pi_e \colon \mathbb{R}^n \to \mathbb{R}, \ \mathbf{x} \mapsto \langle \mathbf{x}, \mathbf{v}_\beta \rangle.$$
(5.4)

The mapping  $\pi_e$  is the projection of  $\mathbb{R}^n$  onto the expanding line along the

V. Berthé, A. Siegel, J. Thuswaldner

contracting space  $\mathbb{H}_c$  followed by a suitable renormalisation that makes it into a mapping with values in  $\mathbb{R}$  and not in  $\mathbb{H}_e$ . One has  $\pi_e(\mathbb{H}_c) = 0$ . The mapping  $\pi_e$  measures in some sense the distance to the hyperplane  $\mathbb{H}_c$ . We thus define the *height* of a vector  $\mathbf{x} \in \mathbb{R}^n$  as  $\langle \mathbf{x}, \mathbf{v}_\beta \rangle = \pi_e(\mathbf{x})$ .

One deduces furthermore from simple algebraic considerations applied to (5.3) that

$$\forall \mathbf{x}, \mathbf{y} \in \mathbb{Q}^n, \ \pi_c(\mathbf{x}) = \pi_c(\mathbf{y}) \iff \langle \mathbf{x}, \mathbf{v}_\beta \rangle = \langle \mathbf{y}, \mathbf{v}_\beta \rangle \iff \mathbf{x} = \mathbf{y}.$$
(5.5)

For more details, see (Canterini and Siegel 2001b, Section 2.1).

# 5.2.2 Central tile

We first introduce the notion of a broken line associated with the fixed point u of  $\sigma$ .

**Definition 5.2.2** The broken line  $L_u$  associated with the fixed point u of the unit Pisot irreducible substitution  $\sigma$  is defined as the broken line in  $\mathbb{R}^n$  whose set of vertices is given by  $\{\mathbf{P}(u_0 \cdots u_{N-1}) \mid N \in \mathbb{N}\}.$ 

We can also describe the broken line as a stair made of a union of segments. More precisely, for  $\mathbf{x} \in \mathbb{Z}^n$  and  $i \in A$ , we denote by  $[\mathbf{x}, i]_g$  the segment  $\{\mathbf{x} + \theta e_i \mid \theta \in [0, 1]\}$ . We call such a segment a *basic geometric* strand, according to (Barge and Kwapisz 2006). We will use and develop this terminology in Section 5.6.1. The broken line associated with u is thus the union of the basic geometric strands  $[\mathbf{P}(u_0 \cdots u_{N-1}), u_N]_g$ , for  $N \in \mathbb{N}$ , *i.e.*,

$$L_u = \bigcup_{N \in \mathbb{N}} [\mathbf{P}(u_0 \cdots u_{N-1}), u_N]_g.$$

**Definition of the central tile.** The central tile (or Rauzy fractal) associated with the unit Pisot irreducible substitution  $\sigma$  is the closure of the projection by  $\pi_c$  onto the contracting space  $\mathbb{H}_c$  of the vertices of the broken line  $L_u$  associated with the fixed point u of  $\sigma$ , *i.e.*,

$$\mathcal{T}_{\sigma} := \overline{\{\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) \mid N \in \mathbb{N}\}}.$$

Subtiles of the central tile  $\mathcal{T}_{\sigma}$  are defined according the the letter  $u_N$  occurring after the word  $u_0 \cdots u_{N-1}$ . Indeed, we set for each  $i \in A$ 

$$\mathcal{T}_{\sigma}(i) := \overline{\{\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) \mid N \in \mathbb{N}, \ u_N = i\}}.$$

By definition, the central tile  $\mathcal{T}_{\sigma}$  consists of the finite union of its subtiles,

*i.e.*,

$$\mathcal{T}_{\sigma} = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i).$$

We will see later (see Corollary 5.2.8, Theorem 5.3.16 and Remark 5.3.17) that the central tile  $\mathcal{T}_{\sigma}$  and the subtiles  $\mathcal{T}_{\sigma}(i)$  do not depend on the choice of u. They only depend on the substitution  $\sigma$ .

**Theorem 5.2.3** Let  $\sigma$  be a unit Pisot irreducible substitution. The central tile  $\mathcal{T}_{\sigma}$  and the subtiles  $\mathcal{T}_{\sigma}(i)$  associated with  $\sigma$  are compact sets.

Proof Note that the compactness of the subtiles  $\mathcal{T}_{\sigma}(i)$  is a direct consequence of the compactness of  $\mathcal{T}_{\sigma}$ , since they are closed subsets of  $\mathcal{T}_{\sigma}$ . To prove the compactness of  $\mathcal{T}_{\sigma}$ , it is enough to show that the points  $\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1})$ , for  $N \in \mathbb{N}$ , remain at a uniformly bounded distance of the origin in  $\mathbb{H}_c$ .

In order to prove this, we use a decomposition of the prefixes  $u_0 \cdots u_{N-1}$ into images by powers of  $\sigma$  of a finite number of words. Since  $\sigma(u) = u$ , there exists a unique  $L \leq N$  such that  $\sigma(u_0 \cdots u_{L-1})$  is a proper prefix of  $u_0 \cdots u_{N-1}$ , and  $u_0 \cdots u_{N-1}$  is a prefix of  $\sigma(u_0 \cdots u_L)$ . In other words, there exists a proper prefix p of  $\sigma(u_L)$ , such that

$$u_0 \cdots u_{N-1} = \sigma(u_0 \cdots u_{L-1}) p \text{ with } \sigma(u_L) = p u_N s.$$
(5.6)

By iterating this process, one gets for every N an expansion of the form

$$u_0 \cdots u_{N-1} = \sigma^K(p_K) \sigma^{K-1}(p_{K-1}) \cdots \sigma(p_1) p_0,$$

where the  $p_i$  belong to a finite set of words that only depends on  $\sigma$ . Note that we have obtained a numeration system on words, the so-called Dumont–Thomas numeration (see Sections 9.4.2 and 5.11 for more details). By (5.2), one has

$$\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) = h_{\sigma}^K \circ \pi_c \circ \mathbf{P}(p_K) + \cdots + h_{\sigma} \circ \pi_c \circ \mathbf{P}(p_1) + \pi_c \circ \mathbf{P}(p_0).$$

We know that  $h_{\sigma}$  is a uniform contraction on  $\mathbb{H}_c$ . As the  $\mathbf{P}(p_i)$  take finitely many values, this implies that the points  $\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1})$ , for  $N \in \mathbb{N}$ , remain at a uniformly bounded distance from the origin, which ends the proof.

# 5.2.3 A graph-directed iterated function system

We now discuss a key property of the central tile and its subtiles, namely they satisfy a set equation. By the solution of a set equation we mean



Fig. 5.1. The central tile and its subtiles for the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$  (left), and its decomposition into subtiles (right).

the following. We are given a collection of finitely many compact sets  $\{K_1, \ldots, K_q\}$ . Each set  $K_i$  can be decomposed as a union of contracted copies of itself and the other sets  $K_j$ . Associated with such a set equation there is a natural graph: its set of vertices is given by  $\{K_i \mid 1 \leq i \leq q\}$  and there is an edge e (labelled  $i \xrightarrow{e} j$ ) from  $K_i$  to  $K_j$  if  $K_j$  appears in the decomposition of  $K_i$ .

Let us formalise this concept by introducing the notion of graph-directed iterated function system. We consider a finite directed graph G with set of vertices  $\{1, \ldots, q\}$  and set of edges E for which each vertex has at least one outgoing edge. With each edge e of the graph, is associated a contractive mapping  $\tau_e : \mathbb{R}^n \to \mathbb{R}^n$ . We call  $(G, \{\tau_e\}_{e \in E})$  a graph-directed iterated function system (GIFS, for short, see (Mauldin and Williams 1988)).

It can be shown by a fixed point argument that given a GIFS  $(G, \{\tau_e\}_{e \in E})$ there exists a unique collection of non-empty compact sets  $K_1, \ldots, K_q \subset \mathbb{R}^n$ having the property that

$$K_i = \bigcup_{\substack{i \stackrel{e}{\longrightarrow} j}} \tau_e(K_j),$$

where the union runs over all edges in G leading away from the vertex i. The sets  $K_i$  are called *GIFS attractors* or solutions of the *GIFS*. Note that the uniqueness statement does not hold for general sets, but only for non-empty compact sets.

Let us see how to apply this formalism to the subtiles  $\mathcal{T}_{\sigma}(i)$ . The graph that will be used is the so-called *prefix-suffix graph*. This graph describes the way images of letters under  $\sigma$  can be decomposed, according to the proof of Theorem 5.2.3. It is the starting point for the construction of several kinds of graphs introduced later in this chapter. For more on this graph, see (Canterini and Siegel 2001a, Canterini and Siegel 2001b).

**Definition 5.2.4 (Prefix-suffix graph)** Let  $\sigma$  be a substitution over the

alphabet A. Let  $P_{\sigma}$  be the finite set

$$P_{\sigma} := \{ (p, i, s) \in A^* \times A \times A^* \mid \exists j \in A, \, \sigma(j) = pis \}.$$

$$(5.7)$$

The set of vertices of the prefix-suffix graph  $\mathcal{G}_{\sigma}$  of  $\sigma$  is the alphabet A. There is an edge labelled by  $(p, i, s) \in P_{\sigma}$  from i towards j if, and only if,  $pis = \sigma(j)$ . We then use the notation  $i \xrightarrow{(p,i,s)} j$ .

**Example 5.2.5** Let us consider as an example the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ , whose central tile is depicted on the left side of Figure 5.1. Its prefix-suffix graph is depicted in Figure 5.2. We recall that  $\varepsilon$  is the empty word.



Fig. 5.2. The prefix-suffix graph for  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ .

By associating with the edge e = (p, i, s) the contraction mapping

$$\tau_e: \nu \in \mathbb{R}^n \mapsto h_\sigma(\nu) + \pi_c \circ \mathbf{P}(p) \in \mathbb{R}^n,$$

we get the GIFS  $(\mathcal{G}_{\sigma}, \{\tau_e\}_{e \in P_{\sigma}})$ . We now can give explicitly the set equation satisfied by the subtiles of the central tile. This is the content of the following theorem (see (Sirvent and Wang 2002) and also (Ito and Rao 2006)).

**Theorem 5.2.6 (Sirvent and Wang 2002)** Let  $\sigma$  be a unit Pisot irreducible substitution over the alphabet A. The subtiles  $\mathcal{T}_{\sigma}(i)$  are the solutions of the GIFS  $(\mathcal{G}_{\sigma}, \{\tau_e\}_{e \in P_{\sigma}})$ , i.e.,

$$\forall i \in A, \ \mathcal{T}_{\sigma}(i) = \bigcup_{\substack{j \in A, \\ i^{(p,i,s)} \neq j}} h_{\sigma}(\mathcal{T}_{\sigma}(j)) + \pi_c \circ \mathbf{P}(p).$$
(5.8)

Furthermore, the union in (5.8) is a measure disjoint union.

258

Before giving a proof of this theorem, let us illustrate it on an example.

**Example 5.2.7** We continue with the substitution  $\sigma$  of Example 5.2.5. In order to decompose  $\mathcal{T}_{\sigma}(1)$  by (5.16) we look for the outgoing edges for the vertex 1 in the prefix-suffix graph. Equation (5.8) gives

$$\begin{aligned} \mathcal{T}_{\sigma}(1) = & h_{\sigma}(\mathcal{T}_{\sigma}(1)) \cup (h_{\sigma}(\mathcal{T}_{\sigma}(1)) + \pi_{c}(\mathbf{e}_{1})) \cup h_{\sigma}(\mathcal{T}_{\sigma}(2)) \\ & \cup (h_{\sigma}(\mathcal{T}_{\sigma}(2)) + \pi_{c}(\mathbf{e}_{1})) \cup h_{\sigma}(\mathcal{T}_{\sigma}(3)), \\ \mathcal{T}_{\sigma}(2) = & h_{\sigma}(\mathcal{T}_{\sigma}(1)) + 2\pi_{c}(\mathbf{e}_{1}), \\ \mathcal{T}_{\sigma}(3) = & h_{\sigma}(\mathcal{T}_{\sigma}(2)) + 2\pi_{c}(\mathbf{e}_{1}). \end{aligned}$$

Hence, the largest subtile  $\mathcal{T}_{\sigma}(1)$  can be decomposed into two shrunken copies of  $\mathcal{T}_{\sigma}(1)$ , two shrunken copies of  $\mathcal{T}_{\sigma}(2)$  and one shrinked copy of  $\mathcal{T}_{\sigma}(3)$ . The subtile  $\mathcal{T}_{\sigma}(2)$  is the geometrically similar image of  $\mathcal{T}_{\sigma}(1)$ , and  $\mathcal{T}_{\sigma}(3)$  is the image of  $\mathcal{T}_{\sigma}(2)$ . This decomposition is illustrated in Figure 5.1 above. Note that the number of pieces in the decomposition of the subtile  $\mathcal{T}_{\sigma}(i)$  is equal to the number of outgoing edges of the vertex *i* in the prefix-suffix graph.

Proof of Theorem 5.2.6 We fix  $i \in A$  and assume that  $u_N = i$ . By definition, one has  $\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) \in \mathcal{T}_{\sigma}(i)$ . By (5.6), there exist L and a decomposition of  $\sigma(u_L)$  as  $\sigma(u_L) = pu_N s = pis$  such that  $\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) = h_{\sigma} \circ \pi_c \circ \mathbf{P}(u_0 \cdots u_{L-1}) + \pi_c \circ \mathbf{P}(p)$ . We thus get  $\pi_c \circ \mathbf{P}(u_0 \cdots u_{N-1}) \in h_{\sigma}(\mathcal{T}_{\sigma}(u_L)) + \pi_c \circ \mathbf{P}(p)$ . As this is true for each N with  $u_N = i$ , by grouping by the values of  $u_L$  and taking the closure, we obtain the decomposition (5.8) for  $\mathcal{T}_{\sigma}(i)$ , *i.e.*,

$$\mathcal{T}_{\sigma}(i) = \bigcup_{(p,j,s), \ \sigma(j) = pis} h_{\sigma}(\mathcal{T}_{\sigma}(j)) + \pi_c \circ \mathbf{P}(p).$$

Recall that  $h_{\sigma}$  scales down the (n-1)-dimensional Lebesgue measure  $\mu_{n-1}$  by the factor  $1/\beta$ . We deduce from (5.8) that

$$\forall i \in A, \ \beta \,\mu_{n-1}(\mathcal{T}_{\sigma}(i)) \leq \sum_{j \in A} m_{ij} \,\mu_{n-1}(\mathcal{T}_{\sigma}(j)), \tag{5.9}$$

where the coefficients  $m_{ji}$  denote the entries of the incidence matrix  $\mathbf{M}_{\sigma}$ . As  $\beta$  is the Perron–Frobenius eigenvalue of  $\mathbf{M}_{\sigma}$ , Lemma 1.4.4 implies the reverse inequality. We thus get equality in (5.9). This implies that no overlap with positive measure occurs in the union in (5.8).

Note that (5.8) admits the following k-fold iteration for any  $k \in \mathbb{N}$  and

 $i \in A$ 

$$\mathcal{T}_{\sigma}(i) = \bigcup_{(p,j,s), \sigma^{k}(j)=pis} h_{\sigma}^{k}(\mathcal{T}_{\sigma}(j)) + \pi_{c} \circ \mathbf{P}(p).$$
(5.10)

From the uniqueness of the solution of (5.8) for non-empty compact sets we deduce the following result.

**Corollary 5.2.8** Let  $\sigma$  be a unit Pisot irreducible substitution. The central tiles  $\mathcal{T}_{\sigma}$  and the subtiles  $\mathcal{T}_{\sigma}(i)$ , for  $i \in A$ , do not depend on the choice of the fixed point u of  $\sigma$ .

We know so far that each subtile  $\mathcal{T}_{\sigma}(i)$  can be decomposed into shrunken copies of the subtiles (namely into sets of the form  $h_{\sigma}(\mathcal{T}_{\tau}(j)) + \pi_c \circ \mathbf{P}(p)$ ) that are disjoint in measure. To ensure that the subtiles  $\mathcal{T}_{\sigma}(i)$ , for  $i \in A$ , themselves are pairwise disjoint in measure, we introduce the following combinatorial condition on substitutions. For substitutions of constant length this condition goes back to (Dekking 1978), see details in (Pytheas Fogg 2002, Chapter 7).

**Definition 5.2.9 (Arnoux and Ito 2001)** A substitution  $\sigma$  over the alphabet A satisfies the *combinatorial strong coincidence condition* if for every pair  $(j_1, j_2) \in A^2$ , there exist  $k \in \mathbb{N}$  and  $i \in A$  such that  $\sigma^k(j_1) = p_1 i s_1$  and  $\sigma^k(j_2) = p_2 i s_2$  with  $\mathbf{P}(p_1) = \mathbf{P}(p_2)$ .

The combinatorial strong coincidence condition is satisfied by every Pisot irreducible substitution over a two-letter alphabet (Barge and Diamond 2002). It is conjectured that every Pisot irreducible substitution satisfies this condition.

The following theorem relates the combinatorial strong coincidence condition to the disjointness of the interiors of the subtiles  $\mathcal{T}_{\sigma}(i), i \in A$ .

**Theorem 5.2.10 (Arnoux and Ito 2001)** Let  $\sigma$  be a unit Pisot irreducible substitution. If  $\sigma$  satisfies the combinatorial strong coincidence condition, then the subtiles  $\mathcal{T}_{\sigma}(i)$  of the central tile  $\mathcal{T}_{\sigma}$  are measure disjoint.

Proof The combinatorial strong coincidence condition implies that for every pair of letters  $(j_1, j_2)$  there exist a common letter i, a positive integer k and a common abelianised prefix  $\mathbf{P}(p)$  such that  $h_{\sigma}^k(\mathcal{T}_{\sigma}(j_1)) + \pi_c \circ \mathbf{P}(p)$  and  $h_{\sigma}^k(\mathcal{T}_{\sigma}(j_2)) + \pi_c \circ \mathbf{P}(p)$  both appear in the k-fold iteration (5.10) of the decomposition of  $\mathcal{T}_{\sigma}(i)$  given by (5.8). Theorem 5.2.6 yields that these tiles are disjoint in measure.

# 5.3 Tilings

In this section we define a tiling as well as a multiple tiling associated with a unit Pisot irreducible substitution  $\sigma$ . We start with some definitions.

# 5.3.1 General definitions

Let  $K_i$ ,  $i \in A$ , be a finite collection of compact sets of a subspace  $\mathbb{H}$  of  $\mathbb{R}^n$ , with each of the  $K_i$  being the closure of its interior. Let p be a positive integer. A *multiple tiling* of degree p of the space  $\mathbb{H}$  by the compact sets  $K_i$ is a collection of translated copies of the sets  $K_i$  of the form  $\mathcal{I} := \{K_i + \gamma \mid (\gamma, i) \in \Gamma\}$ , where  $\Gamma$  is a subset  $\mathbb{H} \times A$ , that satisfies the following conditions.

(i) The entire space  $\mathbb{H}$  is covered by the elements of  $\mathcal{I}$ , *i.e.*,

$$\mathbb{H} = \bigcup_{(\gamma,i)\in\Gamma} K_i + \gamma. \tag{5.11}$$

- (ii) Each compact subset of  $\mathbb{H}$  intersects a finite number of elements of  $\mathcal{I}$ .
- (iii) Almost every point in  $\mathbb{H}$  (with respect to the Lebesgue measure) is covered exactly p times.

The set  $\Gamma$  is called the *translation set*. If the union  $\{K_i + \gamma \mid (\gamma, i) \in \Gamma\}$ only satisfies (i), it is said to be a *covering* of  $\mathbb{H}$ . The sets  $K_i + \gamma$  are called *tiles*. In other words, a multiple tiling is a union of tiles  $\bigcup_{(\gamma,i)\in\Gamma} K_i + \gamma$  that covers the full space  $\mathbb{H}$  with possible overlaps in such a way that almost every point belongs to exactly p tiles. This is illustrated in Figure 5.3 for p = 2with an example obtained in the framework of symmetric beta-expansions taken from (Kalle and Steiner 2009). If p = 1, then the multiple tiling is called a *tiling*. See also Figure 5.7 for an example of a tiling.

Condition (ii) means that the first coordinate projection of  $\Gamma$  into  $\mathbb{H}$  is a *locally finite* subset of  $\mathbb{H}$ , *i.e.*, each point in  $\mathbb{H}$  has a neighbourhood that intersects only finitely many projected elements of  $\Gamma$ . We also say that  $\Gamma$  is a locally finite set.

Dynamical systems can be associated with tilings in close analogy to dynamical systems associated with substitutions. Indeed, the terminology introduced in Chapter 1 concerning words extends in a natural way to tilings. For more on tiling dynamical systems, see (Solomyak 1997) and (Robinson 2004).

Consider a collection of non-empty compact sets  $\{K_i + \gamma \mid (\gamma, i) \in \Gamma\}$ (that is not necessarily a covering or a multiple tiling). A set  $K_i + \gamma$  is said to *occur* in  $\{K_i + \gamma \mid (\gamma, i) \in \Gamma\}$  if  $(\gamma, i) \in \Gamma$ . A *patch* is defined as



Fig. 5.3. A multiple tiling with p = 2.

a finite subset of  $\Gamma$ . It corresponds to a finite union of tiles that occur in  $\{K_i + \gamma \mid (\gamma, i) \in \Gamma\}$ . The translate of a patch  $P = \{(\gamma_1, i_1), \dots, (\gamma_n, i_n)\}$  by  $\nu_0 \in \mathbb{H}$  is defined as  $P + \nu_0 := \{(\gamma_1 + \nu_0, i_1), \dots, (\gamma_n + \nu_0, i_n)\}$ . Two patches  $P = \{(\gamma_1, i_1), \dots, (\gamma_n, i_n)\}$  and  $P' = \{(\gamma'_1, i'_1), \dots, (\gamma'_n, i'_n)\}$  are said to be *equivalent* if they coincide up to a translation vector, that is, if there exists  $\nu_0 \in \mathbb{H}$  such that  $P' = \{(\gamma_1 + \nu_0, i_1), \dots, (\gamma_n + \nu_0, i_n)\}$ .

We now consider a covering of  $\mathbb{H}$ . We say that a ball  $B(\nu, R)$  in  $\mathbb{H}$  is contained in a patch  $P = \{(\gamma_1, i_1), \ldots, (\gamma_n, i_n)\}$  if  $B(\nu, R)$  is a subset of the convex hull of the points  $\gamma$  such that  $(\gamma, i) \in P$ . We define in a similar way the fact that a patch is contained in a ball. The set  $\Gamma$  is said to be *repetitive* if for any finite patch P, there exists R > 0 such that every ball of radius Rin  $\mathbb{H}$  contains a patch which is equivalent to P. This notion is an analogue of the notion of uniform recurrence for words (see Definition 1.2.9).

A subset of  $\mathbb{R}^n$  is said to be a *Delone set* if it is both *uniformly discrete* (there exists r > 0 such that any open ball of radius r contains at most one point of this set) and *relatively dense* (there exists R > 0 such that every closed ball of radius R contains at least one point of this set). We say by extension that  $\Gamma$  is a *Delone set* if its first coordinate projection on  $\mathbb{H}$  is a Delone set. Delone sets have been introduced in the context of point sets and model sets, see *e.g.* (Moody 1997). See also (Lagarias and Pleasants 2002) and (Lagarias and Pleasants 2003) for complexity results on Delone sets that can be compared with analogous results in combinatorics of words on the factor complexity and on the recurrence function.

#### 5.3.2 Tiling of the expanding line

We first associate with  $\sigma$  a tiling by intervals of the expanding half-line  $\mathbb{R}_+ \mathbf{u}_{\beta} \subset \mathbb{H}_e$ . It is obtained by projecting the broken line  $L_u$  associated with u (see Definition 5.2.2) onto the expanding line  $\mathbb{H}_e$  along the contracting

hyperplane  $\mathbb{H}_c$  (see Figure 5.4). This induces a tiling of the half-line  $\mathbb{R}_+ \mathbf{u}_\beta \subset \mathbb{H}_e$ . Using the projection  $\pi_e \colon \mathbb{R}^n \to \mathbb{R}$ ,  $\mathbf{x} \mapsto \langle \mathbf{x}, \mathbf{v}_\beta \rangle$ , we even get a tiling of  $\mathbb{R}_+$  whose tiles are certain translates of the intervals  $I_i = [0, \langle \mathbf{e}_i, \mathbf{v}_\beta \rangle]$  for  $i = 1, \ldots, n$ . In particular, this tiling is obtained by taking the tiles  $I_{u_0}$ ,  $I_{u_1}, \ldots$  adjacent to each other, starting from the origin. The translation set is called the *self-similar translation set* and is equal to

$$\Gamma_e = \{ (\pi_e \circ \mathbf{P}(u_0 \cdots u_{N-1}), u_N) \mid N \ge 0 \}.$$

Since 0 is an endpoint of a tile and since we have assumed that the coordinates of  $\mathbf{v}_{\beta}$  belong to  $\mathbb{Z}[\beta]$ , the endpoints of all tiles are contained in  $\mathbb{Z}[\beta]$ .

We denote the resulting tiling of  $\mathbb{R}_+$  by  $\mathcal{E}_u$  and refer to it as the *self-similar tiling of the expanding line*. For an illustration, see Figure 5.4. One has

$$\mathcal{E}_u := \{ \pi_e[\mathbf{x}, i]_g \mid [\mathbf{x}, i] \in \Gamma_e \},$$
(5.12)

where the basic geometric strand  $[\mathbf{x}, i]_g$  is equal to the segment  $\{\mathbf{x} + \theta \mathbf{e}_i \mid \theta \in [0, 1]\}$ .

The repetitivity of the tiling  $\mathcal{E}_u$  is an easy consequence of the fact that u is uniformly recurrent (see Proposition 1.4.6). The terminology 'self-similar' comes from the fact that the set of endpoints of tiles in  $\mathcal{E}_u$  is stable by multiplication by  $\beta$ . Sections 5.6, 5.7 and 5.8 rely on this tiling.



Fig. 5.4. Projecting the broken line  $L_u$ . In order to illustrate the relation between the tiling  $\mathcal{E}_u$  and the broken line  $L_u$  we draw the tiling  $\mathcal{E}_u$  parallel to the expanding eigendirection  $\mathbf{u}_\beta$  of  $\mathbf{M}_\sigma$  and not in the real line, for  $\sigma(1) = 112$ ,  $\sigma(2) = 21$ .

## 5.3.3 Self-replicating translation set

We now introduce a multiple tiling associated with the substitution  $\sigma$ . The tiles of this multiple tiling are given by the subtiles  $\mathcal{T}_{\sigma}(i)$ ,  $i \in A$ . The

corresponding set of translation vectors is obtained by projecting a suitable subset of points of  $\mathbb{Z}^n$  on the contracting space  $\mathbb{H}_c$ . Let us define this set.

Following (Reveillès 1991), we define a notion of discretisation for the hyperplane  $\mathbb{H}_c$ . The discretised hyperplane is usually called *standard arithmetic discrete hyperplane*. We will use here the shorthand terminology *discrete hyperplane*. We recall that  $\mathbb{H}_c$  is the hyperspace orthogonal to the vector  $\mathbf{v}_{\beta}$ .

**Definition 5.3.1 (Discrete hyperplane)** The discrete hyperplane associated with  $\mathbb{H}_c$  is defined as the set of points  $\mathbf{x} \in \mathbb{Z}^n$  that satisfy

$$0 \le \langle \mathbf{x}, \mathbf{v}_{\beta} \rangle < \sum_{i \in A} \langle \mathbf{e}_i, \mathbf{v}_{\beta} \rangle = ||\mathbf{v}_{\beta}||_1.$$
(5.13)

A discrete hyperplane is a discrete set of points. We now introduce a 'continuous' counterpart to this notion.

**Definition 5.3.2 (Stepped hyperplane)** The stepped hyperplane associated with  $\mathbb{H}_c$  is defined as the union of faces of unit cubes whose vertices belong to the discrete hyperplane associated with  $\mathbb{H}_c$ .

We now want to label the faces contained in a stepped hyperplane. For  $\mathbf{x} \in \mathbb{Z}^n$  and  $i \in A$ , the *face of type i located at*  $\mathbf{x}$  is defined as the face orthogonal to the *i*th canonical vector of the translate of the unit cube located at  $\mathbf{x}$ , *i.e.*,

$$\mathbf{x} + \{\theta_1 \mathbf{e}_1 + \dots + \theta_{i-1} \mathbf{e}_{i-1} + \theta_{i+1} \mathbf{e}_{i+1} + \dots + \theta_n \mathbf{e}_n \mid \theta_j \in [0, 1] \text{ for } j \neq i\}.$$

One checks that a face of type i located at  $\mathbf{x}$  is a subset of the stepped hyperplane if, and only if, one has

$$0 \le \langle \mathbf{x}, \mathbf{v}_{\beta} \rangle < \langle \mathbf{e}_i, \mathbf{v}_{\beta} \rangle. \tag{5.14}$$

For more details, see for instance the references (Berthé and Vuillon 2000), (Arnoux, Berthé, and Ito 2002) or else (Arnoux, Berthé, and Siegel 2004).

Note that a stepped hyperplane is a hypersurface that lives in  $\mathbb{R}^n$ , whereas a discrete hyperplane is a subset of  $\mathbb{Z}^n$ . The discrete hyperplane contains all the vertices of the faces contained in the stepped hyperplane, whereas faces of the stepped hyperplane are labelled by pairs  $(\mathbf{x}, i)$  that satisfy (5.14). This labelling thus consists in selecting some vertices among all the vertices of the discrete hyperplane according to the value  $\langle \mathbf{x}, \mathbf{v}_\beta \rangle$ , hence the difference between the right-hand sides of Inequalities (5.13) and (5.14).

We now project the faces of the stepped hyperplane by  $\pi_c$ .

**Proposition 5.3.3** The collection of projections of the faces of the stepped hyperplane, i.e.,

$$\{\pi_c([\mathbf{x},i]_q) \mid \mathbf{x} \in \mathbb{Z}^n, \ i \in A, \ 0 \le \langle \mathbf{x}, \mathbf{v}_\beta \rangle < \langle \mathbf{e}_i, \mathbf{v}_\beta \rangle \}$$

is a polyhedral tiling of  $\mathbb{H}_c$  by n types of projected faces.

For an explicit proof, see (Berthé and Vuillon 2000) or (Arnoux, Berthé, and Ito 2002). A piece of a stepped hyperplane together with its projection by  $\pi_c$  is depicted in Figure 5.5.

Note that in the Pisot reducible case,  $\mathbb{H}_c$  is no longer a hyperplane and the projections of faces do overlap. There is no universal construction known to obtain an analogue polyhedral tiling (for special cases where this is possible, see (Ei and Ito 2005, Ei, Ito, and Rao 2006)). Nevertheless, one obtains a polyhedral covering.



Fig. 5.5. A stepped hyperplane and its projection on  $\mathbb{H}_c$  as a polyhedral tiling.

We are now going to replace in this polyhedral tiling projected faces by corresponding subtiles (see Figure 5.7). We will see in Section 5.3.5 that this will yield a multiple tiling (Theorem 5.3.13) which is conjectured to be a tiling. This multiple tiling will be called the *self-replicating multiple tiling*.

With each face of type *i* located at **x** included in the stepped hyperplane, we associate a copy of the tile  $\mathcal{T}_{\sigma}(i)$  located at  $\pi_c(\mathbf{x})$  in the contracting space  $\mathbb{H}_c$ . The self-replicating translation set  $\Gamma_c$  is defined as

$$\Gamma_c = \{(\gamma, i) \in \pi_c(\mathbb{Z}^n) \times A \mid \gamma = \pi_c(\mathbf{x}), \, \mathbf{x} \in \mathbb{Z}^n, \ 0 \le \langle \mathbf{x}, \mathbf{v}_\beta \rangle < \langle \mathbf{e}_i, \mathbf{v}_\beta \rangle \}.$$
(5.15)

An element of the form  $(\gamma, i) \in \pi_c(\mathbb{Z}^n) \times A$  is called a *tip*. We denote it by  $[\gamma, i]^*$ . Tips can be considered as symbolic representations of projections of faces. We denote by  $[\gamma, i]_g^*$  the projection by  $\pi_c$  of the face of type *i* located at **x**, with  $\gamma = \pi_c(\mathbf{x})$ , *i.e.*,

$$[\gamma, i]_q^* := \pi_c([\mathbf{x}, i]^*)$$
 with  $\gamma = \pi_c(\mathbf{x})$ .

We thus make the distinction, thanks to the subscript g, between the projected face  $[\pi_c(\mathbf{x}), i]_g^*$  and the tip  $[\pi_c(\mathbf{x}), i]^*$ . The definition of the graphs and the formalism introduced in Sections 5.4 and 5.5 will illustrate the importance of working with symbolic representations.

Note that the discretisation process underlying Definition 5.3.1 is in some sense 'dual' to the notion of broken line (see Definition 5.2.2), hence, the superscript '\*' in the notation  $[\mathbf{x}, i]_g^*$ . Projected faces and segments can also be considered as 'dual'. The use of the symbol '\*' allows us to make the distinction between the notation used for segments and tips. We will develop this duality idea in Section 5.6.1.

Before stating and proving Proposition 5.3.6 below, we need a density result (see Corollary 5.3.5). This density result will be a direct consequence of Kronecker's theorem that we recall here without proof (a proof of this theorem can be found for instance in (Hardy and Wright 1985)).

**Theorem 5.3.4 (Kronecker's theorem)** Let  $r \ge 1$  and let  $\alpha_1, \ldots, \alpha_r$  be real numbers such that  $1, \alpha_1, \ldots, \alpha_r$  are rationally independent. For every  $\varepsilon > 0$  and for every  $(x_1, \ldots, x_r) \in \mathbb{R}^r$ , there exist an element  $N \in \mathbb{N}$  and  $(p_1, \ldots, p_r) \in \mathbb{Z}^r$  such that

$$\forall i \in \{1, \dots, r\}, \ |N\alpha_i - p_i - x_i| < \varepsilon.$$

The proof of the following corollary of Kronecker's theorem can be easily adapted from the proof of (Akiyama 1999, Proposition 1) where it is given in the framework of the beta-numeration, by recalling that the coordinates of  $\mathbf{v}_{\beta}$  are rationally independent. A similar argument can be found in (Canterini and Siegel 2001b, Section 3) stated in terms of minimality of a toral addition.

**Corollary 5.3.5** Let  $\sigma$  be a unit Pisot irreducible substitution. The set  $\pi_c(\{\mathbf{z} \in \mathbb{Z}^n \mid \langle \mathbf{z}, \mathbf{v}_\beta \rangle \ge 0\})$  is dense in  $\mathbb{H}_c$ .

**Proposition 5.3.6** Let  $\sigma$  be a unit Pisot irreducible substitution. Then the following assertions are true.

- (i) The set  $\Gamma_c$  is a Delone set.
- (ii) The union  $\{\mathcal{T}_{\sigma}(i) + \gamma \mid [\gamma, i]^* \in \Gamma_c\}$  is a covering of  $\mathbb{H}_c$ .

*Proof* By Proposition 5.3.3, the projections of the faces of the stepped hyperplane by  $\pi_c$  form a polyhedral tiling of  $\mathbb{H}_c$  with translation set  $\Gamma_c$ , which implies (i).

Let us prove (ii). Let  $\mathbf{z} \in \mathbb{Z}^n$  with  $\langle \mathbf{z}, \mathbf{v}_\beta \rangle \geq 0$ . There exists  $N \in \mathbb{N}$ 

such that if we set  $\mathbf{x} := \mathbf{P}(u_0 \cdots u_{N-1})$ , then  $\langle \mathbf{x}, \mathbf{v}_\beta \rangle \leq \langle \mathbf{z}, \mathbf{v}_\beta \rangle < \langle \mathbf{x}, \mathbf{v}_\beta \rangle + \langle \mathbf{e}_{u_N}, \mathbf{v}_\beta \rangle$ , where  $\mathbf{e}_{u_N} = \mathbf{P}(u_N)$ . One deduces that  $\mathbf{z} - \mathbf{x}$  satisfies (5.14) with  $i = u_N$ . As  $\mathbf{z} = \mathbf{x} + (\mathbf{z} - \mathbf{x})$  this implies that  $\pi_c(\mathbf{z}) \in \mathcal{T}_\sigma(i) + \gamma$  for  $[\gamma, i]^* \in \Gamma_c$  with  $\gamma = \pi_c(\mathbf{z} - \mathbf{x})$  and  $i = u_N$ .

Let  $\nu \in \mathbb{H}_c$ . By Corollary 5.3.5 there exists a sequence  $(\pi_c(\mathbf{z}_k))_{k \in \mathbb{N}}$  with  $\langle \mathbf{z}_k, \mathbf{v}_\beta \rangle \geq 0$  for all k that converges to  $\nu$ . Furthermore, we have seen that for each k, there exists  $[\gamma_k, i_k]^* \in \Gamma_c$  such that  $\pi_c(\mathbf{z}_k) \in \mathcal{T}_\sigma(i_k) + \gamma_k$ . Since the subtiles  $\mathcal{T}_\sigma(i)$ , for  $i \in A$ , are bounded and  $\Gamma_c$  is uniformly discrete, there are infinitely many k for which  $(\gamma_k, i_k)$  takes the same value, say  $(\gamma, i)$ . We thus get  $\nu \in \mathcal{T}_\sigma(i) + \gamma$ , which implies the covering property. This ends the proof of (ii).

**Corollary 5.3.7** The subtiles  $\mathcal{T}_{\sigma}(i)$ , for  $i \in A$ , have non-empty interior.

**Proof** Since the set  $\Gamma_c$  is countable and according to Proposition 5.3.6 (ii), we deduce from Baire's theorem that there exists  $i \in A$  such that the interior of  $\mathcal{T}_{\sigma}(i)$  is not empty. We then deduce from the GIFS equation (5.8) and from the primitivity of  $\sigma$  which implies that the prefix-suffix graph  $\mathcal{G}_{\sigma}$  is strongly connected that all subtiles have non-empty interior.

## 5.3.4 Tip substitutions

It remains to prove that the collection  $\mathcal{I}_{\sigma} := \{\mathcal{T}_{\sigma}(i) + \gamma \mid [\gamma, i]^* \in \Gamma_c\}$ yields a multiple tiling of  $\mathbb{H}_c$ . This will be the content of Theorem 5.3.13 in Section 5.3.5. In order to prove this theorem, we first need to highlight the self-replicating properties of  $\Gamma_c$ . Indeed,  $\Gamma_c$  is stabilised by an inflation mapping acting on  $\pi_c(\mathbb{Z}^n) \times A$ . This inflation mapping is nothing but a substitution on tips (or, equivalently, on faces of cubes), that is inspired by the GIFS equation (5.8) satisfied by the subtiles. We explain this more precisely in the present section.

**Definition 5.3.8 (GIFS substitution)** The (*n*-dimensional) GIFS substitution on tips associated with the (one-dimensional) substitution  $\sigma$ , denoted by  $\mathbf{E}_{1}^{*}$ , is defined on patches of tips by

$$\mathbf{E}_{1}^{*}\{[\gamma, i]^{*}\} = \bigcup_{(p, j, s), \sigma(j) = pis} \{[h_{\sigma}^{-1}(\gamma + \pi_{c} \circ \mathbf{P}(p)), j]^{*}\},$$
(5.16)  
$$\mathbf{E}_{1}^{*}(X_{1}) \cup \mathbf{E}_{1}^{*}(X_{2}) = \mathbf{E}_{1}^{*}(X_{1} \cup X_{2}).$$

We will use the notation  $\mathbf{E}_1^*([\gamma, i]^*)$  for  $\mathbf{E}_1^*\{[\gamma, i]^*\}$ .

Note that we use here the assumption that  $\sigma$  is a unimodular substitution (*i.e.*, its incidence matrix  $\mathbf{M}_{\sigma}$  has determinant  $\pm 1$ ) to ensure that  $h_{\sigma}^{-1}$  maps

 $\pi_c(\mathbb{Z}^n)$  onto  $\pi_c(\mathbb{Z}^n)$ . Indeed, we use the fact that  $h_\sigma \circ \pi_c = \pi_c \circ \mathbf{M}_\sigma$  (see (5.2)), and that the first coordinate  $\gamma$  of a tip belongs to  $\pi_c(\mathbb{Z}^n)$ .

There is a deep relation between the GIFS substitution  $\mathbf{E}_1^*$  and the GIFS equation (5.8), which can indeed be rewritten as

$$\forall [\gamma, i]^* \in \Gamma_c, \ \mathcal{T}_{\sigma}(i) + \gamma = \bigcup_{[\eta, j]^* \in \mathbf{E}_1^*([\gamma, i]^*)} h_{\sigma}(\mathcal{T}_{\sigma}(j) + \eta).$$
(5.17)

This formalism will thus be a particularly convenient way to describe the GIFS equation (5.8) in the graph constructions of Section 5.5. It has been introduced by (Arnoux and Ito 2001) and (Sano, Arnoux, and Ito 2001) (under the name generalised substitutions with the notation  $E_1^*(\sigma)$ ). We omit here the reference to  $\sigma$  for the sake of simplicity in the notation  $\mathbf{E}_1^*$ . The subscript of  $\mathbf{E}_1^*$  stands for the codimension of faces (in the present chapter, they are codimension one faces of hypercubes), while the superscript of  $\mathbf{E}_1^*$  indicates that it is the dual mapping of some mapping  $\mathbf{E}_1$ , that we will introduce in Section 5.6.1. Examples of generalised substitutions are given in (Pytheas Fogg 2002, Chapter 8). Extensions to more general spaces based on faces of hypercubes having higher codimension have also been provided in (Sano, Arnoux, and Ito 2001).

**Example 5.3.9** We continue with the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$  considered in Examples 5.2.5 and 5.2.7.

In order to compute  $\mathbf{E}_1^*([0,i]^*)$  by (5.16) we look for the occurrences of the letter 1 in  $\sigma(1)$ ,  $\sigma(2)$  and  $\sigma(3)$ . This yields

 $\mathbf{E}_1^*([0,1]^*) = [0,1]^* \cup [0,2]^* \cup [0,3]^* \cup [h_{\sigma}^{-1} \circ \pi_c \circ \mathbf{P}(1),1]^* \cup [h_{\sigma}^{-1} \circ \pi_c \circ \mathbf{P}(1),2]^*.$ 

We similarly compute

$$\mathbf{E}_{1}^{*}([0,2]^{*}) = [h_{\sigma}^{-1} \circ \pi_{c} \circ \mathbf{P}(11), 1]^{*}, \qquad \mathbf{E}_{1}^{*}(0,3]^{*}) = [h_{\sigma}^{-1} \circ \pi_{c} \circ \mathbf{P}(11), 2]^{*}.$$

By applying the commutation relation  $\mathbf{h}_c^{-1} \circ \pi_c = \pi_c \circ \mathbf{M}_{\sigma}^{-1}$  (see (5.2)), one gets  $h_{\sigma}^{-1} \circ \pi_c \circ \mathbf{P}(1) = h_{\sigma}^{-1} \circ \pi_c(\mathbf{e}_1) = \pi_c(\mathbf{M}_{\sigma}^{-1}(\mathbf{e}_1)) = \pi_c(\mathbf{e}_3)$ . We thus deduce the following relations

$$\begin{aligned} \mathbf{E}_{1}^{*}([0,1]^{*}) &= [0,1]^{*} \cup [0,2]^{*} \cup [0,3]^{*} \cup [\pi_{c}(\mathbf{e}_{3}),1]^{*} \cup [\pi_{c}(\mathbf{e}_{3}),2]^{*} \\ \mathbf{E}_{1}^{*}([0,2]^{*}) &= [2\pi_{c}(\mathbf{e}_{3}),1]^{*} \\ \mathbf{E}_{1}^{*}([0,3]^{*}) &= [2\pi_{c}(\mathbf{e}_{3}),2]^{*}. \end{aligned}$$

These images are depicted in Figure 5.6, by representing tips as projected faces. Compare with the computation of the decomposition of the subtiles given in Example 5.2.7 which is illustrated in Figure 5.1.

V. Berthé, A. Siegel, J. Thuswaldner



Fig. 5.6. An illustration of the images of the tips  $[0,i]^*$ , for i = 1, 2, 3, under  $\mathbf{E}_1^*$  for the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ . We represent here the tip  $[\pi_c(\mathbf{x}), i]^*$  by its projection  $[\pi_c(\mathbf{x}), i]_g^*$ .

Important properties of  $\mathbf{E}_1^*$  are subsumed in the following theorem. For a proof, see (Arnoux and Ito 2001) and see also (Arnoux, Berthé, and Siegel 2004).

**Theorem 5.3.10 (Arnoux and Ito 2001)** Let  $\sigma$  be a unit Pisot irreducible substitution. Let  $\mathbf{E}_1^*$  be its associated GIFS substitution.

- (i) The images of two different tips in  $\Gamma_c$  under  $\mathbf{E}_1^*$  share no tip in common.
- (ii) The translation set  $\Gamma_c$  is stable under the action of the mapping  $\mathbf{E}_1^*$ .
- (iii) The substitution  $\mathbf{E}_1^*$  maps  $\Gamma_c$  onto  $\Gamma_c$ , i.e.,  $\mathbf{E}_1^*(\Gamma_c) = \Gamma_c$ .

According to Assertion (iii) of Theorem 5.3.10, the set of positions of tiles (given by  $\Gamma_c$ ) is stable under the action of an inflation rule, namely the mapping  $\mathbf{E}_1^*$ , which plays the role of the multiplication by  $\beta$  acting on the tiling of the expanding line introduced in Section 5.3.2. In other

words,  $\Gamma_c$  can be seen as the fixed point of a multidimensional combinatorial transformation, namely  $\mathbf{E}_1^*$ . This explains why the set  $\Gamma_c$  is called *self-replicating* translation set. Note that we use the term 'self-replicating' and not 'self-similar' since the mapping  $h_{\sigma}$  is possibly not a similarity.

This was the first step towards the proof of the multiple tiling property of  $\{\mathcal{T}_{\sigma}(i) + \gamma \mid (\gamma, i) \in \Gamma_c\}$ . Before detailing the proof of this property, let us state the following fundamental result.

**Proposition 5.3.11** Let  $\sigma$  be a unit Pisot irreducible substitution and  $\mathbf{E}_1^*$  be its associated GIFS substitution. If  $[\eta_1, j_1]^*, [\eta_2, j_2]^* \in \mathbf{E}_1^{*N}[\gamma, i]^*$  holds for some  $[\gamma, i]^* \in \Gamma_c$  and some N, then

$$\mu_{n-1}((\mathcal{T}_{\sigma}(j_1) + \eta_1) \cap (\mathcal{T}_{\sigma}(j_2) + \eta_2)) = 0.$$

*Proof* As the GIFS equation (5.17) can be iterated, we obtain the following N-fold iteration of the decomposition of  $\mathcal{T}_{\sigma}(i)$ , *i.e.*,

$$\forall [\gamma, i]^* \in \Gamma_c, \ \mathcal{T}_{\sigma}(i) + \gamma = \bigcup_{[\eta, j]^* \in \mathbf{E}_1^{*N}([\gamma, i]^*)} h_{\sigma}^N(\mathcal{T}_{\sigma}(j) + \eta).$$
(5.18)

According to Theorem 5.2.6 we know that all pairs of pieces in the union on the right-hand side intersect on a set with zero measure.

Thus  $[\eta_1, j_1]^*, [\eta_2, j_2]^* \in \mathbf{E}_1^{*N}[\gamma, i]^*$  implies that the intersection  $h_{\sigma}^N(\mathcal{T}_{\sigma}(j_1) + \eta_1) \cap h_{\sigma}^N(\mathcal{T}_{\sigma}(j_2) + \eta_2)$  has zero measure, which yields that the intersection  $(\mathcal{T}_{\sigma}(j_1) + \eta_1) \cap (\mathcal{T}_{\sigma}(j_2) + \eta_2)$  has measure zero, too.

This proposition can be read as follows: the GIFS equation (5.17) implies that tiles translated by vectors issued from the tips in  $\mathbf{E}_1^{*N}([\gamma, i]^*)$  cannot intersect. This property will be exploited all through this chapter.

# 5.3.5 Self-replicating multiple tiling

We are now going to prove the multiple tiling property. First we need the following statement on subtiles, whose proof follows the proofs of (Praggastis 1999, Proposition 1.1) and (Sing 2006, Proposition 4.99).

**Theorem 5.3.12** Let  $\sigma$  be a unit Pisot irreducible substitution. The boundary of the central tile  $\mathcal{T}_{\sigma}$  as well as the boundary of each of its subtiles  $\mathcal{T}_{\sigma}(i)$ has zero measure. Moreover,  $\mathcal{T}_{\sigma}$  as well as each of its subtiles is the closure of its interior.

Proof One has  $\tau_e(\partial X) = \partial(\tau_e(X))$ , for every  $e \in P_{\sigma}$  and every set X, since in the GIFS equation (5.8) defining the subtiles  $\mathcal{T}_{\sigma}(i)$  the mappings  $\tau_e$  are V. Berthé, A. Siegel, J. Thuswaldner

homeomorphisms. One has furthermore  $\partial(A \cup B) \subseteq \partial A \cup \partial B$ . We use the same notation as in the proof of Theorem 5.2.6. From (5.8) we deduce that

$$\beta \,\mu_{n-1}(\partial \mathcal{T}_{\sigma}(i)) \le \sum_{j \in A} m_{ij} \,\mu_{n-1}(\partial \mathcal{T}_{\sigma}(j)).$$
(5.19)

Similarly as in the proof of Theorem 5.2.6, we obtain equality in (5.19). As the union in (5.8) is measure disjoint, the same is true for the sets  $\partial \mathcal{T}_{\sigma}(i)$ , for  $i \in A$ . In particular, they have either all positive measure, or all zero measure. Assume that they have all positive measure. By Corollary 5.3.7, the subtiles  $\mathcal{T}_{\sigma}(i)$  have non-empty interior. Let  $i \in A$ . Take an open ball Bincluded in the interior of  $\mathcal{T}_{\sigma}(i)$ . We consider (5.18) applied to  $\mathcal{T}_{\sigma}(i)$ , *i.e.*,

$$\mathcal{T}_{\sigma}(i)) = \bigcup_{[\eta,j]^* \in \mathbf{E}_1^{*N}([0,i]^*)} h_{\sigma}^N(\mathcal{T}_{\sigma}(j) + \eta).$$
(5.20)

We then take N large enough for  $\tau_e^N(\mathcal{T}_{\sigma}(j)) \subseteq B$ , for some j such that  $\sigma^N(j) = pis$  and  $e = (p, i, s) \in P_{\sigma^N}$ . Here e = (p, i, s) is an edge of the prefix-suffix graph associated with  $\sigma^N$ . One has  $\tau_e(\nu) = h_{\sigma}^N(\nu) + \pi_c \circ \mathbf{P}(p)$  for  $\nu \in \mathbb{H}_c$ . Note also that such an integer N exists since the mappings  $\tau_e$  are contractions. This implies that  $\partial(\tau_e^N(\mathcal{T}_{\sigma}(j))) \cap \partial \mathcal{T}_{\sigma}(i) = \emptyset$ . We also assume N to be large enough for  $m_{ij}^N > 0$  (here we use the primitivity of  $\mathbf{M}_{\sigma}$  and  $m_{ij}^N$  are the entries of  $\mathbf{M}_{\sigma}^N$ ). We deduce from (5.20) that

$$\partial \mathcal{T}_{\sigma}(i) \subseteq \bigcup_{[\eta,k]^* \in \mathbf{E}_1^{*N}([0,i]^*), \ [\eta,k]^* \neq [\pi_c \circ \mathbf{P}(p),j]^*} \partial h_{\sigma}^N(\mathcal{T}_{\sigma}(k) + \eta).$$
(5.21)

This implies that

$$\mu_{n-1}(\partial \mathcal{T}_{\sigma}(i)) < \beta^{-N} \sum_{k \in A} m_{ik}^{N} \mu_{n-1}(\partial \mathcal{T}_{\sigma}(k)),$$

by recalling that the sets in the union on the right-hand side of (5.21) are disjoint in measure. However, this contradicts with the Nth iteration of (5.19) (where the inequality has been proved to be an equality). We thus have proved that the boundary of each subtile has measure zero.

Let us prove now that each subtile is the closure of its interior. Let  $i \in A$  and let  $\nu \in \mathcal{T}_{\sigma}(i)$ . Let B be an open ball with centre  $\nu$ . We use as previously the Nth decomposition formula (5.18) for N large enough, and obtain  $\nu \in \tau_e^N(\mathcal{T}_{\sigma}(j)) \subseteq B$ , for some j such that  $\sigma^N(j) = pis$  and  $e = (p, i, s) \in P_{\sigma^N}$ . By Corollary 5.3.7,  $\mathcal{T}_{\sigma}(j)$  has non-empty interior, and so does  $\tau_e^N(\mathcal{T}_{\sigma}(j))$ . Hence, B contains interior points of  $\mathcal{T}_{\sigma}(i)$ . We thus have proved that any open ball centred at  $\nu$  contains interior points of  $\mathcal{T}_{\sigma}(i)$  is the closure of its interior. As  $\mathcal{T}_{\sigma}(i) = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$  the same is true for  $\mathcal{T}_{\sigma}$ .

We now have gathered all prerequisites to be able to prove the following theorem (see (Sirvent and Wang 2002), (Berthé and Siegel 2005) and (Ei, Ito, and Rao 2006)).

**Theorem 5.3.13** Let  $\sigma$  be a unit Pisot irreducible substitution. The collection  $\mathcal{I}_{\sigma} = \{\mathcal{T}_{\sigma}(i) + \gamma \mid [\gamma, i]^* \in \Gamma_c\}$  is a multiple tiling of  $\mathbb{H}_c$ . Moreover,  $\Gamma_c$  is repetitive.

*Proof* We subdivide the proof into three parts.

The translation set  $\Gamma_c$  is locally finite. As  $\mathcal{T}_{\sigma}(i)$  is compact for each  $i \in A$  and as  $\Gamma_c$  is a uniformly discrete set according to Proposition 5.3.6, the collection  $\mathcal{I}_{\sigma}$  is *locally finite*, *i.e.*, there exists a positive integer p such that each point of  $\mathbb{H}_c$  is covered at most p times.

The translation set  $\Gamma_c$  is repetitive. We have to prove that for each patch P, there exists R > 0 such that each ball of radius R contains a translate of P. Let us fix a finite patch  $P = \{[\pi_c(\mathbf{z}_k), i_k]^* \mid 1 \leq k \leq \ell\}$  of  $\Gamma_c$ . Let  $R_P$  be chosen in a way that the ball  $B(0, R_P)$  contains the patch P.

We introduce the notion of *slice* above  $\mathbb{H}_c$ . We denote by  $L[a, b] = \{\mathbf{x} \in \mathbb{Z}^n \mid a \leq \langle \mathbf{x}, \mathbf{v}_\beta \rangle < b\}$  the set of points whose height is between a and b. Recall that the set  $\Gamma_c$  corresponds to the projection of points in  $L[0, ||\mathbf{v}_\beta||_{\infty}]$ .

By (5.15), there exists  $\varepsilon_k > 0$  such that  $\mathbf{z}_k$  belongs to the slice  $L[0, (1 - \varepsilon_k)\langle \mathbf{e}_{i_k}, \mathbf{v}_\beta \rangle]$  for each  $k \in \{1, \ldots, \ell\}$ . Set  $\varepsilon := \frac{1}{2} \min_k \langle \varepsilon_k \mathbf{e}_{i_k}, \mathbf{v}_\beta \rangle$  (note that  $\varepsilon > 0$  since P is finite). Still by the definition of  $\Gamma_c$ , we deduce that for every  $\mathbf{x} \in \mathbb{Z}^n$ , assuming  $\mathbf{x} \in L[0, \varepsilon]$  implies that the patch  $\pi_c(\mathbf{x}) + P$  belongs to  $\Gamma_c$ .

It now remains to prove that there exists R > 0 such that any ball of radius R in  $\mathbb{H}_c$  contains a point  $\pi_c(\mathbf{x})$  with  $\mathbf{x} \in L[0, \varepsilon]$ . Recall that the coordinates of  $\mathbf{v}_\beta$  are rationally independent. By Kronecker's theorem (Theorem 5.3.4), there exists  $\mathbf{x}_0 \in \mathbb{Z}^n$  such that  $\mathbf{x}_0 \in L[0, \varepsilon/2]$ . Let us divide the slice  $L[0, ||\mathbf{v}_\beta||_{\infty}]$  into  $N = \lceil ||\mathbf{v}_\beta||_{\infty} 2/\varepsilon \rceil$  slices  $L[j\varepsilon/2, (j+1)\varepsilon/2]$ of height  $\varepsilon/2$ . Since  $0 < \langle \mathbf{x}_0, \mathbf{v}_\beta \rangle < \varepsilon/2$ , each slice can be translated into  $L(0, \varepsilon)$ : for all  $j \leq N$ , there exists  $m_j$  such that  $m_j \mathbf{x}_0 + L[j\varepsilon/2, (j+1)\varepsilon/2] \subset L[0, \varepsilon]$ .

Let us fix a point  $\nu$  in  $\mathbb{H}_c$ . We use the fact that  $\Gamma_c$  is a Delone set, and in particular, that it is relatively dense (see Proposition 5.3.6). Let R' > 0 such that every ball of radius R' > 0 contains the image by  $\pi_c$  of a point of the discrete hyperplane (see Definition 5.3.1). In particular, the ball  $B(\nu, R')$ contains a point  $\pi_c(\mathbf{x})$  with  $\mathbf{x} \in L[0, ||\mathbf{v}_\beta||_{\infty}]$ . There exists j such that the point  $\mathbf{x}$  belongs to one slice  $L[j\varepsilon/2, (j+1)\varepsilon/2]$ , hence there exists  $m_j$  such that  $\mathbf{x} + m_j \mathbf{x}_0 \in L[0, \varepsilon]$ . From above, this implies that  $\pi_c(\mathbf{x} + m_j \mathbf{x}_0) + P$  occurs in  $\Gamma_c$ .

We deduce that the ball centred at  $\nu$  with radius  $R := R' + \max_k ||m_k \mathbf{x}_0|| + R_P$  contains a copy of the initial patch P up to translation. As  $\nu \in \mathbb{H}_c$  was arbitrary this proves the repetitivity of  $\Gamma_c$ .

The collection  $\mathcal{I}_{\sigma}$  is a multiple tiling. Suppose that this is wrong. Since the boundary of each subtile has measure zero by Theorem 5.3.12, the union of the boundaries of all elements of  $\mathcal{I}_{\sigma}$  also has measure zero. Thus there are  $\nu_1, \nu_2 \in \mathbb{H}_c$ , positive integers  $\ell_1 \neq \ell_2$  and  $\varepsilon > 0$  such that  $B(\nu_j, \varepsilon)$  is covered exactly  $\ell_j$  times by the collection  $\mathcal{I}_{\sigma}$ , for j = 1, 2, i.e., the points contained in  $B(\nu_j, \varepsilon)$  belong to exactly  $\ell_j$  tiles of the collection  $\mathcal{I}_{\sigma}$ . More precisely, there are patches  $P_1, P_2 \subset \Gamma_c$  of cardinality  $\ell_1$  and  $\ell_2$ , respectively, such that  $B(\nu_j, \varepsilon) \subset \bigcap_{[\gamma,i]^* \in P_j} (\mathcal{I}_{\sigma}(i) + \gamma)$ , for j = 1, 2. Moreover,  $B(\nu_j, \varepsilon)$  has empty intersection with each tile of  $\mathcal{I}_{\sigma}$  that is not contained in  $P_j$ . We assume w.l.o.g. that  $\ell_1 < \ell_2$ .

Consider now the inflated family  $h_{\sigma}^{-m}\mathcal{I}_{\sigma}$  (we recall that the inverse of  $h_{\sigma}$  is an expansive mapping). By the arguments above each point in  $h_{\sigma}^{-m}B(\nu_1,\varepsilon)$ is contained in exactly  $\ell_1$  tiles of  $h_{\sigma}^{-m}\mathcal{I}_{\sigma}$ . By Theorem 5.3.10 (iii), each tile of  $h_{\sigma}^{-m}\mathcal{I}_{\sigma}$  has the shape  $h_{\sigma}^{-m}(\mathcal{I}_{\sigma}(i) + \gamma)$ , with  $[\gamma, i]^* \in \Gamma_c$ . By (5.17) and Proposition 5.3.11, such a tile can be decomposed as a finite union of tiles in  $\mathcal{I}_{\sigma}$  which are pairwise disjoint in measure. Thus almost each point in  $h_{\sigma}^{-m}B(\nu_1,\varepsilon)$  is contained in exactly  $\ell_1$  tiles of the family  $\mathcal{I}_{\sigma}$ .

Since the translation set  $\Gamma_c$  is repetitive, we can choose m so large that  $h_{\sigma}^{-m}B(\nu_1,\varepsilon)$  contains a translated copy  $P_2 + \gamma$  of the patch  $P_2$ . This means that  $B(\nu_2,\varepsilon) + \gamma$  is contained in  $h_{\sigma}^{-m}B(\nu_1,\varepsilon)$  for a large enough m. Recall that  $B(\nu_2,\varepsilon)$  is covered exactly  $\ell_2$  times by  $\mathcal{I}_{\sigma}$ . There is a priori no reason for  $B(\nu_2,\varepsilon) + \gamma$  to be covered exactly  $\ell_2$  times by  $\mathcal{I}_{\sigma}$ . Indeed other tiles might 'invade'  $B(\nu_2,\varepsilon) + \gamma$ . Nevertheless, it is covered at least  $\ell_2$  times by elements of  $\mathcal{I}_{\sigma}$ . This yields a contradiction since almost every point in  $h_{\sigma}^{-m}B(\nu_1,\varepsilon)$  is contained in exactly  $\ell_1$  tiles of  $\mathcal{I}_{\sigma}$ , and  $\ell_1 < \ell_2$ .

**Definition 5.3.14** Let  $\sigma$  be a unit Pisot irreducible substitution. We call the multiple tiling  $\mathcal{I}_{\sigma}$  defined in Theorem 5.3.13 the *self-replicating multiple tiling* associated with  $\sigma$ .

For all known examples of unit Pisot irreducible substitutions the self-replicating multiple tiling is indeed a tiling, as illustrated in Figure 5.7, for instance.



Fig. 5.7. The self-replicating (multiple) tiling for  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ . This multiple tiling is indeed a tiling for this substitution.

**Definition 5.3.15 (Tiling property)** A unit Pisot irreducible substitution  $\sigma$  satisfies the *tiling property* if the self-replicating multiple tiling is a tiling.

The Pisot conjecture states that as soon as  $\sigma$  is a unit Pisot irreducible substitution, the tiling property holds. Let us note that the Pisot conjecture has been proved to hold for unit Pisot irreducible substitutions over a two-letter alphabet in (Hollander and Solomyak 2003). The proof strongly relies on the fact that the combinatorial strong coincidence condition is satisfied by every Pisot irreducible substitution over a two-letter alphabet (Barge and Diamond 2002), although the combinatorial strong coincidence condition does not imply the tiling property for a general alphabet.

Note that an immediate reformulation of the tiling property is that  $\mu_{n-1}((\mathcal{T}(i) + \gamma) \cap (\mathcal{T}(j) + \eta)) = 0$ , for every pair of distinct tiles  $\{\mathcal{T}(i) + \gamma, \mathcal{T}(j) + \eta\}$  of the self-replicating multiple tiling.

Note also that in view of the following theorem the assumption that u is generated by  $\sigma$  causes no loss of generality.

**Theorem 5.3.16** Let  $\sigma$  be a unit Pisot irreducible substitution. Let  $k, \ell$  be two positive integers. One has  $\mathcal{T}_{\sigma^k} = \mathcal{T}_{\sigma^\ell}$ . Furthermore, the substitution  $\sigma^k$  satisfies the tiling property if, and only if,  $\sigma^\ell$  satisfies the tiling property.

*Proof* Let us note that  $h_{\sigma^k} = h_{\sigma}^k$  for all k. According to (5.10), the central tiles  $\mathcal{T}_{\sigma^k}$  and  $\mathcal{T}_{\sigma^\ell}$  are seen to satisfy

V. Berthé, A. Siegel, J. Thuswaldner

$$\forall i \in A, \ \mathcal{T}_{\sigma^{k}}(i) = \bigcup_{\substack{j, (p, i, s), \ \sigma^{k\ell}(j) = pis}} h_{\sigma}^{k\ell}(\mathcal{T}_{\sigma^{k}}(j)) + \pi_{c} \circ \mathbf{P}(p) \quad \text{and}$$
$$\forall i \in A, \ \mathcal{T}_{\sigma^{\ell}}(i) = \bigcup_{\substack{j, (p, i, s), \ \sigma^{k\ell}(j) = pis}} h_{\sigma}^{k\ell}(\mathcal{T}_{\sigma^{\ell}}(j)) + \pi_{c} \circ \mathbf{P}(p),$$

respectively. One deduces that  $\mathcal{T}_{\sigma^k}(i)$  and  $\mathcal{T}_{\sigma^\ell}(i)$  satisfy the same GIFS equation, and thus, that they coincide. Furthermore, the set  $\Gamma_c$  only depends on  $\mathbf{v}_{\beta}$ , which is a common left eigenvector for  $\sigma^k$  and  $\sigma^{\ell}$ . This concludes the proof.

**Remark 5.3.17** Let  $\sigma$  be a unit Pisot irreducible substitution that is possibly not prolongable. Assume that  $\sigma^k$  is prolongable for some k (such a k always exists by primitivity of  $\sigma$ ). Let u be generated by  $\sigma^k$  with  $\sigma^k(u) = u$ . We define the *central tile* associated with  $\sigma$  as  $\mathcal{T}_{\sigma} := \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$ , where the non-empty compact sets  $\mathcal{T}_{\sigma}(i)$  are uniquely determined by the following GIFS equation

$$\forall i \in A, \ \mathcal{T}_{\sigma}(i) = \bigcup_{j, (p, i, s), \ \sigma^{j} = pis} h_{\sigma}(\mathcal{T}_{\sigma}(j)) + \pi_{c} \circ \mathbf{P}(p).$$

By taking the k-fold iteration of this equation and by uniqueness of its solution, we deduce that  $\mathcal{T}_{\sigma}(i) = \mathcal{T}_{\sigma^k}(i)$ , for every  $i \in A$ .

#### 5.4 Ancestor graphs and tiling conditions

In the remaining part of this chapter we present various conditions for the self-replicating multiple tiling to be a tiling. Recall that the substitution  $\sigma$  satisfies the tiling property if, and only if, each intersection of distinct tiles in the self-replicating multiple tiling has zero measure. In the present section we focus on effective ways to control the measure of intersections of tiles. In Section 5.4.1 we introduce a sufficient condition for the tiling property. In Section 5.4.2 we define a graph that provides an effective way to check this sufficient condition. This leads us to introduce a more intricate graph in Section 5.4.3. This graph yields a necessary and sufficient condition for the tiling property.

## 5.4.1 Finiteness properties

We have already gained information on intersections of subtiles with zero measure. Indeed, Theorem 5.2.6 states that the shrunken copies of subtiles occurring in the decomposition of each subtile  $\mathcal{T}_{\sigma}(i)$  are disjoint in measure. Moreover, by Theorem 5.2.10, the subtiles  $\mathcal{T}_{\sigma}(i)$ ,  $i \in A$ , are disjoint if the

substitution  $\sigma$  satisfies the combinatorial strong coincidence condition. We now define a sufficient condition that allows this information to be spread on zero measure intersections throughout the self-replicating multiple tiling, and thus, to exhibit a sufficient condition for the multiple tiling to be indeed a tiling.

Let U denote the patch

$$U := [0,1]^* \cup [0,2]^* \cup \dots \cup [0,n]^*.$$
(5.22)

It is easy to see that  $U \subset \Gamma_c$  by (5.15). Pursuing the analogy between tips and their geometric representations in terms of projected faces (see Section 5.3.3) we call U by slight abuse of language the *lower unit cube*.

One easily checks that U is contained in  $\mathbf{E}_1^*(U)$ . Indeed, for any  $j \in A$ ,  $[0, j^*]$  is contained in  $\mathbf{E}_1^*([0, i]^*)$ , where i is the first letter of  $\sigma(j)$ . Hence, we deduce from Theorem 5.3.10 (ii) that the sequence of patches  $(\mathbf{E}_1^{*m}(U))_{m\geq 0}$ is an increasing sequence of subsets of  $\Gamma_c$  with respect to inclusion. A specific case occurs when  $\mathbf{E}_1^{*m}(U)$  eventually covers the entire self-replicating translation set  $\Gamma_c$  if m tends to infinity. As an illustration, the set  $\mathbf{E}_1^{*m}(U)$ is depicted in Figures 5.8 and 5.9, in each case for a specific m, for the substitutions  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$  and  $\tau(1) = 2$ ,  $\tau(2) = 3$ ,  $\tau(3) = 12$ . These pictures indicate that  $\mathbf{E}_1^{*m}(U)$  eventually covers the whole self-replicating translation set  $\Gamma_c$  in the case of  $\sigma$ , but not in the case of  $\tau$ .



Fig. 5.8. The patch  $\mathbf{E}_1^{*5}(U)$  for the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ .

**Definition 5.4.1 (Geometric finiteness property)** Let  $\sigma$  be a unit Pisot irreducible substitution and  $\mathbf{E}_1^*$  be its associated GIFS substitution V. Berthé, A. Siegel, J. Thuswaldner



Fig. 5.9. The patch  $\mathbf{E}_{1}^{*15}(U)$  for the substitution  $\tau(1) = 2, \tau(2) = 3, \tau(3) = 12$ .

on tips. We say that  $\sigma$  satisfies the geometric finiteness property if

$$\Gamma_c = \bigcup_{m \in \mathbb{N}} \mathbf{E}_1^{*m}(U).$$
(5.23)

Let us see how to propagate information on zero measure intersections inside the subtiles  $\mathcal{T}_{\sigma}(i)$  to all intersections occurring in the self-replicating tiling when  $\sigma$  satisfies the geometric finiteness property.

**Theorem 5.4.2** Let  $\sigma$  be a unit Pisot irreducible substitution. If  $\sigma$  satisfies both the geometric finiteness property and the combinatorial strong coincidence condition, then the self-replicating multiple tiling is a tiling.

Proof Let us consider two tiles in the self-replicating multiple tiling, namely  $\mathcal{T}_{\sigma}(i_1) + \gamma_1$  and  $\mathcal{T}_{\sigma}(i_2) + \gamma_2$ . By the geometric finiteness property and by (5.17), there exist  $N, j_1, j_2$  such that  $\mathcal{T}_{\sigma}(i_1) + \gamma_1$  (respectively  $\mathcal{T}_{\sigma}(i_2) + \gamma_2$ ) is a piece of the Nth level decomposition (5.18) of a subtile  $\mathcal{T}_{\sigma}(j_1)$  (respectively  $\mathcal{T}_{\sigma}(j_2)$ ). If  $j_1 = j_2$ , we are done because we fall into the assumptions of Proposition 5.3.11. If  $j_1 \neq j_2$ , we know from Theorem 5.2.10 and the combinatorial strong coincidence assumption that  $\mathcal{T}_{\sigma}(j_1)$  and  $\mathcal{T}_{\sigma}(j_2)$  are disjoint up to a set of zero measure.

A more restrictive condition for tiling is the following *superfiniteness property* (compare with Definition 5.4.1).

**Definition 5.4.3 (Geometric superfiniteness property)** Let  $\sigma$  be a unit Pisot irreducible substitution and  $\mathbf{E}_1^*$  be its associated GIFS substitution on tips. We say that  $\sigma$  satisfies the *geometric superfiniteness property* if there exists  $i \in A$  such that

$$\Gamma_c = \bigcup_{m \in \mathbb{N}} \mathbf{E}_1^{*m}([0,i]^*).$$

Note that in this case, the proof of Theorem 5.4.2 applies without requiring the assumption of the combinatorial strong coincidence.

## 5.4.2 The ancestor graph

We will now discuss how to check in an effective way whether the geometric finiteness property holds. The idea is to prove that the geometric finiteness property is satisfied if, and only if, an explicit finite patch (depending on  $\sigma$ ) is eventually covered by the iterations of  $\mathbf{E}_1^*$  on the lower unit cube U.

As a consequence of Theorem 5.3.10, every tip has a unique pre-image under the action of  $\mathbf{E}_1^*$ . We will refer to this pre-image as *ancestor*.

**Definition 5.4.4 (Ancestor of a tip)** The ancestor of  $[\eta, j]^* \in \Gamma_c$  is the unique tip  $[\gamma, i]^* \in \Gamma_c$  for which  $[\eta, j]^* \in \mathbf{E}_1^*([\gamma, i]^*)$ .

We have worked so far with the Euclidean norm in  $\mathbb{H}_c$ . We now introduce a more convenient norm based on (5.3). Let  $|| \cdot ||_c$  denote the maximum norm on  $\mathbb{H}_c$  with respect to vectors  $\mathbf{u}_{\beta^{(i)}}$  for  $i \geq 2$ , *i.e.*,

$$\forall \nu \in \mathbb{H}_c, \, ||\nu||_c = \max\{|\langle \nu, \mathbf{v}_{\beta^{(i)}}\rangle| \mid i = 2, \dots, r+s\}.$$

$$(5.24)$$

Let  $\beta_{\max} := \max\{|\beta^{(j)}| \mid j \ge 2\}$ . One has

$$||h_{\sigma}(\nu)||_{c} \leq \beta_{\max} ||\nu||_{c} \text{ for all } \nu \in \mathbb{H}_{c}.$$
(5.25)

We denote by  $B_c(\nu, R)$  the ball centred at  $\nu$  of radius R with respect to this norm. Let  $M_{\sigma} := \max\{||\pi_c \circ \mathbf{P}(p)||_c \mid (p, a, s) \in P_{\sigma}\}$  (see (5.7) for the definition of  $P_{\sigma}$ ).

**Definition 5.4.5 (Seed patch)** The seed patch  $V_{\sigma}$  associated with the substitution  $\sigma$  is defined as

$$V_{\sigma} := \left\{ [\gamma, i]^* \in \Gamma_c \mid ||\gamma||_c \le \frac{M_{\sigma}}{1 - \beta_{\max}} \right\}.$$
(5.26)

**Remark 5.4.6** Note that  $M_{\sigma}/(1 - \beta_{\max})$  is an upper bound for the diameter of the tiles  $\mathcal{T}_{\sigma}(i)$ , according to the proof of Corollary 5.2.8. Thus  $0 \in \mathcal{T}_{\sigma}(j) + \gamma$  implies that  $[\gamma, j]^* \in V_{\sigma}$ .

**Theorem 5.4.7** Let  $\sigma$  be a unit Pisot irreducible substitution. One has

$$\Gamma_c = \bigcup_{m \in \mathbb{N}} \mathbf{E}_1^{*m}(V_{\sigma})$$

*Proof* The definition of  $\mathbf{E}_1^*$  yields that if  $[\gamma, i]^*$  is the ancestor of the tip  $[\eta, j]^*$ , then

$$\gamma = h_{\sigma}(\eta) - \pi_c \circ \mathbf{P}(p), \qquad (5.27)$$

where p is a prefix of  $\sigma(i)$ . We fix  $[\eta, j]^* \in \Gamma_c$ . Following Definition 5.4.4, let  $[\gamma_k, i_k]^*$  be the successive ancestors of  $[\eta, j]^*$ , *i.e.*,  $[\eta, j]^* \in \mathbf{E}_1^*([\gamma_1, i_1]^*)$ and  $[\gamma_k, i_k]^* \in \mathbf{E}_1^*([\gamma_{k+1}, i_{k+1}]^*)$  for all  $k \geq 1$ . By (5.27), one has  $\gamma_{k+1} = h_\sigma(\gamma_k) - \pi_c \circ \mathbf{P}(p)$  where p is a prefix of  $\sigma(i_k)$ . Therefore we have by (5.25)

$$||\gamma_{k+1}||_c \le \beta_{\max}||\gamma_k||_c + M_{\sigma}.$$
(5.28)

Let  $\alpha \in (\beta_{\max}, 1)$ . Then, if  $a \ge M_{\sigma}/(\alpha - \beta_{\max})$ , one has  $\beta_{\max}a + M_{\sigma} \le \alpha a$ . This implies

$$||\gamma_k||_c \ge \frac{M_\sigma}{\alpha - \beta_{\max}} \implies ||\gamma_{k+1}||_c \le \alpha ||\gamma_k||_c.$$
(5.29)

Let  $V^{(\alpha)} := \left\{ [\eta, j]^* \in \Gamma_c \mid ||\gamma||_c < \frac{M_{\sigma}}{\alpha - \beta_{\max}} \right\}$ . All the  $V^{(\alpha)}$  are finite patches since  $\Gamma_c$  is uniformly discrete. We also notice that

$$\bigcap_{\beta_{\max} < \alpha < 1} V^{(\alpha)} = V_{\sigma}$$

Therefore, there exists  $\alpha_0 < 1$  such that  $V^{(\alpha_0)} = V_{\sigma}$  for all  $\alpha_0 \leq \alpha < 1$ .

By iterating (5.29) we deduce that there is  $k \in \mathbb{N}$  such that the *k*th ancestor  $[\gamma_k, i_k]^*$  of  $[\eta, j]^*$  satisfies  $[\gamma_k, i_k]^* \in V_{\alpha_0}$ . As  $V^{(\alpha_0)} = V_{\sigma}$ , this implies that  $\Gamma_c = \bigcup_{m \in \mathbb{N}} \mathbf{E}_1^{*m}(V_{\sigma})$ .

Theorem 5.4.7 is based on the fact that  $\beta$  is a Pisot number. Analogous statements appear in various frameworks, see for instance the references (Akiyama 2000), (Arnoux, Berthé, and Siegel 2004), (Barge and Diamond 2002), (Barge and Kwapisz 2006), (Fernique 2006), (Fuchs and Tijdeman 2006) or (Ito and Rao 2006).

**Remark 5.4.8** According to (5.28), one checks that  $V_{\sigma}$  contains the ancestors of all its elements (but note also that  $V_{\sigma}$  is not stable under the action of  $\mathbf{E}_1^*$ ). Furthermore, the seed patch  $V_{\sigma}$  is easily seen to be effectively computable. Note also that  $U \subseteq V_{\sigma}$ .

We deduce from Theorem 5.4.7 the following corollary.

**Corollary 5.4.9** If there exists  $m \geq 1$  such that  $\mathbf{E}_1^{*m}(U)$  contains  $V_{\sigma}$ , then the geometric finiteness property holds. In this case, we can effectively exhibit such an m.

The proof of Theorem 5.4.7 mostly relies on the notion of ancestor. In order to obtain an algorithmic way to check the geometric finiteness property, we construct a directed graph based on the notion of ancestor and on the seed patch.

**Definition 5.4.10 (Ancestor graph)** The vertices of the ancestor graph are the tips that occur in the seed patch  $V_{\sigma}$  introduced in Definition 5.4.5. There is an edge from  $[\eta, j]^*$  to  $[\gamma, i]^*$  if  $[\gamma, i]^*$  is the ancestor of  $[\eta, j]^*$ , *i.e.*,  $[\eta, j]^* \in \mathbf{E}_1^*([\gamma, i]^*)$ .

The computation of the ancestor graph is straightforward. First, list the tips that belong to the seed patch  $V_{\sigma}$ . Then, for every  $[\gamma, i]^* \in V_{\sigma}$ , compute the tips  $[\eta, j]^* \in \mathbf{E}_1^*([\gamma, i]^*)$ , and draw an edge from every  $[\eta, j]^*$  to  $[\gamma, i]^*$ , if  $[\eta, j]^* \in V_{\sigma}$ .

Remark 5.4.11 The choice of orientation we have made here for the ancestor graph (which consists in following the ancestor relation) might seem to be counter-intuitive at first sight, and in contradiction with the orientation of edges in the prefix-suffix graph. Note that a converse choice has been made in (Siegel and Thuswaldner 2010) for similar graph constructions. Their purpose was to study the boundary of subtiles and thus, to be able to zoom inside the subtiles. On the opposite, here we want to be able to zoom outside the subtiles in order to cover the self-replicating multiple tiling, hence, to trace back ancestors.

By uniqueness of the ancestor and by stability of  $V_{\sigma}$  with respect to ancestors, every vertex in the ancestor graph admits exactly one outgoing egde. This implies that every sufficiently long path in this finite graph reaches a cycle and cannot exit from it. By a *cycle*, we mean a closed directed path. Note furthermore that a tip of the form  $[0, j]^*$  admits a unique outgoing edge which is also of the form  $[0, i]^*$ . Indeed, the (unique) ancestor of  $[0, j]^*$  is the tip  $[0, i]^*$  where *i* is the first letter of  $\sigma(j)$ . Hence, once a cycle contains a tip of *U*, it contains only elements of *U*. We say that it is *contained* in *U*. This provides a simple effective condition to check the geometric finiteness property.

**Proposition 5.4.12** Let  $\sigma$  be a unit Pisot irreducible substitution. The geometric finiteness property is satisfied if, and only if, all cycles in the ancestor graph are contained in U.

**Proof** If the geometric finiteness property holds, then any sufficiently long path in the ancestor graph contains a tip of U, and thus any cycle in the ancestor graph is contained in U. Conversely, assume that any cycle is contained in U. Any tip in  $V_{\sigma}$  admits in its sequence of successive ancestors one element that belongs to a cycle, hence to U, which ends the proof in view of Corollary 5.4.9.

**Example 5.4.13** Two examples of ancestor graphs are depicted in Figures 5.10 and 5.11. One can see that the graph corresponding to  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$  satisfies the condition of Proposition 5.4.12, whereas the graph corresponding to  $\tau(1) = 2$ ,  $\tau(2) = 3$ ,  $\tau(3) = 12$  does not satisfy it. In this second example one can see that the graph, which is made of two connected components, admits two cycles, only one of them made of tips of U.



Fig. 5.10. The ancestor graph for the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 113$ ,  $\sigma(3) = 1$ . To keep notation simple in the picture of the graph, we omitted the projection  $\pi_c$  in the labels of the vertices.

# 5.4.3 The two-piece ancestor graph

The geometric finiteness property means that the self-replicating translation set  $\Gamma_c$  can be covered by iterating  $\mathbf{E}_1^*$  on the patch U. It turns out that



Fig. 5.11. The ancestor graph for the substitution  $\tau(1) = 2$ ,  $\tau(2) = 3$ ,  $\tau(3) = 12$ . To keep notation simple in the picture of the graph, we omitted the projection  $\pi_c$  in the labels of the vertices.

this condition is not necessary for the tiling property. To overcome this, we will use the repetitivity of  $\Gamma_c$ , and deal with translations of the sets  $\mathbf{E}_1^{*m}(U)$ . This will lead us to introduce a further graph, inspired by the ancestor graph, which will allow an algorithmic criterion equivalent to the tiling property to be given, and not only a sufficient condition, such as Proposition 5.4.12.

**Theorem 5.4.14 (Ito and Rao 2006)** Let  $\sigma$  be a unit Pisot irreducible substitution and  $\mathbf{E}_1^*$  be its associated GIFS substitution on tips. The selfreplicating multiple tiling is a tiling if, and only if, for every  $i \in A$ , the radius of the largest ball contained in the union

$$\bigcup_{[\gamma,j]^* \in \mathbf{E}_1^{*m}([0,i]^*)} [\gamma,j]_g^*$$
(5.30)

tends to infinity with m.

Proof Let us note that this statement does not depend on the choice of the norm by the equivalence of norms. Assume that  $\sigma$  satisfies the tiling property. We fix  $i \in A$ . For  $m \in \mathbb{N}$ , let  $B_c(\delta_m, R_m)$  be the ball (for the norm  $|| \cdot ||_c$ ) with largest radius contained in  $h_{\sigma}^{-m}(\mathcal{T}_{\sigma}(i))$ . Since  $\mathcal{T}_{\sigma}(i)$  has non-empty interior (Corollary 5.3.7) and since  $h_{\sigma}^{-1}$  is an expansion, we have that  $\lim_{m\to+\infty} R_m = \infty$ . The GIFS equation (5.17) yields that  $h_{\sigma}^{-m}\mathcal{T}_{\sigma}(i)$ is covered by the tiles  $\mathcal{T}_{\sigma}(j) + \gamma$  with  $[\gamma, j]^* \in \mathbf{E}_1^{*m}([0, i]^*)$ . From the tiling assumption, we deduce that for every tip  $[\eta, k]^* \notin \mathbf{E}_1^{*m}([0, i]^*)$ , the tile  $\mathcal{T}_{\sigma}(k) + \eta$  is measure disjoint from  $B_c(\delta_m, R_m)$ . Let C denote the diameter of the central tile  $\mathcal{T}_{\sigma}$ , *i.e.*,  $C = \sup\{||\nu - \nu'||_c \mid \nu, \nu' \in \mathcal{T}_{\sigma}\}$ . Therefore, every tip  $[\gamma, j]^*$  with  $||\gamma - \delta_m||_c < R_m - C$  has to belong to  $\mathbf{E}_1^{*m}([0, i]^*)$ . This implies that the radius of the largest ball contained in the union in (5.30) tends to infinity with m.

Conversely, assume that the radius of the largest ball contained in the union in (5.30) tends to infinity with m. Let P be a patch of  $\Gamma_c$ . By repetitivity (Theorem 5.3.13), P is contained, up to a translation vector, in any large enough ball of  $\Gamma_c$ . Therefore there exist  $\nu \in \mathbb{H}_c$  and m > 0 such that  $\nu + P \subset \mathbf{E}_1^{*m}([0,i]^*)$ , and thus  $P \subset \mathbf{E}_1^{*m}([h_{\sigma}^m\nu,i]^*)$  by the definition of  $\mathbf{E}_1^*$  in (5.16). Proposition 5.3.11 then yields that the tiles  $\mathcal{T}_{\sigma}(j) + \gamma$  with  $[\gamma, j] \in P$  have pairwise disjoint interiors. This implies that  $\sigma$  satisfies the tiling property.

**Corollary 5.4.15** Let  $\sigma$  be a unit Pisot irreducible substitution. The self-replicating multiple tiling is a tiling if, and only if, for every pair of tips  $([\eta_1, j_1]^*, [\eta_2, j_2]^*) \in \Gamma_c^2$  there exist  $\delta \in \mathbb{H}_c$ ,  $m \ge 0$ , and  $i \in A$  such that

$$\delta + \{ [\eta_1, j_1]^*, [\eta_2, j_2]^* \} \subset \mathbf{E}_1^{*m}([0, i]^*).$$
(5.31)

Proof If this condition is satisfied, Proposition 5.3.11 implies that  $\mathcal{T}_{\sigma}(j_1) + \eta_1$ and  $\mathcal{T}_{\sigma}(j_2) + \eta_2$  do not overlap for arbitrarily chosen  $[\eta_1, j_1]^*$  and  $[\eta_2, j_2]^*$ . Therefore the tiling property is satisfied. Conversely, by Theorem 5.4.14, the tiling property implies that  $\mathbf{E}_1^{*m}([0,i]^*)$  contains arbitrarily large balls for large m. Thus, by the repetitivity assertion of Theorem 5.3.13, a translation of each patch  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\} \subset \Gamma_c$  occurs in  $\mathbf{E}_1^{*m}([0,i]^*)$  for some  $i \in A$  and some  $m \in \mathbb{N}$ .

The formulation of the tiling property given by Corollary 5.4.15 means that every pair of tips belongs to the image of a tip in U up to a common translation vector. In order to check (5.31), we need to trace back ancestors of patches up to a translation vector. When dealing with (5.31), we will use the existence of the translation vector  $\delta$  in order to work only with pairs of tips for which at least one of the elements  $\eta_1, \eta_2$  equals 0. We thus introduce the following definition.

**Definition 5.4.16 (Two-piece ancestor)** Let  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\}$  be a two-piece patch in  $\Gamma_c$ . A two-piece ancestor of this patch is a two-piece patch of the shape  $\{[0, i_1]^*, [\gamma, i_2]^*\} \subset \Gamma_c$  for which there exists  $\delta \in \mathbb{H}_c$  such that

$$\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\} \subset \delta + \mathbf{E}_1^*\{[0, i_1]^*, [\gamma, i_2]^*\}$$
(5.32)

with  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\} \cap (\delta + \mathbf{E}_1^*[0, i_1]^*) \neq \emptyset$  and  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\} \cap (\delta + \mathbf{E}_1^*[\gamma, i_2]^*) \neq \emptyset$ .

In other words, this means that  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\}$  appears in the image of the patch  $\{[0, i_1]^*, [\gamma, i_2]^*\}$  up to a translated vector, and that the two images of  $[0, i_1]^*$  and  $[\gamma, i_2]^*$  both have non-empty intersection with  $\{[\eta_1, j_1]^*, [\eta_2, j_2]^*\}$ . Note that, contrary to the uniquely defined ancestor of a tip (see Definition 5.4.4), a two-piece patch can have several two-piece ancestors. This is due to the freedom given by the translation vector  $\delta$ . Another important remark is that the tips  $[\eta_1, j_1]^*$  and  $[\eta_2, j_2]^*$  in Definition 5.4.16 are not required to be different. The same holds for  $[0, i_1]^*$  and  $[\gamma, i_2]^*$ .

In order to check the tiling condition (5.31), we need to recursively check ancestor relations. We thus define a new graph, namely the *two-piece ancestor graph*. To this end we need a new seed patch which is defined as follows.

**Definition 5.4.17 (Two-piece seed patch)** The *two-piece seed patch*  $W_{\sigma}$  associated with the substitution  $\sigma$  is defined as

$$W_{\sigma} := \left\{ [\gamma, i]^* \in \Gamma_c \mid ||\gamma||_c \le \frac{2M_{\sigma}}{1 - \beta_{\max}} \right\}.$$
(5.33)

**Remark 5.4.18** Note that  $2M_{\sigma}/(1 - \beta_{\max})$  is at least twice as large as the diameter of the tiles  $\mathcal{T}_{\sigma}(i)$ . Thus  $\mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \gamma) \neq \emptyset$  implies that  $[\gamma, j]^* \in W_{\sigma}$ .

Moreover, we represent the pair of tips  $\{[0,k]^*, [\gamma,\ell]^*\}$  as  $[k,\gamma,\ell]^*$ , with  $k \leq \ell$  if  $\gamma = 0$ . The condition  $k \leq \ell$  if  $\gamma = 0$  simply avoids redundancies.

**Definition 5.4.19 (Two-piece ancestor graph)** The set of vertices of the *two-piece ancestor graph* is equal to

$$\{[k,\gamma,\ell]^* \mid (k,\gamma,\ell) \in A \times \mathbb{H}_c \times A, \ [\gamma,\ell]^* \in W_\sigma, \ k \le \ell \text{ if } \gamma = 0\}.$$

There is an edge from  $[j_1, \eta, j_2]^*$  to  $[i_1, \gamma, i_2]^*$  if the patch  $\{[0, i_1]^*, [\gamma, i_2]^*\}$  is a two-piece ancestor of the patch  $\{[0, j_1]^*, [\eta, j_2]^*\}$ .

One checks that each vertex admits at least one outgoing edge. Nevertheless, there might be several outgoing edges.

Following (Siegel and Thuswaldner 2010), the construction of this graph is straightforward when recalling that similar to  $V_{\sigma}$ , the two-piece seed patch  $W_{\sigma}$  can be explicitly computed (see Section 5.4.2). The construction can thus be performed in two steps. One first computes the list of vertices

 $[i_1, \gamma, i_2]^*$  of the graph, based on the computation of the two-piece seed patch  $W_{\sigma}$ . Then, by noticing that  $[\eta, j]^* \subset \delta + \mathbf{E}_1^*([\gamma, i]^*)$  if, and only if, there exists a prefix p of  $\sigma(j)$  such that  $\sigma(j) = pis$  and  $\delta = \eta - h_{\sigma}^{-1}(\gamma + \pi_c \circ \mathbf{P}(p))$ , one checks whether condition (5.32) is satisfied for each pair of vertices  $([j_1, \eta, j_2]^*, [i_1, \gamma, i_2]^*)$ .

We need the following easy lemma.

**Lemma 5.4.20** Let  $\{[\gamma_1, i_1]^*, [\gamma_2, i_2]^*\}$  be a patch in  $\Gamma_c$ . Then at least one of the sets  $\{[0, i_1]^*, [\gamma_2 - \gamma_1, i_2]^*\}$  and  $\{[0, i_2]^*, [\gamma_1 - \gamma_2, i_1]^*\}$  is a patch in  $\Gamma_c$ .

Proof By (5.5), there exists a unique pair vectors  $\{\mathbf{x}_1, \mathbf{x}_2\} \subset \mathbb{Z}^n$  such that  $\gamma_i = \pi_c(\mathbf{x}_i)$ , for i = 1, 2. If  $\langle \mathbf{x}_1, \mathbf{v}_\beta \rangle \leq \langle \mathbf{x}_2, \mathbf{v}_\beta \rangle$  then  $\{[0, i_1]^*, [\gamma_2 - \gamma_1, i_2]^*\}$  is a patch of  $\Gamma_c$ . If the reverse inequality holds,  $\{[0, i_2]^*, [\gamma_1 - \gamma_2, i_1]^*\} \subset \Gamma_c$  and we are done.

We now can state the main result of this section.

**Theorem 5.4.21 (Two-piece ancestor graph tiling condition)** Let  $\sigma$  be a unit Pisot irreducible substitution. The substitution  $\sigma$  satisfies the tiling condition if, and only if, from any vertex in the two-piece ancestor graph, there exists a path to a vertex of the shape  $[i, 0, i]^*$ , for  $i \in A$ .

*Proof* The tiling property is equivalent to (5.31), which is itself equivalent to the following condition: for every two-piece patch  $\{[0, j_1]^*, [\eta, j_2]^*\}$ , there exist  $m \in \mathbb{N}$ ,  $i \in A$  and  $\delta \in \mathbb{H}_c$  such that  $\{[0, j_1]^*, [\eta, j_2]^*\} \subset \delta + \mathbf{E}_1^{*m}([0, i]^*)$ . By the definition of the two-piece ancestor this is equivalent to the fact that for each  $\{[0, j_1]^*, [\eta, j_2]^*\}$  there is  $m \in \mathbb{N}$  and  $i \in A$  such that

$$\{[0, i^*], [0, i^*]\}$$
 is an *m*th ancestor of  $\{[0, j_1^*], [\eta, j_2^*]\}.$  (5.34)

In order to deduce Theorem 5.4.21, it is sufficient to show that we can assume w.l.o.g. that  $[\eta, j_2]^* \in W_{\sigma}$  holds in (5.34).

Suppose on the contrary that  $[\eta, j_2]^* \notin W_{\sigma}$ . Then

$$||\eta||_c > \frac{2M_\sigma}{1 - \beta_{\max}}.$$
(5.35)

There exists a unique set of two elements  $\{[\gamma_1, i_1]^*, [\gamma_2, i_2]^*\} \subset \Gamma_c$  such that

$$[0, j_1]^* \in \mathbf{E}_1^*[\gamma_1, i_1]^*$$
 and  $[\eta, j_2]^* \in \mathbf{E}_1^*[\gamma_2, i_2]^*.$  (5.36)

By Lemma 5.4.20 one of the sets  $\{[0, i_1]^*, [\gamma_2 - \gamma_1, i_2]^*\}, \{[0, i_2]^*, [\gamma_1 - \gamma_2, i_1]^*\}$  is contained in  $\Gamma_c$ . Assume that this is true for the first one (the
second alternative is handled analogously). Then  $\{[0, i_1]^*, [\gamma_2 - \gamma_1, i_2]^*\}$  is a two-piece ancestor of  $\{[0, j_1]^*, [\eta, j_2]^*\}$ . By (5.27) and (5.36) we have

$$||\gamma_1||_c \leq M_\sigma$$
 and  $||\gamma_2||_c \leq \beta_{\max}||\eta||_c + M_\sigma$ 

which implies together with (5.35) that  $||\gamma_2 - \gamma_1||_c \leq \frac{2M_{\sigma}}{1-\beta_{\max}} < ||\eta||_c$ . Thus, arguing in the same way as in the proof of Theorem 5.4.7 we see that there is a positive integer m' such that  $\{[0, j_1]^*, [\eta, j_2]^*\}$  admits an m'th two-piece ancestor  $\{[0, k_1]^*, [\gamma', k_2]^*\}$  which satisfies  $\gamma' \in W_{\sigma}$ . Thus it suffices to assume  $[\eta, j_2]^* \in W_{\sigma}$  in (5.34) and we are done.

# 5.5 Boundary and contact graphs

The tiling condition of Theorem 5.4.21 can be checked in an effective way by constructing the two-piece ancestor graph. However, it turns out that the two-piece ancestor graph is not so easy to handle. Indeed, it can be quite big especially if n is large, and as a second drawback, contrary to the ancestor graph it is not deterministic. We thus introduce two subgraphs of the two-piece ancestor graph, and establish associated tiling conditions, inspired by Proposition 5.4.12 and Theorem 5.4.21.

## 5.5.1 Boundary graphs

We first state as an immediate consequence of Lemma 5.4.20 the following proposition which shows that we only have to consider intersections between the subtiles  $\mathcal{T}(i)$ ,  $i \in A$ , and their neighbours in  $\mathcal{I}_{\sigma}$  (see Definition 5.3.14) to check the tiling property.

**Proposition 5.5.1** The tiling property is satisfied if, and only if,  $\mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \gamma)$  has zero measure for every  $i \in A$  and every  $[\gamma, j]^* \in \Gamma_c$  with  $[\gamma, j]^* \neq [0, i]^*$ .

Proof The tiling property holds if, and only if, for any two distinct tips  $[\gamma, i]^*, [\eta, j]^* \in \Gamma_c$ , we have  $\mu_{n-1}((\mathcal{T}_{\sigma}(i) + \gamma) \cap (\mathcal{T}_{\sigma}(j) + \eta)) = 0$ . But  $(\mathcal{T}_{\sigma}(i) + \gamma) \cap (\mathcal{T}_{\sigma}(j) + \eta)$  is equal, up to a translation, to  $\mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \eta - \gamma)$ , and to  $\mathcal{T}_{\sigma}(j) \cap (\mathcal{T}_{\sigma}(i) + \gamma - \eta)$ . Lemma 5.4.20 implies that either  $[\eta - \gamma, j]^* \in \Gamma_c$  or  $[\gamma - \eta, i]^* \in \Gamma_c$ . This ends the proof.

As a motivation for the definition of the boundary graph (see Definition 5.5.3 below), let us dwell upon the topological information provided by cycles in the ancestor graph.

**Lemma 5.5.2** A vertex  $[\gamma, i]^*$  belongs to a cycle in the ancestor graph if, and only if,  $0 \in \mathcal{T}_{\sigma}(i) + \gamma$ .

Proof We first assume that  $[\gamma, i]^*$  belongs to a cycle of the ancestor graph. Thus, there exists m > 0 such that  $[\gamma, i]^* \in \mathbf{E}_1^{*m}([\gamma, i]^*)$ . In view of (5.17) this implies  $h_{\sigma}^m(\mathcal{T}_{\sigma}(i) + \gamma) \subset \mathcal{T}_{\sigma}(i) + \gamma$ . By iterating this relation and by using the fact that  $h_{\sigma}$  is a contraction, we deduce that  $0 \in \mathcal{T}_{\sigma}(i) + \gamma$ .

Conversely, assume that  $0 \in \mathcal{T}_{\sigma}(i) + \gamma$ . We decompose the tile  $\mathcal{T}_{\sigma}(i) + \gamma$ according to (5.17). Since  $0 \in \mathcal{T}_{\sigma}(i) + \gamma$ , for every  $m \geq 1$  there exists  $[\gamma_m, i_m]^* \in \mathbf{E}_1^{*m}([\gamma, i]^*)$  such that  $0 \in \mathcal{T}_{\sigma}(i_m) + \gamma_m$ . As, by Remark 5.4.6, we see that  $0 \in \mathcal{T}_{\sigma}(i_m) + \gamma_m$  implies that  $\gamma_m \in V_{\sigma}$ , the element  $[\gamma_m, i_m]^*$  is a vertex of the ancestor graph. Thus we get the walk

$$[\gamma_m, i_m]^* \to \cdots \to [\gamma_1, i_1]^* \to [\gamma, i]^*$$

in this graph. By the finiteness of  $V_{\sigma}$ , the sequence  $([\gamma_m, i_m]^*)_{m \ge 1}$  takes twice the same value. Let  $m_1 < m_2$  be such that  $[\gamma_{m_1}, i_{m_1}] = [\gamma_{m_2}, i_{m_2}]$ , with  $m_1 \neq m_2$ . Then,  $[\gamma_{m_1}, i_{m_1}]^*$  is contained in a cycle of this graph. Furthermore, there is a walk from  $[\gamma_{m_1}, i_{m_1}]^*$  to  $[\gamma, i]^*$  in the ancestor graph. Since each vertex in the ancestor graph has a single outgoing edge, this implies that  $[\gamma, i]^*$  belongs to the same cycle of the ancestor graph as  $[\gamma_{m_1}, i_{m_1}]^*$ .

Lemma 5.5.2 indicates that non-emptyness for solutions of a GIFS equation can be deduced from cycles of the related graph. We now apply this idea together with Proposition 5.5.1 to the two-piece ancestor graph.

**Definition 5.5.3 (Boundary graph)** Let  $\sigma$  be a unit Pisot irreducible substitution. The *boundary graph* of  $\sigma$  is the subgraph of the two-piece ancestor graph that contains the vertices  $[i, \gamma, j]^*$  with  $\gamma \neq 0$  or  $i \neq j$ , for  $i, j \in A$ , that belong to a cycle, as well as vertices contained in paths leading away from these cycles.

The motivation for this definition will become clearer in the sketch of the proof of next theorem.

**Theorem 5.5.4 (Boundary graph tiling condition)** Let  $\sigma$  be a unit Pisot irreducible substitution. The tiling condition is satisfied if, and only if, the spectral radius of the boundary graph (that is, the largest eigenvalue of its adjacency matrix) is strictly smaller than the Perron–Frobenius eigenvalue  $\beta$  of  $\mathbf{M}_{\sigma}$ . If this relation holds, the boundary graph provides a GIFS description of the boundary.

*Proof* [Sketch] First note that (5.17) implies, assuming the first alternative of Lemma 5.4.20 (the second one can be handled analogously), that

$$\mathcal{T}_{\sigma}(i_{1}) \cap (\mathcal{T}_{\sigma}(i_{2}) + \gamma) = h_{\sigma} \left( \bigcup_{\substack{[\eta_{1}, j_{1}]^{*} \in \mathbf{E}_{1}^{*}[0, i_{1}]^{*} \\ [\eta_{2}, j_{2}]^{*} \in \mathbf{E}_{1}^{*}[\gamma, i_{2}]^{*}} ((\mathcal{T}_{\sigma}(j_{1}) + \eta_{1}) \cap (\mathcal{T}_{\sigma}(j_{2}) + \eta_{2})) \right)$$
$$= h_{\sigma} \left( \bigcup_{\substack{[0, j_{1}]^{*} \in -\eta_{1} + \mathbf{E}_{1}^{*}[0, i_{1}]^{*} \\ [\eta_{2} - \eta_{1}, j_{2}]^{*} \in -\eta_{1} + \mathbf{E}_{1}^{*}[\gamma, i_{2}]^{*}}} ((\mathcal{T}_{\sigma}(j_{1}) \cap (\mathcal{T}_{\sigma}(j_{2}) + \eta_{2} - \eta_{1})) + \eta_{1}) \right).$$

Since, in view of Remark 5.4.18, the intersections are non-empty only for  $\eta_2 - \eta_1 \in W_{\sigma}$ , the equation can be rewritten as

$$\mathcal{T}_{\sigma}(i_1) \cap (\mathcal{T}_{\sigma}(i_2) + \gamma) = h_{\sigma} \left( \bigcup_{[j_1, \eta, j_2]^* \to [i_1, \gamma, i_2]^*} \left( (\mathcal{T}_{\sigma}(j_1) \cap (\mathcal{T}_{\sigma}(j_2) + \eta)) + \delta \right) \right)$$
(5.37)

where the union is taken over all edges of the two-piece ancestor graph which lead to the vertex  $[i_1, \gamma, i_2]^*$ . Such an intersection is interesting if  $[i_1, \gamma, i_2]^*$ is not of the form  $[i, 0, i]^*$  for some  $i \in A$  since we are only interested in intersections of two different tiles of  $\mathcal{I}_{\sigma}$ . If the union on the right-hand side of (5.37) is empty, *i.e.*, if a vertex of the two-piece ancestor graph has no incoming edge, then also the intersection on the left-hand side is empty. Thus we may successively delete all vertices from the two-piece ancestor graph which have no incoming edges. However, as the two-piece ancestor graph is finite these are vertices which either belong to a cycle or to a path that leads away from a cycle. We may also cancel the vertices of the form  $[i, 0, i]^*$ , for  $i \in A$ . Indeed, an edge from a vertex of the form  $[i, 0, i]^*$  reaches a vertex which is also of the same form  $[j, 0, j]^*$  for some  $j \in A$ . Hence if  $[i_1, \gamma, i_2]^*$  is not of the form  $[i, 0, i]^*$ , then no vertex of this form admits an edge to it. The graph obtained when removing all these vertices is exactly the boundary graph.

Since  $h_{\sigma}$  is a contraction whose application scales down the (n-1)dimensional Lebesgue measure by the factor  $1/\beta$ , one deduces that the intersection  $\mathcal{T}_{\sigma}(i_1) \cap (\mathcal{T}_{\sigma}(i_2) + \gamma)$  has zero measure if the largest eigenvalue of the adjacency matrix of the boundary graph is strictly smaller than  $\beta$ .

The proof of the converse, *i.e.*, that the tiling property implies that the spectral radius of the boundary graph is smaller than  $\beta$  is left as an exercise (see Exercise 5.3).

Note that (5.37) can be regarded as a GIFS equation for the intersections  $\mathcal{T}_{\sigma}(i_1) \cap (\mathcal{T}_{\sigma}(i_2) + \gamma)$ . In view of Proposition 5.5.5 below, this GIFS can be used to describe  $\partial \mathcal{T}_{\sigma}(i)$  if the tiling condition holds, hence the terminology 'boundary graph'. For more details, see (Siegel and Thuswaldner 2010).

**Proposition 5.5.5** Let  $\sigma$  be a unit Pisot irreducible substitution. One has

 $\partial \mathcal{T}_{\sigma}(i) \subseteq \bigcup_{[\gamma,j]^* \neq [0,i]^*, \ [\gamma,j]^* \in \Gamma_c} \mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \gamma),$ 

where equality holds if the tiling property is satisfied.

Proof We fix  $i \in A$ . Let  $\nu \in \partial \mathcal{T}_{\sigma}(i)$  and assume that  $\nu$  is not contained in an intersection of the form  $\mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \gamma)$ , for  $\mathcal{T}_{\sigma}(j) + \gamma \neq \mathcal{T}_{\sigma}(i)$ . Since the tiles  $\mathcal{T}_{\sigma}(j) + \gamma$  are compact, local finiteness implies that the set  $\bigcup_{[\gamma,j]^* \neq [0,i]^*, \ [\gamma,j]^* \in \Gamma_c} (\mathcal{T}_{\sigma}(j) + \gamma)$ ) is a closed set in  $\mathbb{H}_c$ , and thus its complement in  $\mathbb{H}_c$  is an open set. Since the complement contains  $\nu$ , there exists an open neighbourhood B of  $\nu$  which contains only points of  $\mathcal{T}_{\sigma}(i)$ , and no point of other tiles of the self-replicating multiple tiling  $\mathcal{I}_{\sigma}$  of Definition 5.3.14. However, since  $\nu \in \partial \mathcal{T}_{\sigma}(i)$ , there is some point  $\nu' \in B$  which is not contained in  $\mathcal{T}_{\sigma}(i)$ . But then  $\nu'$  is contained in no tile of  $\mathcal{I}_{\sigma}$ , which contradicts the covering property of  $\mathcal{I}_{\sigma}$ . We thus have proved the inclusion.

Assume now that the tiling property holds. Moreover, assume that there is  $\nu \in \operatorname{int}(\mathcal{T}_{\sigma}(i))$  which is contained in  $\mathcal{T}_{\sigma}(j) + \gamma$  for some  $[\gamma, j]^* \in \Gamma_c$ . Then, because  $\mathcal{T}_{\sigma}(i)$  as well as  $\mathcal{T}_{\sigma}(j) + \gamma$  is the closure of its interior (Theorem 5.3.12), there is a  $\nu' \in \operatorname{int}(\mathcal{T}_{\sigma}(i)) \cap \operatorname{int}(\mathcal{T}_{\sigma}(j) + \gamma)$ . Thus, there is a small disk around  $\nu'$  which is covered by both of these tiles, a contradiction to the tiling property.

Note that we deduce the following interesting corollary. We only give a proof of the 'if' implication. For a proof of the converse implication, see (Siegel and Thuswaldner 2010).

**Corollary 5.5.6** Let  $\sigma$  be a unit Pisot irreducible substitution. The substitution  $\sigma$  satisfies the geometric finiteness property if, and only if, 0 is an inner point of the central tile  $\mathcal{T}_{\sigma} = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$ , and 0 belongs to no other tile of the self-replicating tiling.

Proof By Proposition 5.4.12 and by Lemma 5.5.2, the geometric finiteness property implies that 0 belongs to no tile of the form  $\mathcal{T}_{\sigma}(i) + \gamma$  with  $\gamma \neq 0$ and  $i \in A$ . Since the tiling property holds and by Proposition 5.5.5, the boundary of  $\mathcal{T}_{\sigma} = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$  is exactly described by the intersections of  $\mathcal{T}_{\sigma} = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$  with the other tiles of the tiling  $\mathcal{I}_{\sigma}$ . We deduce that 0

does not belong to the boundary of  $\mathcal{T}_{\sigma}(i)$ , and thus that 0 is an inner point of  $\mathcal{T}_{\sigma} = \bigcup_{i \in A} \mathcal{T}_{\sigma}(i)$ .

Remark 5.5.7 Corollary 5.5.6 was already stated in (Akiyama 2002) in the beta-numeration context. The union of cycles in the ancestor graph is called the *zero-expansion graph* in (Siegel and Thuswaldner 2010). Following the proof of Lemma 5.5.2, labels of edges in cycles allow all Dumont– Thomas expansions (see Sections 5.11 and 9.4.2 for more details) that can be obtained for 0 to be computed explicitly, hence the terminology 'zeroexpansion'. Compare also with the notion of *zero automaton* in Chapter 2. The zero-expansion graph allows tiles in  $\mathcal{I}_{\sigma}$  that contain 0 (they are related to cycles in the ancestor graph) to be characterised, and thus, to give a further characterisation of the geometric finiteness property.

Let us end this section with the following statement (given without a proof) that is an analogue of Lemma 5.5.2 and which provides an explicit description of the neighbours of a subtile  $\mathcal{T}_{\sigma}(i)$  with respect to the boundary graph. For a proof, see (Siegel and Thuswaldner 2010).

**Proposition 5.5.8** Let  $\sigma$  be a unit Pisot irreducible substitution. Let  $i \in A$ and let  $[\gamma, j]^* \in \Gamma_c$  with  $\gamma \neq 0$  or  $i \neq j$ . The intersection  $\mathcal{T}_{\sigma}(i) \cap (\mathcal{T}_{\sigma}(j) + \gamma)$ is non-empty if, and only if,  $[i, \gamma, j]^*$  is a vertex of the boundary graph.

#### 5.5.2 Approximations of the boundary and contact graphs

Even if the boundary graph is much smaller than the two-piece ancestor graph, its computation relies on the pre-computation of the two-piece seed patch  $W_{\sigma}$ . However, there is another approach which does not require the pre-computation of this patch. The underlying idea is developed in (Thuswaldner 2006), based on polyhedral approximations of the central tile and its subtiles. Recall that with each tip  $[\gamma, i]^*$  we associate the compact polyhedron  $[\gamma, i]_g^*$  which is obtained by projecting the corresponding face of type *i* by  $\pi_c$  (see Section 5.3.3 for details).

**Definition 5.5.9** Let  $\sigma$  be a unit Pisot irreducible substitution. The *mth* polyhedral approximation of a subtile  $\mathcal{T}_{\sigma}(i)$  is the union of polyhedra given by

$$\mathcal{T}^{(m)}_{\sigma}(i) := \bigcup_{[\gamma,i]^* \in \mathbf{E}_1^{*m}[0,i]^*} h^m_{\sigma}([\gamma,i]^*_g).$$

**Proposition 5.5.10** The Hausdorff limit of the polyhedral approximations  $\mathcal{T}_{\sigma}^{(m)}(i)$  for  $m \to \infty$  is the subtile  $\mathcal{T}_{\sigma}(i)$ .

*Proof* This is a direct consequence of the definition of  $\mathbf{E}_1^*$ , the GIFS equation (5.8) and the uniqueness of the solution of a GIFS. For more details, see (Arnoux and Ito 2001).

**Remark 5.5.11** Figure 5.9 depicts the image of  $\bigcup_{i \in A} \mathcal{T}_{\tau}^{(15)}(i)$  inflated by  $h_{\tau}^{-15}$ , for the substitution  $\tau(1) = 2$ ,  $\tau(2) = 3$ ,  $\tau(3) = 12$ . We have seen in Example 5.4.13 that  $\sigma$  does not satisfy the geometric finiteness property. Note that Figure 5.9 gives some indication of the fact that 0 (which is dotted) is not an interior point of the central tile.

**Definition 5.5.12 (Contact graph)** Let  $\sigma$  be a unit Pisot irreducible substitution. A 0 *level vertex* of the two-piece ancestor graph is a vertex of the form  $[i, \gamma, j]^*$ , such that the ((n - 1)-dimensional) geometric tips  $[0, i]_g^*$  and  $[\gamma, j]_g^*$  intersect on exactly one face of dimension n - 2. The *contact graph* is the subgraph of the two-piece ancestor graph whose vertices are the 0 level vertices together with all vertices that can be reached by a path in the two-piece ancestor graph starting at a 0 level vertex.

The contact graph is described in (Thuswaldner 2006). Among the graphs considered for tiling criteria so far, the contact graph is the one which is easiest to compute. Indeed, it can be computed with a recursive procedure that always terminates and does not require the computation of the two-piece seed patch  $W_{\sigma}$ . We first compute the set of 0 level vertices. For  $(i, j) \in A^2$ , we take the triple  $[i, \pi_c(\mathbf{e}_j), i]^*$  if  $\langle \mathbf{v}_{\beta}, \mathbf{e}_j \rangle < \langle \mathbf{v}_{\beta}, \mathbf{e}_i \rangle$ , or  $[i, \pi_c(\mathbf{e}_j - \mathbf{e}_i), j]^*$ , otherwise. Note that this proves that the tips coming from a 0 level vertex belong to  $W_{\sigma}$ . We then construct recursively the contact graph, by computing successors (in the two-piece ancestor graph) of already computed vertices. The choice of the initial set implies that such a vertex always belongs to the two-piece ancestor graph (note that these computations do not require the knowledge of the two-piece ancestor graph; we only use the definition of its edges to compute the successors). The construction ends when no additional edge can be added to the graph, which always happens according to (Thuswaldner 2006).

It turns out that the contact graph provides also a suitable decomposition of the boundary of the central tile, which in turn provides a very simple tiling condition. For a complete proof, see (Thuswaldner 2006) together with (Siegel and Thuswaldner 2010).

**Theorem 5.5.13 (Contact graph tiling condition)** Let  $\sigma$  be a unit Pisot irreducible substitution. The tiling property is satisfied if, and only if, the spectral radius of the contact graph is strictly smaller than the Perron– Frobenius eigenvalue  $\beta$  of  $\mathbf{M}_{\sigma}$ .

Proof [Sketch] In (Thuswaldner 2006) it is proved that all overlaps  $(\mathcal{T}_{\sigma}(k_1) + \gamma_1) \cap (\mathcal{T}_{\sigma}(k_2) + \gamma_2)$  of the self-replicating multiple tiling  $\mathcal{I}_{\sigma}$  are translations of overlaps of the form  $\mathcal{T}_{\sigma}(i_1) \cap (\mathcal{T}_{\sigma}(i_1) + \gamma)$ , where  $[i_1, \gamma, i_2]^*$  is a vertex of the contact graph. Note that the proof heavily relies on the polyhedral tiling of Proposition 5.3.3. Moreover, (Thuswaldner 2006) shows that these intersections are the solution of the GIFS equation

$$\mathcal{T}_{\sigma}(i_{1}) \cap (\mathcal{T}_{\sigma}(i_{2}) + \gamma) = \bigcup_{[j_{1},\eta,j_{2}]^{*} \to [i_{1},\gamma,i_{2}]^{*}} ((\mathcal{T}_{\sigma}(j_{1}) \cap (\mathcal{T}_{\sigma}(j_{2}) + \eta)) + \delta)$$
(5.38)

where the union is taken over all edges of the contact graph which lead to the vertex  $[i_1, \gamma, i_2]^*$ . The proof can now be finished in the same way as the proof of Theorem 5.5.4.

This tiling condition is definitely the simplest tiling condition that we have considered so far since the contact graph can be computed recursively without the precomputation of the two-piece seed patch  $W_{\sigma}$ .

It also provides a 'minimal' GIFS for the boundary of the central tile in the sense that it removes from the boundary graph all the intersections that are redundant, *i.e.*, that are included in other intersections. For details and examples of contact graphs for families of substitutions associated with beta-numeration we refer again to (Thuswaldner 2006).

#### 5.6 Geometric coincidences

We consider now a further set of conditions each of which is equivalent to the tiling property. Using the concept of duality, they can be expressed in terms of the tiling of the expanding line.

### 5.6.1 Strands and duality

Up to now, we have worked with the set  $\Gamma_c$  consisting of tips which correspond to projections of faces by  $\pi_c$ . The notion of tip has allowed us to define the GIFS substitution  $\mathbf{E}_1^*$  (see Definition 5.3.8) inspired by the action of the GIFS equations (5.8) that govern the subtiles  $\mathcal{T}_{\sigma}(i)$ , for  $i \in A$ . The tiling property (see Definition 5.3.15) was then expressed in terms of pre-images of tips under  $\mathbf{E}_1^*$  (see *e.g.* Theorem 5.4.21).

We now wish to adopt a dual one-dimensional viewpoint. Instead of working with faces of hypercubes, we will work with line segments, and the projection  $\pi_e$  will play the role of the projection  $\pi_c$ . We consider formal strands to represent stairs joining points in the integer grid  $\mathbb{Z}^n$ . An element of the form  $(\mathbf{x}, i) \in \mathbb{Z}^n \times A$  is called a *basic formal strand*. In the sequel, we use the notation  $[\mathbf{x}, i]$  instead of  $(\mathbf{x}, i)$  for this object. The *basic geometric strand*  $[\mathbf{x}, i]_g$  is defined as the segment connecting  $\mathbf{x}$  with  $\mathbf{x} + \mathbf{e}_i$ . A *formal strand* is then defined as a union of basic formal strands. We similarly define a *geometric strand*.

Note that there is no more '\*' in the notation for strands:  $[\mathbf{y}, j]_g$  represents a segment in  $\mathbb{R}^n$  while  $[\gamma, i]_g^*$  is the projection in  $\mathbb{H}_c$  of a face. Faces of hypercubes and segments can be considered as *dual* in the sense of the duality principle of linear algebra (see (Arnoux and Ito 2001)). Note that in (Sano, Arnoux, and Ito 2001) a Poincaré type duality between generalisations of faces and segments is established. As we shall see in the sequel, this duality will allow us to translate properties of the set  $\Gamma_c$  into properties of the self-similar translation set  $\Gamma_e$ , and thus to work with the tiling  $\mathcal{E}_u$  of the expanding line.

If we consider a word  $w \in A^*$ , the point  $\mathbf{P}(w) \in \mathbb{Z}^n$  is the abelianisation of w and one builds in a natural way a formal strand and a geometric strand from **0** to  $\mathbf{P}(w)$  by simply reading the letters in w. Strands allow one to keep track of the combinatorics of a word that would be lost in the abelianisation process.

We now extend the action of  $\sigma$  to unions of basic strands  $[\mathbf{x}, i] \in \mathbb{Z}^n \times A$ , according to the formalism of (Arnoux and Ito 2001).

**Definition 5.6.1 (Geometric realisation of a substitution)** Let  $\sigma$  be a unit Pisot irreducible substitution. The *one-dimensional geometric reali*sation of  $\sigma$  is defined on the sets of formal strands by

$$\mathbf{E}_1\{[\mathbf{y}, j]\} = \bigcup_{(p, i, s, ), \sigma(j) = pis} \{[\mathbf{M}_{\sigma}\mathbf{y} + \mathbf{P}(p), i]\},$$
$$\mathbf{E}_1(Y_1 \cup Y_2) = \mathbf{E}_1(Y_1) \cup \mathbf{E}_1(Y_2).$$

We also use here the notation  $\mathbf{E}_1[\mathbf{y}, j]$  for  $\mathbf{E}_1\{[\mathbf{y}, j]\}$ . With this formalism at hand, the set  $\bigcup_{k\geq 0} \mathbf{E}_1^k[\mathbf{0}, u_0]$  generates the broken line  $L_u$ , by replacing formal strands by geometric strands. This implies that the broken line  $L_u$ is invariant under the action of  $\mathbf{E}_1$ .

The following lemma enhances the relation between  $\mathbf{E}_1$  and the GIFS substitution  $\mathbf{E}_1^*$  acting on the translation set  $\Gamma_c$ .

Lemma 5.6.2 (Duality lemma) Let  $\mathbf{x}, \mathbf{y} \in \mathbb{Z}^n$ . Then

$$[\pi_c(\mathbf{y}), j]^* \in \mathbf{E}_1^*([\pi_c(\mathbf{x}), i]^*) \iff [-\mathbf{x}, i] \in \mathbf{E}_1([-\mathbf{y}, j]).$$

Proof Let  $\mathbf{x}, \mathbf{y} \in \mathbb{Z}^n$ . Then  $[\pi_c(\mathbf{y}), j]^* \in \mathbf{E}_1^*([\pi_c(\mathbf{x}), i]^*)$  if, and only if, there exists p such that  $\sigma(j) = pis$  and  $\pi_c(\mathbf{y}) = h_{\sigma}^{-1}(\pi_c(\mathbf{x}) + \pi_c(\mathbf{P}(p))) =$  $\pi_c(\mathbf{M}_{\sigma}^{-1}\mathbf{x} + \mathbf{P}(p))$ . Equivalently by (5.5) we have  $\mathbf{y} = \mathbf{M}_{\sigma}^{-1}(\mathbf{x} + \mathbf{P}(p))$ , *i.e.*,  $-\mathbf{x} = \mathbf{M}_{\sigma}(-\mathbf{y}) + \mathbf{P}(p)$ , which gives  $[-\mathbf{x}, i] \in \mathbf{E}_1([-\mathbf{y}, j])$ .

By defining suitable vector spaces on strands and tips,  $\mathbf{E}_1^*$  and  $\mathbf{E}_1$  are linked to each other by the duality principle of linear algebra (up to a reverse of the orientation of the space that leads to introduce the '-' sign in the statement of Lemma 5.6.2). This is worked out in (Arnoux and Ito 2001). As a geometric interpretation, we can also say that the broken line  $L_u$  and the stepped hyperplane are dual to each other.

From Lemma 5.6.2 and the definition of the self-replicating translation set  $\Gamma_e$ , we deduce the following dictionary. The proof is immediate.

**Lemma 5.6.3 (Dictionary)** Let  $\sigma$  be a unit Pisot irreducible substitution. The following assertions are true.

- (i) Ancestor. A tip [π<sub>c</sub>(**x**), i]\* is the ancestor of [π<sub>c</sub>(**y**), j]\* if, and only if, [-**x**, i] is a segment of the strand E<sub>1</sub>[-**y**, j].
- (ii) Common ancestor. Two tips [π<sub>c</sub>(y<sub>1</sub>), j<sub>1</sub>]\* and [π<sub>c</sub>(y<sub>2</sub>), j<sub>2</sub>]\* have a common ancestor [π<sub>c</sub>(x), i]\* if, and only if, the strands E<sub>1</sub>[-y<sub>1</sub>, j<sub>1</sub>] and E<sub>1</sub>[-y<sub>2</sub>, j<sub>2</sub>] both contain the segment [-x, i].
- (iii) Two-piece ancestor Equation (5.32). There exists  $\delta$  such that  $\{[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_1), j_2]^*\} \subset \delta + \mathbf{E}_1^*\{[0, i_1]^*, [\pi_c(\mathbf{x}), i_2]^*\}$  if, and only if, there exists  $\mathbf{z}$  such that  $\mathbf{z} + \{[\mathbf{0}, i_1], [-\mathbf{x}, i_2]\} \subset \mathbf{E}_1\{[-\mathbf{y}_1, j_1], \mathbf{E}_1[-\mathbf{y}_2, j_2]\}.$
- (iv) Two-piece patch ancestor. There exists an element  $\delta$  such that  $\{[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_2), j_2]^*\} \subset \delta + \mathbf{E}_1^*([0, i]^*)$  if, and only if, there exists  $\mathbf{z}$  such that  $[\mathbf{z}, i] \in \mathbf{E}_1[-\mathbf{y}_1, j_1] \cap \mathbf{E}_1[-\mathbf{y}_2, j_2]$ .

### 5.6.2 Super coincidence condition

The relations established in Section 5.6.1 naturally lead to the following definition which extends the notion of combinatorial strong coincidence introduced in Definition 5.2.9. It can be considered as the dual property of having a common ancestor. This condition was first defined in (Ito and Rao 2006).

Definition 5.6.4 (Geometric strong coincidence) We say that the

basic strands  $[\mathbf{y}_1, j_1]$  and  $[\mathbf{y}_2, j_2]$  have geometric strong coincidence if there exists a positive integer such that  $\mathbf{E}_1^N[\mathbf{y}_1, j_1]$  and  $\mathbf{E}_1^N[\mathbf{y}_2, j_2]$  have at least one basic formal strand in common.

Note that the combinatorial strong coincidence condition of Definition 5.2.9 is equivalent to the fact that  $[\mathbf{0}, j_1]$  and  $[\mathbf{0}, j_2]$  have geometric strong coincidence for each pair  $(j_1, j_2) \in A^2$ . Indeed, assume that  $\mathbf{E}_1^N[\mathbf{0}, j_1]$ and  $\mathbf{E}_1^N[\mathbf{0}, j_2]$  have one basic formal strand in common, say  $[\mathbf{x}, i]$ . This is equivalent to the existence of the decompositions  $\sigma^N(j_1) = p_1 i s_1$  and  $\sigma^N(j_2) = p_2 i s_2$ , with  $\pi_e(\mathbf{x}) = \pi_e \circ \mathbf{P}(p_1) = \pi_e \circ \mathbf{P}(p_2)$ . We deduce from (5.5) that  $\mathbf{P}(p_1) = \mathbf{P}(p_2)$ .

As we shall see, one checks that a two-piece patch  $\{[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_2), j_2]^*\}$  is a patch of  $\Gamma_c$  up to translation (*i.e.*, there exists  $\delta$  such that  $\{[\pi_c(\mathbf{y}_1) + \delta, j_1]^*, [\pi_c(\mathbf{y}_2) + \delta, j_2]^*\} \subset \Gamma_c)$  if, and only if, the projections of the segments  $[-\mathbf{y}_1, j_1]_g$  and  $[-\mathbf{y}_2, j_2]_g$  along the expanding direction intersect on a non-degenerate interval. Before we make this more precise, we introduce the following definition, due to (Barge and Kwapisz 2006).

**Definition 5.6.5 (Height)** We say that the basic strands  $[\mathbf{x}, i]$  and  $[\mathbf{y}, j]$  have the same height if

$$\operatorname{int}(\pi_e([\mathbf{x},i])_g) \cap \operatorname{int}(\pi_e([\mathbf{y},j]_g)) \neq \emptyset.$$

**Lemma 5.6.6** Let  $\mathbf{x}, \mathbf{y} \in \mathbb{Z}^n$  such that  $\langle \mathbf{x}, \mathbf{v}_\beta \rangle \leq \langle \mathbf{y}, \mathbf{v}_\beta \rangle$ . Then for each  $i \in A$  the following assertions are equivalent.

- The tip  $[\pi_c(-\mathbf{x}+\mathbf{y}), i]^*$  belongs to  $\Gamma_c$ .
- For each  $j \in A$ , the basic strands  $[\mathbf{x}, i]$  and  $[\mathbf{y}, j]$  have the same height.

*Proof* Let  $[\mathbf{x}, i]$  and  $[\mathbf{y}, j]$  be two formal strands having the same height with  $\langle \mathbf{x}, \mathbf{v}_{\beta} \rangle \leq \langle \mathbf{y}, \mathbf{v}_{\beta} \rangle$ . By (5.4) we have that

$$\langle \mathbf{x}, \mathbf{v}_eta 
angle \leq \langle \mathbf{y}, \mathbf{v}_eta 
angle < \langle \mathbf{x} + \mathbf{e}_i, \mathbf{v}_eta 
angle.$$

This implies that  $0 \leq \langle -\mathbf{x}+\mathbf{y}, \mathbf{v}_{\beta} \rangle < \langle \mathbf{e}_i, \mathbf{v}_{\beta} \rangle$ . We thus deduce that  $[\pi_c(-\mathbf{x}+\mathbf{y}), i]^* \in \Gamma_c$ . The proof of the converse implication follows along the same lines.

**Definition 5.6.7 (Super coincidence condition)** A unit Pisot irreducible substitution  $\sigma$  satisfies the *super coincidence condition* if any two basic strands  $[\mathbf{x}, i]$  and  $[\mathbf{y}, j]$  have geometric strong coincidence whenever they have the same height.



Fig. 5.12. The super coincidence condition.

An illustration of the notion of super coincidence is given in Figure 5.12.

According to (Barge and Kwapisz 2006, Ito and Rao 2006), the tiling condition of Corollary 5.4.15 can be restated as follows.

**Theorem 5.6.8 (Ito and Rao 2006)** A unit Pisot irreducible substitution  $\sigma$  satisfies the super coincidence condition if, and only if, the tiling property is satisfied.

*Proof* By Corollary 5.4.15, the tiling property is satisfied if, and only if, for any  $[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_2), j_2]^* \in \Gamma_c$  there is  $\delta \in \mathbb{H}_c$  and  $i \in A$  such that

$$\{[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_2), j_2]^*\} \subset \delta + \mathbf{E}_1^*([0, i]^*).$$

Lemma 5.6.3 (iv) implies that this is equivalent to the fact that there exists  $\mathbf{z}$  such that  $[\mathbf{z}, i] \in \mathbf{E}_1[-\mathbf{y}_1, j_1] \cap \mathbf{E}_1[-\mathbf{y}_2, j_2]$ . Thus, the tiling property is satisfied if, and only if,  $[\mathbf{y}_1, j_1]$  and  $[\mathbf{y}_2, j_2]$  have geometric strong coincidence for any  $[\pi_c(\mathbf{y}_1), j_1]^*, [\pi_c(\mathbf{y}_2), j_2]^* \in \Gamma_c$ . In view of Lemma 5.4.20, we may assume w.l.o.g. that  $[\pi_c(\mathbf{y}_1) - \pi_c(\mathbf{y}_2), j_1]^* \in \Gamma_c$ . Since  $[\mathbf{y}_1, j_1]$  and  $[\mathbf{y}_2, j_2]$  have geometric strong coincidence if, and only if,  $[\mathbf{y}_1 - \mathbf{y}_2, j_1]$  and  $[\mathbf{0}, j_2]$  have geometric strong coincidence, the above assertion is equivalent to the fact that  $[\mathbf{y}, j_1]$  and  $[\mathbf{0}, j_2]$  have geometric strong coincidence if  $[\pi_c(\mathbf{y}), j_1]^* \in \Gamma_c$ . By Lemma 5.6.6 this is equivalent to the fact that  $[\mathbf{y}, j_1]$  and  $[\mathbf{0}, j_2]$  have the same height. However, as  $[\mathbf{y}_1, j_1]$  and  $[\mathbf{y}_2, j_2]$  have the same height if, and only if, the same is true for  $[\mathbf{y}_1 - \mathbf{y}_2, j_1]$  and  $[\mathbf{0}, j_2]$ , this is equivalent to the super coincidence condition.

We now define a graph that turns out to be isomorphic to the two-piece ancestor graph. This graph is described in (Barge and Kwapisz 2006).

Similarly as in Section 5.4.3 we introduce the notation  $[i_1, \mathbf{x}, i_2]$  for triples  $(i_1, \mathbf{x}, i_2) \in A \times \mathbb{Z}^n \times A$  with  $[-\pi_c(\mathbf{x}), i_2]^* \in W_\sigma$  where we assume that  $i_1 \leq i_2$  if  $\mathbf{x} = \mathbf{0}$  to avoid redundancies. The triple  $[i_1, \mathbf{x}, i_2]$  represents the pair of tips  $[\mathbf{0}, i_1]$  and  $[-\mathbf{x}, i_2]$  having the same height by Lemma 5.6.6.

**Definition 5.6.9 (Configuration graph)** The set of vertices of the *configuration graph* is equal to

$$\{[i_1, \mathbf{x}, i_2] \mid (i_1, \mathbf{x}, i_2) \in A \times \mathbb{Z}^n \times A, \ [-\pi_c(\mathbf{x}), i_2]^* \in W_\sigma, \ i_1 \le i_2 \text{ if } \mathbf{x} = \mathbf{0}\}.$$

There is an edge from  $[j_1, \mathbf{y}, j_2]$  to  $[i_1, \mathbf{x}, i_2]$  if there exists  $\mathbf{z} \in \mathbb{Z}^n$  such that

$$\mathbf{z} + \{[\mathbf{0}, i_1], [\mathbf{x}, i_2]\} \subset \mathbf{E}_1\{[\mathbf{0}, j_1], [\mathbf{y}, j_2]\}$$

with  $\mathbf{z} + [\mathbf{0}, i_1] \cap \mathbf{E}_1\{[\mathbf{0}, j_1], [\mathbf{y}, j_2]\} \neq \emptyset$  and  $\mathbf{z} + [\mathbf{x}, i_2] \cap \mathbf{E}_1\{[\mathbf{0}, j_1], [\mathbf{y}, j_2]\} \neq \emptyset$ .

Using the duality statements of Lemma 5.6.3 we obtain the following isomorphic graphs.

**Proposition 5.6.10** The configuration graph and the two-piece ancestor graph are isomorphic.

Proof By definition,  $[i_1, \mathbf{x}, i_2]$  is a vertex of the configuration graph if, and only if,  $[i_1, -\pi_c(\mathbf{x}), i_2]^*$  is a vertex of the two-piece ancestor graph. By (5.15), we thus get a one-to-one correspondence between the sets of vertices of each graph. Let us consider now edges. According to Lemma 5.6.3 (iii), there exists  $\delta \in \mathbb{H}_c$  such that

$$\{[0, j_1]^*, [\pi_c(\mathbf{y}), j_2]^*\} \subset \delta + \mathbf{E}_1^* \{[0, i_1]^*, [\pi_c(\mathbf{x}), i_2]^*\}$$

with

$$\{[0, j_1]^*, [\pi_c(\mathbf{y}), j_2]^*\} \cap (\delta + \mathbf{E}_1^*([0, i_1]^*)) \neq \emptyset \quad \text{and} \\ \{[0, j_1]^*, [\pi_c(\mathbf{y}), j_2]^*\} \cap (\delta + \mathbf{E}_1^*([\pi_c(\mathbf{x}), i_2]^*)) \neq \emptyset$$

if, and only if, there exists  $\mathbf{z} \in \mathbb{Z}^n$  such that

$$\mathbf{z} + \{[\mathbf{0}, i_1], [-\mathbf{x}, i_2]\} \subset \mathbf{E}_1\{[\mathbf{0}, j_1], [-\mathbf{y}, j_2]\}$$

with

$$(\mathbf{z} + [\mathbf{0}, i_1]) \cap \mathbf{E}_1\{[\mathbf{0}, j_1], [-\mathbf{y}, j_2]\} \neq \emptyset \quad \text{and}$$
$$(\mathbf{z} + [-\mathbf{x}, i_2]) \cap \mathbf{E}_1\{[\mathbf{0}, j_1], [-\mathbf{y}, j_2]\} \neq \emptyset.$$

Hence, also the edges of the configuration graph are in one-to-one correspondence with the edges of the two-piece ancestor graph .  $\hfill\square$ 

The following theorem, which has been first proved in (Barge and Kwapisz 2006, Proposition 17.1), is a direct consequence of Proposition 5.6.10 and Theorem 5.4.21.

**Theorem 5.6.11 (Barge and Kwapisz 2006, Proposition 17.1)** A unit Pisot irreducible substitution  $\sigma$  satisfies the tiling property if, and only if, from every vertex in the configuration graph, there exists a path to a vertex of the shape [k, 0, k].

### 5.7 Overlap coincidences

We are now going to introduce a new viewpoint on configuration graphs and on the super coincidence condition, following the concept of *overlap coincidence*. Overlap coincidence was first used in (Solomyak 1997) for two-dimensional tilings, and later extended to one-dimensional substitution tilings in (Sirvent and Solomyak 2002). As we shall see, this framework allows a graph to be defined that is related to the configuration graph (and the two-piece ancestor graph). Moreover, it provides a simple combinatorial algorithm which allows the tiling property to be decided. In particular, this algorithm avoids having to compute the two-piece seed patch  $W_{\sigma}$ .

# 5.7.1 Definitions

So far we have considered pairs of basic strands in  $\mathbb{R}^n$  with the same height and checked whether a common basic strand occurs under iterations of  $\mathbf{E}_1$ . This is the super coincidence condition. By projecting basic strands of the same height by  $\pi_e$ , one recovers intersecting segments, called *overlaps*. The viewpoint used here is to work directly with such intersections. We do not consider all pairs of basic strands with the same height, but we restrict ourselves to basic strands which occur in some translated copies of the broken line  $L_u$ .

In order to define suitable translation vectors for the translated copies of  $L_u$ , we introduce the set of all possible distances between two tiles of the same type in the self-similar tiling of the expanding line  $\mathcal{E}_u = \{\pi_e([\mathbf{x}, i]_g) \mid [\mathbf{x}, i] \in \Gamma_e\}$ . Since tiles in the tiling  $\mathcal{E}_u$  are ordered according to the fixed point  $u = u_0 u_1 u_2 \cdots$ , the set of distances between tiles is described by

$$\Xi(\mathcal{E}_u) = \{\lambda \in \mathbb{R}_+ \mid \lambda = \pi_e \circ \mathbf{P}(u_N \cdots u_{N+m-1}), \ N, m \ge 0, \ u_N = u_{N+m} \}$$
$$= \{\lambda \in \mathbb{R}_+ \mid \exists T, T' \in \mathcal{E}_u, \ T' = T + \lambda \}.$$

One has  $\Xi(\mathcal{E}_u) \subseteq \mathbb{Z}[\beta]$ , since each coordinate of  $\mathbf{v}_\beta$  belongs to  $\mathbb{Z}[\beta]$ . Note that we get  $\beta \equiv (\mathcal{E}_u) \subset \Xi(\mathcal{E}_u)$  by using the invariance of u under  $\sigma$ .

We first introduce the notion of overlap which corresponds to the intersection of the projections of basic geometric strands of the same height, following (Sirvent and Solomyak 2002). Let S, T be two tiles occurring in the self-similar tiling  $\mathcal{E}_u$ . Note that S and T are intervals. For  $\lambda \in \mathbb{R}_+$ , the triple  $(T, S, \lambda)$  is called *overlap* if  $\operatorname{int}(T \cap (S - \lambda)) \neq \emptyset$ . See Figures 5.13 and 5.14 for an illustration.

We now restrict ourselves to sets of pairs of tiles T, S in the self-similar tiling  $\mathcal{E}_u$  that are separated by a length belonging to  $\Xi(\mathcal{E}_u)$ , *i.e.*,

$$\mathcal{O}_u = \{ (T, S, \lambda) \mid \operatorname{int}(T \cap (S - \lambda)) \neq \emptyset, T, S \in \mathcal{E}_u, \lambda \in \Xi(\mathcal{E}_u) \}.$$

The reason for this restriction for the set of lengths  $\lambda$  will become clear in Section 5.7.2.



Fig. 5.13. In order to illustrate the relation between the tiling  $\mathcal{E}_u$  and the broken lines in  $\mathbb{R}^n$  we draw the tilings  $\mathcal{E}_u$  and  $\mathcal{E}_u - \lambda$  parallel to the expanding eigendirection of  $M_\sigma$  and not in the real line for  $\sigma(1) = 112$ ,  $\sigma(2) = 21$ .



Fig. 5.14. An example of overlaps of a tiling in the real line for  $\sigma(1)=112,$   $\sigma(2)=21.$ 

The overlaps contained in  $\mathcal{O}_u$  can be built in a quite simple geometric way. Indeed, we consider the tiling  $\mathcal{E}_u$ , and look at the distance  $\lambda$  between two tiles of the same type in  $\mathcal{E}_u$ . We then shift the tiling  $\mathcal{E}_u$  by  $\lambda$ , and take the new tiling by intervals  $(\mathcal{E}_u - \lambda) \cap \mathbb{R}_+$ , which is not 'synchronised' with  $\mathcal{E}_u$ , *i.e.*, endpoints of tiles in each tiling do not correspond. We thus define the synchronised tiling associated with two tilings by intervals as the tiling by intervals obtained when taking the union of the set of endpoints of both tilings. We use the notation  $\mathcal{E} \cap \mathcal{E}'$  for the synchronised tiling associated with  $\mathcal{E}$  and  $\mathcal{E}'$ . We say that we synchronise two tilings when we take their synchronised tiling. Synchronising the two tilings  $\mathcal{E}_u$  and  $\mathcal{E}_u - \lambda$  thus creates new smaller tiles, corresponding to overlaps in  $\mathcal{O}_u$ . In more geometric terms, we work with the projections of  $L_u$  and  $L_u - \mathbf{z}$ , with  $\pi_e(\mathbf{z}) = \lambda$  (see again Figures 5.13 and 5.14).

The number of overlaps in the pair  $(\mathcal{E}_u, (\mathcal{E}_u - \lambda) \cap \mathbb{R}^+)$  of self-similar tilings is infinite. We now classify them, up to a translation vector, according to the following equivalence relation: two overlaps  $(T, S, \lambda), (T', S', \lambda')$  are said to be *equivalent* if there exists  $\delta \in \mathbb{R}$  such that  $T' = T + \delta, S' - \lambda' = S - \lambda + \delta$ . The equivalence class of  $(T, S, \lambda)$  is denoted by  $[T, S, \lambda], i.e,$ 

$$[T, S, \lambda] = \{ (T', S', \lambda') \in \mathcal{O}_u \mid T' = T + \delta, S' - \lambda' = S - \lambda + \delta, \delta \in \mathbb{R} \}.$$

The set of equivalence classes of elements in  $\mathcal{O}_u$  is denoted by  $[\mathcal{O}_u]$ . Let us note that  $[\mathcal{O}_u]$  obviously depends on the substitution  $\sigma$ . However, it is independent of the choice of the fixed point u (see Exercise 5.5).

**Remark 5.7.1** An equivalence class  $[T, S, \lambda]$  is clearly determined by the type of T, the type of S and the difference  $\nu$  of the starting points of T and  $S - \lambda$ .

For a fixed  $\lambda \in \Xi(\mathcal{E}_u)$  let  $\mathcal{O}_u(\lambda)$  be the set of all overlaps  $(T, S, \lambda)$  with  $S, T \in \mathcal{E}_u$  and denote by  $[\mathcal{O}_u(\lambda)]$  the corresponding set of equivalence classes. We introduce the following terminology.

- A class  $[T, S, \lambda] \in [\mathcal{O}_u]$  is called an *overlap class*.
- An overlap  $(T, S, \lambda)$  is a *coincidence overlap* if  $T = S \lambda$ . This notion extends to overlap classes.
- An overlap  $(T, S, \lambda)$  is a half-coincidence overlap if T and  $S \lambda$  have at least one common endpoint. This notion also extends to overlap classes.

A coincidence overlap class is *a fortiori* a half-coincidence overlap class.

**Lemma 5.7.2** Let  $\sigma$  be a unit Pisot irreducible substitution. The set  $[\mathcal{O}_u]$  of overlap classes is finite.

Note that the proof of this lemma uses similar arguments as, for instance, the proof of Proposition 2.3.33 and Lemma 2.4.7.

**Proof** Let  $(T, S, \lambda)$  be an overlap. There exist basic geometric strands  $[\mathbf{x}, i]_g$ ,  $[\mathbf{y}, j]_g$  in  $L_u$  such that  $T = \pi_e([\mathbf{x}, i]_g)$ ,  $S = \pi_e([\mathbf{y}, j]_g)$ . Let  $\nu = \pi_e(\mathbf{y} - \mathbf{x}) - \lambda$  be the difference of the starting points of T and  $S - \lambda$ . In view of Remark 5.7.1 (note that there are finitely many types of tiles), it suffices to show that only finitely many choices of  $\nu$  are possible. We have

$$T \cap (S - \lambda) = \pi_e(\mathbf{x}) + (\pi_e[\mathbf{0}, i]_g \cap (\pi_e[\mathbf{0}, j]_g + \nu)).$$

Note that  $(T, S, \lambda)$  is an overlap, if, and only if,  $-\pi_e(\mathbf{e}_j) < \nu < \pi_e(\mathbf{e}_i)$ .

Furthermore, since  $\lambda \in \Xi(\mathcal{E}_u)$ , there exists  $\mathbf{z} \in \mathbb{Z}^n$  such that  $\lambda = \langle \mathbf{z}, \mathbf{v}_\beta \rangle$ . Recall that we have assumed that the coordinates of  $\mathbf{v}_\beta$  all belong to  $\mathbb{Z}[\beta]$ . This implies that  $\nu = \langle \mathbf{y} - \mathbf{x} - \mathbf{z}, \mathbf{v}_\beta \rangle \in \mathbb{Z}[\beta]$ , and that the Galois conjugates of  $\nu$  are given by  $\langle \mathbf{y} - \mathbf{x} - \mathbf{z}, \mathbf{v}_\beta \rangle \in \mathbb{Z}[\beta]$ , and that the Galois conjugates of  $\nu$  are given by  $\langle \mathbf{y} - \mathbf{x} - \mathbf{z}, \mathbf{v}_\beta \rangle \in \mathbb{Z}[\beta]$ , we shall prove that these Galois conjugates are uniformly bounded. As they are the coordinates of  $\pi_c(\mathbf{y} - \mathbf{x} - \mathbf{z})$  in the basis  $(\mathbf{u}_{\beta^{(i)}})_{i\geq 2}$ , we have to show the boundedness of  $\pi_c(\mathbf{y} - \mathbf{x} - \mathbf{z})$ . Since  $[\mathbf{x}, i]_g$  and  $[\mathbf{y}, j]_g$  are segments  $L_u$ , we have  $\pi_c(\mathbf{x}), \pi_c(\mathbf{y}) \in \mathcal{T}_\sigma$ . Thus Theorem 5.2.3 yields that  $\pi_c(\mathbf{x})$  and  $\pi_c(\mathbf{y})$  are uniformly bounded. Moreover, as  $\pi_e(\mathbf{z}) \in \Xi(\mathcal{E}_u)$ , we get that  $\pi_c(\mathbf{z}) \in \mathcal{T}_\sigma - \mathcal{T}_\sigma$ . Using Theorem 5.2.3 again, this implies that  $\pi_c(\mathbf{z})$  is uniformly bounded, too. Therefore  $\nu$  and all its Galois conjugates are uniformly bounded. Since  $\nu \in \mathbb{Z}[\beta]$ , there are only finitely many possibilities for  $\nu$ .

**Remark 5.7.3** This proof yields a strong relation between overlaps and tips: we have proved that with each overlap class  $[T, S, \lambda]$  we associate the pair of basic strands  $([0, i], [\mathbf{z}, j])$  that have the same height, with  $\mathbf{z} \in \mathbb{Z}^n$  being uniquely determined by  $\pi_e(\mathbf{y} - \mathbf{x}) - \lambda = \pi_e(\mathbf{z})$ , where  $T = \pi_e([\mathbf{x}, i]_g)$ ,  $S = \pi_e([\mathbf{y}, j]_g)$ . Note that the position of the basic strand  $[\mathbf{z}, j]$  is very specific.

With each overlap  $(T, S, \lambda)$  we associate the intersection  $T \cap (S - \lambda)$ . Recall that by the invariance of the broken line under  $\mathbf{E}_1$  we have the selfsimilarity equation

$$\beta \pi_e([\mathbf{y}, j]_g) = \bigcup_{[\mathbf{x}, i] \in \mathbf{E}_1[\mathbf{y}, j]} \pi_e([\mathbf{x}, i]_g).$$

Applying this equation to the intersection  $T \cap (S - \lambda)$  yields the decomposition

$$\beta(T \cap (S - \lambda)) = \bigcup_{T' \subset \beta T, \, S' \subset \beta S, T' \in \mathcal{E}_u, \, S' \in \mathcal{E}_u} (T' \cap (S' - \beta \lambda)).$$
(5.39)

The sets that occur on the right-hand side of (5.39) are tiles of the synchronised tiling associated with the pair  $(\mathcal{E}_u, (\mathcal{E}_u - \beta \lambda) \cap \mathbb{R}_+)$ .

The following graph will allow us to formulate a new notion of coincidence that will give rise to a further tiling criterion.

**Definition 5.7.4 (Graph of overlaps)** The graph of overlaps, denoted by  $\mathcal{G}_{\mathcal{O}}$ , is a directed graph whose set of vertices is the set  $[\mathcal{O}_u]$  of overlap classes. There is an edge from  $[T, S, \lambda]$  to  $[T', S', \lambda']$  if  $T' \cap (S' - \lambda')$  is nonempty and appears in the self-similar decomposition (5.39) of  $\beta(T \cap (S - \lambda))$ .

Note that Lemma 5.7.2 implies that the graph of overlaps  $\mathcal{G}_{\mathcal{O}}$  is finite and does not depend on the choice of u (see Exercise 5.5).

The graph of overlaps is very close to the configuration graph (and thus to the two-piece ancestor graph). The main difference is that the configuration graph and the two-piece ancestor graph are defined in terms of tips or basic strands in the two-piece seed patch  $W_{\sigma}$ , while the construction of the graph of overlaps does not involve  $W_{\sigma}$ . Indeed, as seen in the proof of Lemma 5.7.2, the graph of overlaps selects pairs of tips and pairs of basic strands according to the set  $\Xi(\mathcal{E}_u)$ . This leads to a finite number of pairs of tips or of pairs of basic strands and no reduction to the two-piece seed patch is needed any more.

We introduce a new notion of coincidence which is defined in terms of the graph of overlaps.

**Definition 5.7.5 (Strong overlap coincidence condition)** The unit Pisot irreducible substitution  $\sigma$  satisfies the *strong overlap coincidence condition* if each vertex in the graph of overlaps  $\mathcal{G}_{\mathcal{O}}$  admits a path leading to an overlap coincidence.

We will prove in Section 5.7.2 that this condition together with the combinatorial strong coincidence condition is equivalent to the tiling property. However, checking the strong overlap coincidence condition is hard since *a priori* it requires to identify *all* non-empty overlaps provided by the expanding tiling, *i.e.*, to consider all parameters  $\lambda \in \Xi(\mathcal{E}_u)$ . Thus the first problem is to determine  $\Xi(\mathcal{E}_u)$ . However, fortunately we will see in Section 5.7.2 that we do not need to work with the whole set  $\Xi(\mathcal{E}_u)$  in order to check the strong overlap coincidence condition. Indeed, we can restrict ourselves to an arbitrary *single* element  $\lambda \in \Xi(\mathcal{E}_u)$ . In particular, we will have to consider only synchronisations for the family  $\{\mathcal{E}_u, \mathcal{E}_u - \lambda, \mathcal{E}_u - \beta\lambda, \ldots, \mathcal{E}_u - \beta^m\lambda, \ldots\}$ .

# **Definition 5.7.6 (Graph of overlaps of** $\lambda$ ) Let $\lambda \in \Xi(\mathcal{E}_u)$ . The graph

of overlaps of  $\lambda$ , denoted by  $\mathcal{G}_{\mathcal{O}}(\lambda)$ , is the subgraph of the graph of overlaps whose vertices belong to  $\bigcup_{i>0} [\mathcal{O}_u(\beta^i \lambda)]$ .

As for each  $\lambda \in \Xi(\mathcal{E}_u)$  the graph  $\mathcal{G}_{\mathcal{O}}(\lambda)$  is a subgraph of the finite graph of overlaps  $\mathcal{G}_{\mathcal{O}}$ , it is itself a finite graph. Moreover, as a finite graph has only finitely many pairwise non-isomorphic subgraphs there are only finitely many different graphs in the class  $\{\mathcal{G}_{\mathcal{O}}(\lambda) \mid \lambda \in \Xi(\mathcal{E}_u)\}$ .

To  $\mathcal{G}_{\mathcal{O}}(\lambda)$  we relate the following coincidence condition which will turn out to be equivalent to the strong overlap coincidence condition (see Theorem 5.7.13).

**Definition 5.7.7 (Weak overlap coincidence)** The unit Pisot irreducible substitution  $\sigma$  satisfies the *weak overlap coincidence condition* if there exists  $\lambda \in \Xi(\mathcal{E}_u)$  with  $\lambda \neq 0$  such that each vertex in its associated graph of overlaps  $\mathcal{G}_{\mathcal{O}}(\lambda)$  admits a path to an overlap coincidence.

The advantage of this condition is that it can be checked effectively in an easy way.

The following lemma contains a first result on the relation between the strong and the weak overlap coincidence condition.

**Lemma 5.7.8** The strong overlap coincidence condition is true if, and only if, the weak overlap coincidence condition is true for each  $\lambda \in \Xi(\mathcal{E}_u)$ .

Proof Assume that the strong coincidence condition is true and choose  $\lambda \in \Xi(\mathcal{E}_u)$  arbitrary. Let  $[T, S, \lambda]$  be a vertex of  $\mathcal{G}_{\mathcal{O}}(\lambda)$ . As  $\mathcal{G}_{\mathcal{O}}(\lambda)$  is a subgraph of  $\mathcal{G}_{\mathcal{O}}$ , there is a path in  $\mathcal{G}_{\mathcal{O}}$  from  $[T, S, \lambda]$  to a coincidence. However, if a vertex of  $\mathcal{G}_{\mathcal{O}}$  is contained in  $\mathcal{G}_{\mathcal{O}}(\lambda)$ , then all its successors in  $\mathcal{G}_{\mathcal{O}}$  are contained in  $\mathcal{G}_{\mathcal{O}}(\lambda)$  in view of (5.39). Thus the path from  $[T, S, \lambda]$  to a coincidence which is contained in  $\mathcal{G}_{\mathcal{O}}$  by assumption is also contained in  $\mathcal{G}_{\mathcal{O}}(\lambda)$ .

To prove the converse assume that the weak overlap coincidence is true for each  $\lambda \in \Xi(\mathcal{E}_u)$ . Choose a vertex of  $\mathcal{G}_{\mathcal{O}}$ . This vertex is of the form  $[T, S, \gamma]$  for some  $\gamma \in \Xi(\mathcal{E}_u)$ . Thus, in  $\mathcal{G}_{\mathcal{O}}(\gamma)$  there is a path from  $[T, S, \gamma]$ to a coincidence. Since  $\mathcal{G}_{\mathcal{O}}(\gamma)$  is a subgraph of  $\mathcal{G}_{\mathcal{O}}$ , this path also exists in  $\mathcal{G}_{\mathcal{O}}$ .

#### 5.7.2 Tiling conditions related to overlap graphs

In this section we have two main aims. First we want to prove that the combinatorial strong coincidence condition together with the strong overlap coincidence condition is equivalent to the tiling property. As mentioned

above, the strong overlap coincidence condition is hard to check. Thus, in a second step, we show that strong and weak overlap coincidence are equivalent. Summing up we will arrive at a tiling criterion in terms of combinatorial strong coincidence and weak overlap coincidence.

In all what follows we enumerate the tiles of  $\mathcal{E}_u$  starting from the tile next to the origin by  $T_0, T_1, \ldots$  One has  $T_r = \pi_e [\mathbf{P}(u_0 \cdots u_{r-1}), u_r]_g$  for all  $r \in \mathbb{N}$ . Moreover, for each  $\lambda \in \Xi(\mathcal{E}_u)$ , we consider the union of tiles  $T \in \mathcal{E}_u$ such that  $T + \lambda$  also occurs in  $\mathcal{E}_u$ , *i.e.*,

$$\operatorname{Occ}(\lambda) := \bigcup_{\{T \in \mathcal{E}_u \mid T + \lambda \in \mathcal{E}_u\}} T.$$

We start with the following lemma (cf. (Solomyak 1997, Proposition 6.7) and (Lee, Moody, and Solomyak 2003, Lemma A.8)), which translates the weak overlap coincidence condition in combinatorial terms (see also the related result (Queffélec 1987, Lemma VI.27)).

**Lemma 5.7.9** Let  $\lambda \in \Xi(\mathcal{E}_u)$ . The graph of overlaps of  $\lambda$  satisfies the weak overlap coincidence condition if, and only if, there exists some constant  $c \in (0, 1)$  such that

$$\operatorname{Card}\{T_r \in \mathcal{E}_u \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\} \ll N c^m$$
(5.40)

holds for all  $m \in \mathbb{N}$  when N is large enough in terms of m (the implied constant does not depend on N and m).

*Proof* We first prove that there is a constant  $\tilde{c} \in (0, 1)$  such that

$$\limsup_{N \to \infty} \frac{\mu_1(\{T_r \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\})}{\mu_1(\{T_r \mid r \le N\})} \ll \tilde{c}^m$$
(5.41)

if, and only if, there exists some constant  $c \in (0, 1)$  such that (5.40) holds for all  $m \in \mathbb{N}$  when N is large enough in terms of m.

Assume that (5.40) holds. Let  $L_{\min}$  and  $L_{\max}$  be the length of the shortest and longest tile in  $\mathcal{E}_u$ , respectively. One has for N large enough

$$\mu_1(\{T_r \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\}) \le L_{\max} \operatorname{Card}\{T_r \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\}$$
$$\ll c^m N \le c^m \frac{1}{L_{\min}} \mu_1(\{T_r \mid r \le N\})$$
$$\ll c^m \mu_1(\{T_r \mid r \le N\}).$$

Hence,

$$\limsup_{N \to \infty} \frac{\mu_1(\{T_r \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\})}{\mu_1(\{T_r \mid r \le N\})} \ll c^m$$

and (5.41) is true for  $\tilde{c} = c$ .

Conversely, assume that (5.41) holds for  $\tilde{c} \in (0,1)$ . Let  $c \in (0,1)$  with  $\tilde{c} < c$ . For N large enough in terms of m, one has

$$\mu_1(\{T_r \mid r \le N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u\}) \ll c^m \mu_1(\{T_r \mid r \le N\}).$$

Hence,

$$\begin{aligned} \operatorname{Card}(\{T_r \mid r \leq N, \, T_r + \beta^m \lambda \notin \mathcal{E}_u\}) \\ &\leq \frac{1}{L_{\min}} \mu_1(\{T_r \mid r \leq N, \, T_r + \beta^m \lambda \notin \mathcal{E}_u\}) \\ &\ll c^m \mu_1(\{T_r \mid r \leq N\}) \leq c^m N L_{\max} \ll c^m N, \end{aligned}$$

which ends the proof of the claimed equivalence.

We now prove (5.41). We first assume that the graph of overlaps  $\mathcal{G}_{\mathcal{O}}(\lambda)$ of  $\lambda$  satisfies the weak overlap coincidence condition. By the finiteness of  $\mathcal{G}_{\mathcal{O}}(\lambda)$  there exists a positive integer  $\ell$  such that each vertex admits a path of length bounded by  $\ell$  to an overlap coincidence.

Using (5.39), one checks that  $\mathcal{G}_{\mathcal{O}}(\beta^m\lambda)$  is a subgraph of  $\mathcal{G}_{\mathcal{O}}(\lambda)$  whose paths to overlap coincidences do not get longer as the ones in  $\mathcal{G}_{\mathcal{O}}(\lambda)$ . In particular,  $\beta^{\ell}(T \cap (S - \beta^m\lambda))$  contains an overlap coincidence for each overlap  $(T, S, \beta^m\lambda)$ . One has  $\mu_1(\beta^{\ell}(T \cap (S - \beta^m\lambda))) \leq \beta^{\ell}L_{\max}$ . Moreover, each overlap  $(T, S, \beta^m\lambda)$  produces tiles which belong to  $\operatorname{Occ}(\beta^{m+\ell}\lambda)$ , and whose union has length bounded from below by  $L_{\min}$ . Let  $b := \left(1 - \frac{L_{\min}}{L_{\max}\beta^{\ell}}\right)$ . Since the definition of Occ immediately implies  $\beta \operatorname{Occ}(\beta^m\lambda) \subset \operatorname{Occ}(\beta^{m+1}\lambda)$ , and, hence,  $\beta^{\ell}\operatorname{Occ}(\beta^m\lambda) \subset \operatorname{Occ}(\beta^{m+\ell}\lambda)$ , we deduce that

$$\limsup_{N \to \infty} \frac{\mu_1 \{ T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^{m+\ell} \lambda \notin \mathcal{E}_u \}}{\mu_1 \{ T_r \in \mathcal{E}_u \mid r \leq N \}}$$
$$\leq b \limsup_{N \to \infty} \frac{\mu_1 \{ T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^m \lambda \notin \mathcal{E}_u \}}{\mu_1 \{ T_r \in \mathcal{E}_u \mid r \leq N \}}$$

Since b < 1 does not depend on m and N, by writing  $m = k\ell + s$  with  $0 \le s < \ell$  we easily derive (5.41) by iteration.

To prove the converse assume that (5.41) holds and that  $\mathcal{G}_{\mathcal{O}}(\lambda)$  does not satisfy the weak overlap coincidence condition. Then there exists an overlap class  $[T, S, \beta^m \gamma]$  such that for every  $\ell, m > 0$ 

$$\beta^{\ell}(T \cap (S - \beta^m \lambda)) \subset \mathbb{R}_+ \setminus \operatorname{Occ}(\beta^m \lambda).$$

By the repetitivity of  $\mathcal{E}_u$  this yields

$$\limsup_{N \to \infty} \frac{\mu_1\{T_r \in \mathcal{E}_u \mid r \le N, \ T_r + \beta^{m+\ell} \lambda \notin \mathcal{E}_u\}}{\mu_1\{T_r \in \mathcal{E}_u \mid r \le N\}} \gg 1,$$

uniformly in m, which contradicts (5.41).

We now establish the relation between the tiling property and the strong overlap coincidence condition. The first part of the following lemma is proved in (Lee 2007) in the general context of substitution Delone sets and Meyer sets. Indeed, it can be derived from the *algebraic coincidence condition* introduced in (Lee 2007). For more details, see the notes at the end of the present chapter.

**Lemma 5.7.10** Let  $\sigma$  be a unit Pisot irreducible substitution. If  $\sigma$  satisfies the strong overlap coincidence condition, then the following assertions are true.

- There exists  $m \in \mathbb{N}$  such that  $\beta^m(\Xi(\mathcal{E}_u) \Xi(\mathcal{E}_u)) \cap \mathbb{R}_+ \subset \Xi(\mathcal{E}_u)$ .
- There exists  $m \in \mathbb{N}$  such that  $\beta^m(\Xi(\mathcal{E}_u) + \Xi(\mathcal{E}_u)) \subset \Xi(\mathcal{E}_u)$ .

Proof Let  $\lambda_1, \lambda_2 \in \Xi(\mathcal{E}_u)$  with  $\lambda_2 - \lambda_1 > 0$  be given. Then Lemma 5.7.9 implies that there exists  $m \in \mathbb{N}$  such that for all N large enough and for i = 1, 2, one has

$$\operatorname{Card}\{r \le N \mid T_r + \beta^m \lambda_i \notin \mathcal{E}_u\} < \frac{N}{3}.$$
(5.42)

This implies that

$$\operatorname{Card}\{r \le N \mid T_r + \beta^m \lambda_1 \notin \mathcal{E}_u \text{ or } T_r + \beta^m \lambda_2 \notin \mathcal{E}_u\} < \frac{2N}{3}$$
(5.43)

which means that there is some  $r \in \mathbb{N}$  such that  $T_r + \lambda_1 \beta^m \in \mathcal{E}_u$  and  $T_r + \lambda_2 \beta^m \in \mathcal{E}_u$ . Thus,  $\beta^m (\lambda_2 - \lambda_1) \in \Xi(\mathcal{E}_u)$  which proves the first assertion.

Recall that  $L_{\min}$  denotes the length of the shortest tile in  $\mathcal{E}_u$ . If N is large in terms of m, one has moreover

$$\operatorname{Card}\{\beta^m \lambda_1 / L_{\min} \le r \le N \mid T_r - \beta^m \lambda_1 \notin \mathcal{E}_u\} < \frac{N}{3}$$

which implies

 $\operatorname{Card}\{\beta^m \lambda_1 / L_{\min} \le r \le N \mid T_r - \beta^m \lambda_1 \notin \mathcal{E}_u \text{ or } T_r + \beta^m \lambda_2 \notin \mathcal{E}_u\} < \frac{2N}{3}.$ 

Thus, if N is large in terms of m, there exists  $r \leq N$  such that  $T_r - \beta^m \lambda_1 \in \mathcal{E}_u$  and  $T_r + \lambda_2 \beta^m \in \mathcal{E}_u$ . Thus,  $\beta^m (\lambda_2 + \lambda_1) \in \Xi(\mathcal{E}_u)$  which proves the second assertion.

In order to get a relation between the strong overlap coincidence condition and the tiling condition, we need to ensure that all lengths of tiles in  $\mathcal{E}_u$  multiplied by some power of  $\beta$  belong to the translation set  $\Xi(\mathcal{E}_u)$ . This is realised by assuming the combinatorial strong coincidence condition introduced in Definition 5.2.9.

**Lemma 5.7.11** Let  $\sigma$  be a unit Pisot irreducible substitution that satisfies the combinatorial strong coincidence condition. Then there exists m such that  $\beta^m \pi_e \circ \mathbf{P}(i) \in \Xi(\mathcal{E}_u)$  for every letter  $i \in A$ .

Proof Let  $i \in A$ . Let  $j \in A$  be such that ij is a factor of the fixed point u. By the combinatorial strong coincidence condition, there exist m > 0, a letter k and four words p, q, r, s such that  $\sigma^m(i) = pkq$  and  $\sigma^m(j) = rks$ , with  $\mathbf{P}(p) = \mathbf{P}(r)$ . Since  $\sigma^m(ij)$  is a factor of u, we have that  $\pi_e \circ \mathbf{P}(kqr) \in \Xi(\mathcal{E}_u)$ . The relation  $\mathbf{P}(p) = \mathbf{P}(r)$  yields  $\pi_e \circ \mathbf{P}(kqr) =$  $\pi_e \circ \mathbf{P}(pkq) = \pi_e \circ \mathbf{P}(\sigma^m(i)) = \beta^m \pi_e \circ \mathbf{P}(i) \in \Xi(\mathcal{E}_u)$ .

We can now derive the following relation between the strong overlap coincidence condition and the tiling condition.

**Theorem 5.7.12** Let  $\sigma$  be a unit Pisot irreducible substitution. Then  $\sigma$  satisfies the tiling property if, and only if,  $\sigma$  satisfies both the strong overlap coincidence condition and the combinatorial strong coincidence condition.

**Proof** We first assume that  $\sigma$  satisfies the tiling property. By Theorem 5.6.8,  $\sigma$  satisfies the super coincidence condition and, in particular, the combinatorial strong coincidence condition. According to Remark 5.7.3, each overlap class  $[T, S, \lambda]$  is associated with a pair of basic formal strands  $([\mathbf{0}, i], [\mathbf{z}, j])$  having the same height. Since the super coincidence condition holds, there exists m such that  $\mathbf{E}_1^m[\mathbf{z}, i]$  and  $\mathbf{E}_1^m[0, j]$  contain a common basic formal strand. In other words, one gets an overlap coincidence between  $\beta^m T$  and  $\beta^m (S - \lambda)$ . Therefore the strong overlap coincidence is satisfied.

The converse is slightly more difficult to establish. This is mostly due to the specific positions of the vectors  $\mathbf{z}$  associated with overlap classes as noticed in Remark 5.7.3. We assume that  $\sigma$  satisfies both the strong overlap coincidence condition and the combinatorial strong coincidence condition. Let  $[\mathbf{x}, i]$  and  $[\mathbf{y}, j]$  be a pair of basic formal strands with the same height. Let  $T := \pi_e([\mathbf{x}, i]_g)$  and  $S := \pi_e([\mathbf{y}, j]_g)$ . There exist  $\mathbf{z}_1, \mathbf{z}_2 \in \pi_e(\mathbb{Z}^n)$ such that  $T \in \mathcal{E}_u - \pi_e(\mathbf{z}_1)$  and  $S \in \mathcal{E}_u - \pi_e(\mathbf{z}_2)$ . Assume w.l.o.g. that  $\pi_e(\mathbf{z}_1) < \pi_e(\mathbf{z}_2)$ . Now set  $T' := T + \pi_e(\mathbf{z}_1)$  and  $S' := S + \pi_e(\mathbf{z}_1)$ . Then  $T' \in \mathcal{E}_u$  and  $S' \in \mathcal{E}_u - \lambda$  with  $\lambda := \pi_e(\mathbf{z}_2 - \mathbf{z}_1)$ .

By Lemma 5.7.11, the combinatorial strong coincidence condition implies that there exists  $m_1 \in \mathbb{N}$  such that  $\beta^{m_1} \pi_e \circ \mathbf{P}(k) \in \Xi(\mathcal{E}_u)$  for all  $k \in A$ . As  $\{\mathbf{P}(1), \ldots, \mathbf{P}(n)\} = \{\mathbf{e}(1), \ldots, \mathbf{e}(n)\}$  forms a basis of the lattice  $\mathbb{Z}^n$ ,

according to Lemma 5.7.10 for each  $\mathbf{z} \in \mathbb{Z}^n$  there exists  $m_2 \in \mathbb{N}$  such that  $\beta^{m_2}\pi_e(\mathbf{z}) \in \Xi(\mathcal{E}_u)$ . Thus there exists  $m \in \mathbb{N}$  such that  $\lambda \in \Xi(\mathcal{E}_u)$ . We thus deduce from the strong overlap coincidence condition that  $[\beta^m T, \beta^m S, \beta^m \lambda]$  leads to a coincidence overlap. This implies that the super coincidence condition holds.

We now turn to the second main result of the present section, the equivalence between the strong and the weak overlap coincidence condition. This is proved in a slightly different context in (Solomyak 1997, Section 6) where it is shown that both conditions are equivalent to the fact that the dynamical system associated with the tiling has pure discrete spectrum (see also (Lee 2007, Section 3) where overlap coincidence is related to pure discrete spectrum). We give a new and direct proof of this result here.

**Theorem 5.7.13** Let  $\sigma$  be a unit Pisot irreducible substitution. Then the following assertions are equivalent.

- (i) The substitution  $\sigma$  satisfies the weak overlap coincidence condition.
- (ii) The substitution  $\sigma$  satisfies the strong overlap coincidence condition.

In view of Lemmas 5.7.8 and 5.7.9 it is clear that Theorem 5.7.13 is a direct consequence of the following lemma. In the proof of this lemma we exploit the fact that weak overlap coincidence implies that the fixed point u of  $\sigma$  is mean-almost periodic in the sense of (Queffélec 1987, Definition VI.4).

**Lemma 5.7.14** Let  $\lambda_1, \lambda_2 \in \Xi(\mathcal{E}_u)$  with  $\lambda_1 \neq 0$ . If there exists  $c \in (0, 1)$  such that

$$\operatorname{Card}\{T_r \in \mathcal{E}_u \mid r \le N, \ T_r + \beta^m \lambda_1 \notin \mathcal{E}_u\} \ll N \, c^m \tag{5.44}$$

for all m when N is large enough in terms of m then

Card{ $T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u$ }  $\ll Nc^m$ 

for all m when N is large enough in terms of m. The implied constants do not depend on m and N.

*Proof* Let  $\lambda_1$  be a non-zero element of  $\Xi(\mathcal{E}_u)$  such that (5.44) holds and let  $\lambda_2 \in \Xi(\mathcal{E}_u)$ .

We consider a tile T that occurs in  $Occ(\lambda_2)$ . One has  $T + \lambda_2 \in \mathcal{E}_u$ . By the repetitivity of  $\mathcal{E}_u$ , there is a set of tiles  $W^{(0)} \subset \mathcal{E}_u$  such that the union of tiles of  $W^{(0)}$  is a relatively dense set in  $\mathbb{R}_+$  (*i.e.*, there exists L > 0 such that every interval of length L in  $\mathbb{R}_+$  contains at least one point belonging to one tile of  $W^{(0)}$ ) and  $T_r + \beta \lambda_2 \in \mathcal{E}_u$  whenever  $T_r \in W^{(0)}$ .

We now use the self-similarity of  $\mathcal{E}_u$  (or equivalently, the fact that  $\sigma(u) =$ u). We multiply all tiles of  $W^{(0)}$  by  $\beta^m$  and subdivide accordingly to arrive again at  $\mathcal{E}_u$ . Thus, there exist

$$u_1^{(m)} < v_1^{(m)} < u_2^{(m)} < v_2^{(m)} < u_3^{(m)} < v_3^{(m)} < \cdots$$

with

$$\min_{i}(v_{i}^{(m)} - u_{i}^{(m)}) \gg \beta^{m}, \ \max_{i}(u_{i+1}^{(m)} - v_{i}^{(m)}) \ll \beta^{m}, \ u_{1}^{(m)} \ll \beta^{m}$$
(5.45)

such that  $T_r + \beta^m \lambda_2 \in \mathcal{E}_u$  whenever  $T_r \in W^{(m)}$ , where  $W^{(m)} := \{T_s \mid s \in \mathcal{E}_v \mid s \in \mathcal{E}_v \mid s \in \mathcal{E}_v \}$  $\bigcup_{i\geq 1} \llbracket u_i^{(m)}, v_i^{(m)} \rrbracket$ }. All bounds in (5.45) follow from the fact that

$$\forall a \in A, \ \beta^m \ll |\sigma^m(a)| \ll \beta^m \tag{5.46}$$

(for more details on (5.46) see Section 4.7.3). To get the upper bound for  $\max_i (u_{i+1}^{(m)} - v_i^{(m)})$  also the relative denseness of the union of tiles of  $W^{(0)}$ has to be used.

Let  $m_0 \in \mathbb{N}$  be fixed in a way that  $\beta^{m-m_0} \lambda_1 < \min_i (v_i^{(m)} - u_i^{(m)})$  (such a constant exists in view of (5.45)).

For each  $K \in \mathbb{N}$  define the 'exceptional set'

$$S_K := \{ T_r \in \mathcal{E}_u \mid T_r - k\beta^{m-m_0}\lambda_1 \notin \mathcal{E}_u \text{ for some } 0 \le k \le K \}.$$

Equation (5.44) implies that

$$\operatorname{Card}(S_K \cap \{T_r \mid r \le N\}) \ll Nc^m \tag{5.47}$$

holds for all m if N large enough (note that the implied constant may depend on K but this is not relevant for us as K will be fixed in a moment). By (5.45) and since  $\lambda_1 \neq 0$  we may fix  $K \in \mathbb{N}$  in a way that for all  $T_r \in \mathcal{E}_u \setminus$  $S_K$  and all  $m \in \mathbb{N}$ , there exists  $k \in [0, K]$  such that  $T_r - k\beta^{m-m_0}\lambda_1 \in W^{(m)}$ .

We set for  $k \in [\![0, K]\!]$ 

$$\mathcal{E}_u^{(k,m)} := \{ T_r \in \mathcal{E}_u \mid T_r - k\beta^{m-m_0} \lambda_1 \in W^{(m)} \}.$$

Then

$$\{T_r \mid r \in \mathbb{N}\} = \bigcup_{k=0}^{K} \mathcal{E}_u^{(k,m)} \cup S_K.$$
(5.48)

Note that

$$T_r \in \mathcal{E}_u^{(0,m)} \implies T_r + \beta^m \lambda_2 \in \mathcal{E}_u.$$

We now take  $k \neq 0$ . Let  $T_r \in \mathcal{E}_u^{(k,m)}$  such that  $T_r + \beta^m \lambda_2 \notin \mathcal{E}_u$ . One has  $T_r - k\beta^{m-m_0}\lambda_1 \in W^{(m)}$ , hence,

$$T_r - k\beta^{m-m_0}\lambda_1 + \beta^m\lambda_2 \in \mathcal{E}_u.$$

We thus have

$$T_r - k\beta^{m-m_0}\lambda_1 + \beta^m\lambda_2 \in \mathcal{E}_u, \ (T_r - k\beta^{m-m_0}\lambda_1 + \beta^m\lambda_2) + k\beta^{m-m_0}\lambda_1 \notin \mathcal{E}_u.$$

Hence, by recalling that  $L_{\min}$  is the length of the smallest tile in  $\mathcal{E}_u$ , we get for  $N \in \mathbb{N}$ 

$$\operatorname{Card}\left\{T_r \in \mathcal{E}_u^{(k,m)} \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u\right\} \leq \\\operatorname{Card}\left\{T_r \mid r \leq N + \frac{\beta^m \lambda_2}{L_{\min}}, \ T_r + k\beta^{m-m_0} \lambda_1 \notin \mathcal{E}_u\right\}.$$

Putting this together with (5.48), we obtain

$$\operatorname{Card} \left\{ T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u \right\}$$
$$\leq \sum_{k=0}^{K} \operatorname{Card} \left\{ T_r \in \mathcal{E}_u^{(k,m)} \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u \right\}$$
$$+ \operatorname{Card}(S_K \cap \{T_r \mid r \leq N\})$$
$$\leq \sum_{k=0}^{K} \operatorname{Card}\{T_r \mid r \leq N + \frac{\beta^m \lambda_2}{L_{\min}}, \ T_r + k\beta^{m-m_0} \lambda_1 \notin \mathcal{E}_u \}$$
$$+ \operatorname{Card}(S_K \cap \{T_r \mid r \leq N\}).$$

We deduce from (5.44), (5.45) and (5.47) that

$$\operatorname{Card}\{T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u\} \ll \left(N + \frac{K\beta^m \lambda_2}{L_{\min}}\right) c^m + N c^m.$$

This implies that

$$\operatorname{Card}\{T_r \in \mathcal{E}_u \mid r \leq N, \ T_r + \beta^m \lambda_2 \notin \mathcal{E}_u\} \ll Nc^m$$

holds for all m when N is large in terms of m.

As mentioned above, Theorem 5.7.13 is an immediate consequence of the previous lemma. According to this theorem it is not necessary to build all of the configuration graph (or of the two-piece ancestor graph) in order to check the tiling property. Indeed, it is enough to build the graph from all pairs of tiles in the broken line that are separated by a fixed vector. We sum this up in the following corollary.

**Corollary 5.7.15** Let  $\sigma$  be a unit Pisot irreducible substitution. Then  $\sigma$  satisfies the tiling property if, and only if,  $\sigma$  satisfies both the weak overlap coincidence condition and the combinatorial strong coincidence condition.

The computation of the pairs of such tiles still requires the knowledge of the language of the fixed point. In Section 5.8, we will take advantage of the weak overlap property in order to obtain a purely combinatorial characterisation of the tiling property. To this end we will need the following corollary which focuses on gaps between overlaps instead of focusing on density of overlaps. Its proof is an immediate consequence of the proof of Lemma 5.7.9.

**Corollary 5.7.16** Let  $\lambda \in \Xi(\mathcal{E}_u)$ . The weak overlap coincidence condition is satisfied for a  $\lambda \in \Xi(\mathcal{E}_u)$  if, and only if, the distance between two successive coincidence overlaps in  $(\mathcal{E}_u, (\mathcal{E}_u - \beta^k \lambda) \cap \mathbb{R}_+)$  is bounded uniformly in k.

### 5.8 Balanced pair algorithm

This section is devoted to a further effective condition for the tiling property based on the notion of *balanced pairs* introduced in (Michel 1978) and later used *e.g.* by (Livshits 1987), (Queffélec 1987, Chapter VI), (Sirvent and Solomyak 2002) and (Martensen 2004). The starting point for the balanced pair algorithm is Corollary 5.7.16. It states that the tiling property is strongly related to the uniform boundedness (in k) of the length of gaps between successive coincidence overlaps in pairs of tilings of the form  $(\mathcal{E}_u, (\mathcal{E}_u - \beta^k \lambda) \cap \mathbb{R}_+))$ . According to Theorem 5.7.13, this property has to be checked only for one suitable  $\lambda \in \Xi(\mathcal{E}_u)$ . We will choose  $\lambda$  of the form  $\lambda = \langle \mathbf{P}(w), \mathbf{v}_\beta \rangle$  for some prefix w of u with  $u_0 = u_{|w|}$ . For this choice we get that the first tile in  $\mathcal{E}_u$  coincides with a tile of  $(\mathcal{E}_u - \beta^k \lambda) \cap \mathbb{R}_+$  for each  $k \in \mathbb{N}$ . Indeed, for each k one has  $\beta^k \lambda = \langle \mathbf{M}_{\sigma}^k \mathbf{P}(w), \mathbf{v}_{\beta} \rangle = \langle \mathbf{P}(\sigma^k(w)), \mathbf{v}_{\beta} \rangle$ where  $\sigma^k(w)$  is a prefix of u. Note that the first overlap and the last overlaps of the part of the synchronised tiling  $\mathcal{E}_u \cap (\mathcal{E}_u - \beta^k \lambda)$  that is located between 0 and  $\beta^k \lambda$  are half-coincidences.

In applying the balanced pair algorithm, we will start with gaps between half-coincidence overlaps in order to get the uniform boundedness of gaps between successive coincidence overlaps. A gap between two halfcoincidence overlaps can be described as an interval that can be decomposed in two ways: firstly, as a union of consecutive tiles of  $\mathcal{E}_u$  and secondly, as a union of consecutive tiles of  $(\mathcal{E}_u - \lambda \beta^k) \cap \mathbb{R}_+$ . The types of these consecutive tiles correspond to two finite subwords  $v_1$  and  $v_2$  of u, respectively. Since the coordinates of  $\mathbf{v}_\beta$  are rationally independent, we have  $\mathbf{P}(v_1) = \mathbf{P}(v_2)$ . This leads us to introduce the following combinatorial definition.

**Definition 5.8.1 (Balanced pair)** A pair  $(v_1, v_2) \in A^* \times A^*$  is said to

be a combinatorial balanced pair, or for short, a balanced pair, if  $\mathbf{P}(v_1) = \mathbf{P}(v_2)$ . A one-letter balanced pair (also called *coincidence balanced pair*) is a balanced pair of the form (a, a), with  $a \in A$ .

An *irreducible balanced pair* is a pair  $(v_1, v_2)$  with the property that no pair  $(v'_1, v'_2)$ , where  $v'_i$  is a proper prefix  $v_i$ , i = 1, 2, is balanced.

Note that when  $(v_1, v_2)$  is a balanced pair,  $(\sigma(v_1), \sigma(v_2))$  is balanced as well. Obviously, each balanced pair can be split up uniquely into irreducible balanced pairs. This process is called *reduction*.

The set of irreducible balanced pairs obtained after the reduction of a balanced pair of finite words  $(v_1, v_2)$  is denoted by  $\mathcal{B}(v_1, v_2)$ . We say that a balanced pair  $(v_1, v_2)$  leads to a coincidence if there exists  $N \in \mathbb{N}$  such that  $\mathcal{B}(\sigma^N(v_1), \sigma^N(v_2))$  contains a one-letter balanced pair.

The following algorithm will lead to a powerful criterion for the tiling property.

**Definition 5.8.2 (Balanced pair algorithm)** Let  $I_0$  be a non-empty and finite set of balanced pairs. The *balanced pair algorithm* applied to  $I_0$  successively computes the sets

$$I_k := \bigcup_{(v_1, v_2) \in I_{k-1}} \mathcal{B}(\sigma(v_1), \sigma(v_2)).$$

The algorithm is said to terminate with rank  $k, k \ge 1$ , if  $I_{k+1} = I_k$  and each balanced pair  $(v_1, v_2) \in I_k$  leads to a coincidence. The algorithm is said to terminate if it terminates for some rank  $k \ge 1$ .

Let  $I_0 = \{(v_1, v_2)\}$ . If the balanced pair algorithm terminates there exist only finitely many different words w which occur between two consecutive one-letter balanced pairs in the pairs  $(\sigma^k(v_1), \sigma^k(v_2))$ . Thus the length of such gaps is uniformly bounded in k. We formulate this in a more exact way in the following proposition (see also (Sirvent and Solomyak 2002, Theorem 5.6)).

**Proposition 5.8.3** The balanced pair algorithm applied to a non-empty finite set  $I_0$  of balanced pairs terminates if, and only if, the number of letters between two successive one-letter balanced pairs in  $(\sigma^k(v_1), \sigma^k(v_2))$ for any  $(v_1, v_2) \in \bigcup_k I_k$ , is bounded uniformly in k.

**Example 5.8.4** In this example we want to consider the substitution  $\sigma(1) = 112$ ,  $\sigma(2) = 13$  and  $\sigma(3) = 1$ . It is easy to check that this is a unit Pisot irreducible substitution. We want to perform the balanced pair

algorithm for this example, starting from  $I_0 = \{(12, 21), (13, 31), (23, 32)\}$ . Applying  $\sigma$  and reducing yields

$$\begin{array}{l} (12,21) \rightarrow (11213,13112) \rightarrow (1,1)(1213,3112), \\ (13,31) \rightarrow (1121,1112) \rightarrow (1,1)(1,1)(21,12), \\ (23,32) \rightarrow (131,113) \rightarrow (1,1)(31,13). \end{array}$$

Thus in  $I_1$  we have the one-letter pairs (2, 2), (1, 1) and the new pair (1213, 3112). Repeating the procedure for  $I_1$  yields the new reduction

 $(1213, 3112) \rightarrow (112131121, 111211213) \rightarrow (1, 1)(1, 1)(21, 12)(31121, 11213).$ 

Now  $I_2$  contains the new pair (31121, 11213). It is treated as follows:

$$\begin{array}{l} (31121,11213) \rightarrow (111211213112,112112131121) \\ \rightarrow (1,1)(1,1)(12,21)(1,1)(12,21)(13,31)(1,1)(12,21). \end{array}$$

This implies that  $I_3$  contains no new pair. Moreover, each of the occurring pairs leads to a one-letter balanced pair. Thus the algorithm terminates.

The notions of balanced pair and reduction also make sense for pairs of infinite words. Let w be a non-empty prefix of the infinite word u. It is not hard to see that the set of irreducible balanced pairs occurring by reducing  $(u, S^{|w|}(u))$  (here S denotes the shift) is a finite set (see for instance (Sirvent and Solomyak 2002, Section 3) and Exercise 5.6). Denote this set by  $I_0(w)$ . We then consider the balanced pairs in  $(u, S^{|\sigma(w)|}(u))$ , which amounts to applying the reduction process to all balanced pairs in  $I_0(w)$ . The balanced pair algorithm starting with the set  $I_0 = I_0(w)$  is called the *balanced pair algorithm associated with* w.

The balanced pair algorithm is stated in the literature in several different forms. Besides taking the set  $I_0(w)$  associated with some prefix w of u as the starting point (Sirvent and Solomyak 2002), the initial set  $I_0$  is defined as  $I_0 = \{(ij, ji) \mid i, j \in A, i \neq j\}$  in (Barge and Kwapisz 2006). As we will see, both starting sets lead to the same behaviour. Sometimes, it proves to be more convenient to start with  $\{(ij, ji) \mid i \neq j\}$  instead of starting with  $I_0(w)$  (even if this latter set can be determined in an effective way).

We now relate the balanced pair algorithm to the overlap coincidence condition.

**Theorem 5.8.5** Let  $\sigma$  be a unit Pisot irreducible substitution. Let w be a prefix of u with  $u_{|w|} = u_0$  and set  $\lambda = \pi_e \circ \mathbf{P}(w) \in \Xi(\mathcal{E}_u)$ . The substitution  $\sigma$  satisfies the weak overlap coincidence condition for  $\lambda$  if, and only if, the balanced pair algorithm associated with w terminates.

Proof Since w is a prefix of u,  $(\mathcal{E}_u - \lambda) \cap \mathbb{R}_+$  is obtained from  $\mathcal{E}_u$  by deleting the |w| first tiles. This implies that coincidence overlaps between tiles in  $(\mathcal{E}_u, (\mathcal{E}_u - \lambda) \cap \mathbb{R}_+)$  are in one-to-one correspondence with one-letter balanced pairs in  $(u, S^{|w|}u)$ . The same correspondence holds between  $(\mathcal{E}_u, (\mathcal{E}_u - \beta^k \lambda) \cap \mathbb{R}_+)$  and  $(u, S^{|\sigma^k(w)|}u)$ . Now compare a step from  $I_k(w)$  to  $I_{k+1}(w)$ in the balanced pair algorithm with the symbolic interpretation of the selfsimilarity equation (5.39). The result now follows from Proposition 5.8.3 and Corollary 5.7.16.

Corollary 5.7.15 and Theorem 5.8.5 immediately imply the following result (see also (Sirvent and Solomyak 2002, Section 5)).

**Corollary 5.8.6** A unit Pisot irreducible substitution  $\sigma$  satisfies the tiling property if, and only if, the combinatorial strong coincidence condition is satisfied and there exists a prefix w of u with  $u_0 = u_{|w|}$  such that the balanced pair algorithm associated with w terminates.

An even easier criterion for the tiling property can be obtained by starting the balanced pair algorithm with  $I_0 = \{(ij, ji) \mid i \neq j\}$  instead of  $I_0(w)$ . As will be proved below (see Theorem 5.8.8), applying the balanced pair algorithm with this choice of  $I_0$  allows both the combinatorial strong coincidence condition and the weak overlap condition to be checked at once. This yields the purely combinatorial algorithm for tiling discussed in (Barge and Kwapisz 2006). Before stating and proving Theorem 5.8.8, we establish the following auxiliary result.

**Lemma 5.8.7** Let  $\sigma$  be a unit Pisot irreducible substitution. If the balanced pair algorithm starting with  $I_0 = \{(ij, ji) \mid i \neq j\}$  terminates, then the balanced pair algorithm starting with any balanced pair  $(w, w') \in A^* \times A^*$  terminates.

Proof We assume that the balanced pair algorithm starting with  $I_0 = \{(ij, ji) \mid i \neq j\}$  terminates. Let  $w, w' \in A^*$  be such that  $\mathbf{P}(w) = \mathbf{P}(w')$ . Let  $\ell$  be equal to the common length |w| = |w'| of w and w'. Since  $\mathbf{P}(w) = \mathbf{P}(w')$ , there exists a permutation  $\rho$  of the set  $[1, \ell]$  such that  $w' = w_{\rho}$ where  $w_{\rho}$  is a shorthand for the word  $w_{\rho(1)} \cdots w_{\rho(\ell)}$ . We recall that a *transposition* is a permutation that exchanges two elements and that lets the other elements invariant, and that the symmetric group, *i.e.*, the group of permutations of  $[1, \ell]$  is generated by the permutations  $\tau_i$ , for  $i \in [1, \ell - 1]$ , where  $\tau_i := (i, i + 1)$  is the transposition of  $[1, \ell]$  that exchanges i and i + 1. We set  $T := \{\tau_i \mid i \in [1, \ell - 1]\}$ . V. Berthé, A. Siegel, J. Thuswaldner

We have to show that the balanced pair algorithm starting with  $(w, w_{\rho})$  terminates for each permutation  $\rho$ . We will prove this by induction.

In order to establish the induction start we have to consider the balanced pair algorithm starting with  $(w, w_{\tau})$  for some  $\tau \in T$ . However, since  $\mathcal{B}(w, w_{\tau}) \subset \{(i, i)\} \cup \{(ij, ji) \mid i \neq j\}$  we only have to consider elements of  $I_0 = \{(ij, ji) \mid i \neq j\}$  as the starting set to assure termination of the algorithm. Thus the balanced pair algorithm starting with  $(w, w_{\tau})$  terminates by the assumptions of the lemma.

The induction step is proved if we establish the following 'transitivity property'. Let  $\tau, \tau' \in T$ . If the balanced pair algorithm starting with  $(w, w_{\tau})$  terminates, and if the balanced pair algorithm starting with  $(w_{\tau}, w_{\tau'\circ\tau})$  terminates, then also the balanced pair algorithm starting with  $(w, w_{\tau'\circ\tau})$  terminates. To show this, according to Proposition 5.8.3, it is sufficient to prove that the occurrences between successive one-letter balanced pairs in  $(\sigma^k(w), \sigma^k(w_{\tau'\circ\tau}))$  are bounded uniformly in k. As the balanced pair algorithm starting with  $(w, w_{\tau})$  as well as with  $(w_{\tau}, w_{\tau\circ\tau'})$  terminates, Proposition 5.8.3 implies that there exists a constant C > 0 such that the occurrences between successive one-letter balanced pairs in  $(\sigma^k(w), \sigma^k(w_{\tau}))$ and in  $(\sigma^k(w_{\tau}), \sigma^k(w_{\tau'\circ\tau}))$  are bounded by C for each k.

Let N be large enough such that  $|\sigma^N(a)| \ge C+1$ , for every letter  $a \in A$ . Fix  $k \in \mathbb{N}$  and let j be the index of a one-letter balanced pair in the pair of words  $(\sigma^k(w), \sigma^k(w_\tau))$ , *i.e.*,

$$\sigma^{k}(w) = pas, \ \sigma^{k}(w_{\tau}) = p'as' \text{ with } \mathbf{P}(p) = \mathbf{P}(p') \text{ and } |p| = |p'| = j-1.$$

Let j' be the index of the first letter of the image  $\sigma^N(a)$  of the *j*th letter in  $\sigma^k(w)$  as well as in  $\sigma^k(w_\tau)$  under  $\sigma^N$ . This implies that each letter with index  $\ell \in \{j', \ldots, j' + C\}$  forms a one-letter balanced pair in the reduction of the pair  $(\sigma^{k+N}(w), \sigma^{k+N}(w_\tau))$ .

Since by assumption the gaps between one-letter balanced pairs in  $(\sigma^{k+N}(w_{\tau}), \sigma^{k+N}(w_{\tau'\circ\tau}))$  are bounded by C, there is an index  $j'' \in \{j', \ldots, j' + C\}$  such that the j"th letters of  $\sigma^{k+N}(w_{\tau})$  and  $\sigma^{k+N}(w_{\tau'\circ\tau})$  coincide. However, as  $j'' \in \{j', \ldots, j' + C\}$  also the j"th letters of  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w_{\tau})$  coincide. Combining these two assertions we see that the j"th letters of  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w)$  and  $\sigma^{k+N}(w_{\tau'\circ\tau})$  coincide.

Since this argument goes through for every  $k \in \mathbb{N}$  and for every index j of a one-letter balanced pair in  $(\sigma^k(w), \sigma^k(w_\tau))$ , we deduce that the occurrences between successive one-letter balanced pairs in  $(\sigma^{k+N}(w), \sigma^{k+N}(w_{\tau'\circ\tau}))$  are bounded uniformly in k by  $(C+1)||\sigma^N||$  (recall that the width  $||\sigma||$  of  $\sigma$  is defined as  $||\sigma|| := \max_{a \in A} |\sigma(a)|$  (see Definition 1.2.20)). This establishes the induction step and the lemma is proved.

**Theorem 5.8.8** Let  $\sigma$  be a unit Pisot irreducible substitution. The balanced pair algorithm starting with  $I_0 = \{(ij, ji) \mid i, j \in A, i \neq j\}$  terminates if, and only if, the tiling property is satisfied.

Proof We first prove that if the balanced pair algorithm starting with  $I_0 = \{(ij, ji) \mid i, j \in A, i \neq j\}$  terminates, then  $\sigma$  satisfies the combinatorial strong coincidence condition. Let  $i, j \in A$ . According to Proposition 5.8.3, the distance between two one-letter balanced pairs in  $(\sigma^N(i)\sigma^N(j)), \sigma^N(j)\sigma^N(i))$  is uniformly bounded in N. As  $|\sigma^\ell(k)| \to \infty$  for  $\ell \to \infty$  for each  $k \in A$  this yields the existence of  $N \in \mathbb{N}$  such that  $\sigma^N(i)$  and  $\sigma^N(j)$  can be decomposed as  $\sigma^N(i) = par$  and  $\sigma^N(j) = qas$ , with  $\mathbf{P}(p) = \mathbf{P}(q)$ . Hence,  $\sigma$  satisfies the combinatorial strong coincidence condition.

We now assume that the balanced pair algorithm starting with  $I_0 = \{(ij, ji) \mid i, j \in A, i \neq j\}$  terminates. Let w be a non-empty prefix of u with  $u_{|w|} = u_0$ . Let  $\lambda = \pi_e \circ \mathbf{P}(w) \in \Xi(\mathcal{E}_u)$ . Lemma 5.8.7 implies that the balanced pair algorithm associated with w terminates, since the set  $I_0(w)$  is finite. Hence, by Corollary 5.8.5,  $\sigma$  satisfies the weak overlap coincidence condition.

Thus we proved that  $\sigma$  satisfies both the combinatorial strong coincidence condition and the weak overlap coincidence condition. Corollary 5.7.15 now implies that the tiling property is satisfied.

Let us now assume that  $\sigma$  satisfies the tiling property. This implies that both the strong overlap and the combinatorial strong coincidence condition hold. Let  $(ij, ji) \in I_0$ . Let  $T_i = \pi_e([0, i]_g), S_j = \pi_e([\mathbf{e}_i, j]_g), T_j = \pi_e([0, j]_g), S_i = \pi_e([\mathbf{e}_j, i]_g)$ . By construction  $T_i$  and  $S_j$  are adjacent intervals, as well as  $T_j, S_i$ , and they cover the same interval  $I := T_i \cup S_j = T_j \cup S_i$ . This interval can be decomposed as the union of three overlap subintervals  $O_1, O_2, O_3$  that do not necessarily belong to  $\mathcal{O}_u$ . Assume w.l.o.g. that  $T_i$  is longer than  $S_j$ . Then these intervals are equal to  $T_j, T_i \cap S_i$  and  $S_j$ . We deduce from Lemma 5.7.11 that there exists M such that  $\beta^M O_1, \beta^M O_2, \beta^M O_3$ can be decomposed into overlaps that belong to  $\mathcal{O}_u$ . Corollary 5.7.16 then implies that gaps between two coincidence overlaps in  $\beta^{k+M}O_1, \beta^{k+M}O_2, \beta^{k+M}O_3$  are uniformly bounded in k, which implies that gaps between oneletter balanced pairs ( $\sigma^k(ij), \sigma^k(ji)$ ) are also uniformly bounded in k. This implies that the balanced pair algorithm starting with  $I_0$  terminates.

One may define a graph associated with the balanced pair algorithm in an obvious way. The subgraph of this graph consisting of the coincidences (i, i) corresponds to a slightly modified prefix-suffix graph. If one removes this subgraph, and if the balanced pair algorithm ends, the remaining graph is finite, and its dominant eigenvalue corresponds to the asymptotic order of growth of non-coincidences between two different fixed points of powers of  $\sigma$ .

The balanced pair algorithm only terminates if the substitution satisfies the tiling property. Even if the super coincidence conjecture is true, it may well happen that the algorithm takes a long time before it terminates. We refer the reader to (Sirvent and Solomyak 2002, Section 6) where some examples are presented. They show that even quite simple looking substitutions may lead to quite large sets  $I_k$ . This suggests that it might be hard to get an analysis of the complexity of the balanced pair algorithm. Nevertheless, it is a useful and easy to implement tool for checking the tiling property for a given example with a purely combinatorial algorithm.

# 5.9 Conclusion

As a conclusion, let us summarise the different tiling conditions that we have encountered in the present chapter. The main condition can be formulated as follows: every two-piece patch of  $\Gamma_c$  appears (up to a translation vector) in the iterated image of a tip  $[0, i]^*$  with  $i \in A$ . Several variations around this idea have produced the following graphs.

- The two-piece ancestor graph traces back ancestors of patches under a generalised substitution that acts on tips that belong to a finite patch (the two-piece seed patch  $W_{\sigma}$ ) of the self-replicating translation set  $\Gamma_c$ . The tiling property is equivalent to the fact that from every vertex, there exists a path to a vertex with the specific shape  $[i, 0, i]^*$ , for  $i \in A$ .
- The boundary graph is a subset of the two-piece ancestor graph. It describes the tiles in the self-replicating multiple tiling that intersect the subtiles of the central tile. The tiling property can be expressed by computing the spectral radius associated with this graph.
- The *contact graph* is a subgraph of the two-piece ancestor graph that can be computed iteratively without the knowledge of the two-piece seed patch. The tiling property can also be expressed in terms of the spectral radius associated to this graph.
- The configuration graph is isomorphic to the two-piece ancestor graph. It checks whether every pair of basic strands eventually contains a common basic strand when applying  $\sigma$  iteratively.
- The graph of overlaps follows the same type of construction scheme as the the configuration graph. It is restricted to pairs of strands that are related to the fixed point of the substitution. It has to be combined

with the *combinatorial strong coincidence condition* to provide a tiling property.

• The *balanced pair algorithm* is a simple combinatorial process that describes the growth of gaps between coincidence overlaps and checks whether these gaps are uniformly bounded. It terminates whenever the tiling property is satisfied.

Among these conditions, the balanced pair algorithm is definitively the most combinatorial one. However, it does not always terminate. On the contrary, the contact graph can be computed with a purely algorithmic process that does not require additional computations, and provides a complete tiling characterisation. Note that the relations between the contact graph and the balanced pair algorithm remain unclear so far and deserve specific studies. Note also that the contact graph is not defined in the Pisot reducible case.

#### 5.10 Exercises

**Exercise 5.1** Let  $\beta$  be a Parry number (see Definition 2.3.12). The  $\beta$ -substitution  $\sigma_{\beta}$  associated with  $\beta$  is defined over the alphabet  $\{1, \dots, n\}$ , where n stands for the number of states of the automaton  $S_{\beta}$  defined in the proof of Proposition 2.3.18, as follows: j is the kth letter of  $\sigma_{\beta}(i)$  if, and only if, there is an arrow in  $S_{\beta}$  from the state i to the state j labelled by k - 1. Give the  $\beta$ -substitution  $\sigma_{\beta}$  explicitly with respect to  $\mathsf{d}_{\beta}(1)$ . Compare its prefix-suffix automaton with the automaton  $S_{\beta}$ . Prove that  $\sigma_{\beta}$  is primitive. Show that the Perron–Frobenius eigenvalue of its incidence matrix is equal to  $\beta$ . For more details, see (Lothaire 2002, Chapter 7).

**Exercise 5.2** Prove that the first coordinate projection of  $\Gamma_c$  on  $\mathbb{H}_c$  is not left invariant by any non-zero translation vector.

**Exercise 5.3** Prove that the fact that the self-replicating multiple tiling  $\mathcal{I}_{\sigma}$  is actually a tiling implies that the largest eigenvalue of the adjacency matrix of the boundary graph (as well as of the contact graph) is strictly smaller than the largest eigenvalue of  $\mathbf{M}_{\sigma}$ .

*Hint:* Look at the corresponding result for the contact graph of a self-similar lattice tiling in (Gröchenig and Haas 1994, Section 4).

**Exercise 5.4** Prove that if two basic strands have geometric strong coincidence, then they have the same height.

**Exercise 5.5** Prove that  $[\mathcal{O}_u]$  and  $[\mathcal{O}_u(\lambda)]$  do not depend on the choice of the fixed point u, but only on  $\sigma$ .

*Hint:* Use the repetitivity of  $\mathcal{E}_u$ , or equivalently, the uniform recurrence of the fixed point u.

**Exercise 5.6** Prove that the set of irreducible balanced pairs occurring by reducing  $(u, S^{|w|}(u))$  (here S denotes the shift) is a finite set. *Hint:* Use the uniform recurrence of the fixed point u.

**Exercise 5.7** List among the graphs introduced in the present chapter the ones that contain the prefix-suffix graph (or the graph obtained by reversing the direction of its edges) as a subgraph.

# 5.11 Notes

# Section 5.1

As introduced for instance in (Thurston 1989) and in (Fabre 1995), one can associate in a natural way with the  $\beta$ -shift (see Section 2.3.2.1) a substitution  $\sigma_{\beta}$  called  $\beta$ -substitution, in the case where  $\beta$  is a Parry number. For more details, see Exercise 5.1. Compare also with the ideas underlying Definition 3.4.10. If  $\beta$  is a Pisot number, the associated substitution can be Pisot reducible as well as Pisot irreducible. An example of a Pisot reducible  $\beta$ -substitution is given by the smallest Pisot number  $\beta$  which is the positive root of  $X^3 - X - 1$  (see Example 2.3.54).

Fractal geometry is deeply related to the study of numeration systems. One of the most famous examples of fractal tiles that come from numeration systems is the twin dragon fractal related to expansions of Gaussian integers in base -1 + i (see (Knuth 1998, p. 206)). More generally, tilings can be introduced in the framework of canonical numeration systems (see Section 2.4 and the references in the survey (Akiyama and Thuswaldner 2004)), of shift radix systems (see Section 2.4.4 and (Berthé, Siegel, Steiner, et al. 2009)), or of abstract numeration systems (see Chapter 3 and (Berthé and Rigo 2007a)). In particular, the study of the boundary of central tiles has proved to be particularly efficient in order to derive properties of numeration systems. Under the tiling condition and in the cubic case n = 3, points lying at the intersection of tiles in the self-replicating tiling have been described as complex numbers with multiple expansions in some numeration system (see e.g. (Messaoudi 1998, Messaoudi 2000, Sadahiro 2006)). For more on the relations between central tiles and numeration systems, see the survey (Barat, Berthé, Liardet, et al. 2006).

The construction of central tiles also has consequences for the effective construction of Markov partitions for toral automorphisms, the main eigenvalue of which is a Pisot number. See, for instance, (Kenyon and Vershik 1998), (Praggastis 1999), (Schmidt 2000), and (Lindenstrauss and Schmidt 2005). For more on connections between beta-numeration, Vershik's adic transformation (see (Vershik and Livshits 1992)) and codings of hyperbolic automorphisms, see the survey (Sidorov 2003), and in the same vein, (Einsiedler and Schmidt 2002).

The study of central tiles has also led to particularly interesting applications in number theory. This was one of the motiva-Central tiles and their associated tilings tions of (Rauzy 1982). are indeed efficient tools to compute best simultaneous Diophantine approximations (see (Chekhova, Hubert, and Messaoudi 2001), (Hubert and Messaoudi 2006) and (Ito, Fujii, Higashino, et al. 2003)), or to characterise points with purely periodic beta-expansions (see (Hama and Imahashi 1997), (Akiyama, Barat, Berthé, et al. 2008) or (Adamczewski, Frougny, Siegel, et al. 2010)).

### Section 5.2

In the case of a unit Pisot reducible substitution, besides  $\mathbb{H}_c$  and  $\mathbb{H}_e$  a third space plays a role. This is the space  $\mathbb{H}_s$  generated by the eigenspaces corresponding to the eigenvalues of  $\mathbb{M}_{\sigma}$  that are not conjugate to  $\beta$ . The projection of the broken line  $L_u$  on  $\mathbb{H}_c$  along  $\mathbb{H}_e \oplus \mathbb{H}_s$  still provides a bounded set in  $\mathbb{H}_c$  which allows the definition of a central tile also in this case (for more details, see (Ei, Ito, and Rao 2006, Section 3.2) and (Berthé and Siegel 2005)).

The study of the spectrum of Pisot substitutive dynamical systems was one of the main motivations for the introduction of central tiles. Pisot irreducible substitutions are indeed conjectured to have discrete spectrum. For a detailed account of the spectral theory of substitutive dynamical systems, see (Queffélec 1987), (Pytheas Fogg 2002, Chapter 7) and (Barge and Kwapisz 2006). See also Section 6.9.

The notion of *coincidence* (in its various forms) has proved to be an efficient way for proving discrete spectrum. The coincidence condition was first introduced by Dekking (Dekking 1978) for substitutions with constant length. Using this notion, Dekking completely characterised the substitutions with constant length whose associated symbolic dynamical system has discrete spectrum. Later, Arnoux and Ito introduced the notion of combinatorial strong coincidence (under the name *strong coincidence*) in

### V. Berthé, A. Siegel, J. Thuswaldner

(Arnoux and Ito 2001), which lead Ito and Rao to define the super coincidence condition (under the name super coincidence) in (Ito and Rao 2006). The super coincidence condition has been also introduced independently in (Barge and Kwapisz 2006) (under the name geometric coincidence condition). For a complete proof of the equivalence between discrete spectrum and the super coincidence condition for unit Pisot irreducible substitutions, see (Barge and Kwapisz 2006). The notion of coincidence has also been exploited in the framework of substitution tiling spaces and substitution Delone multisets. Lee (see (Lee 2007)) introduced the notion of algebraic coincidence in order to characterise substitution Delone multisets that have a pure point diffraction spectrum. Using the notation of Section 5.7 algebraic coincidence can be stated as follows. Let  $\Lambda_i := \bigcup_{n \in \mathbb{N}, u_n=i} T_n$ . We say that the substitution  $\sigma$  satisfies the algebraic coincidence condition if there exist a positive integer M and  $\xi \in \Lambda_i$ for each  $i \in A$  such that  $\xi + \beta^M \Xi(\mathcal{E}_u) \subseteq \Lambda_i$ . For a review of various notions of coincidences that are related to substitutions and substitution Delone multisets, see (Sing 2006) and the discussion in (Lee 2007). See also (Lee, Moody, and Solomyak 2003) and (Fretlöh and Sing 2007) for the related notion of modular coincidence.

The numeration system based on words alluded to in the proof of Theorem 5.2.3 is known as the Dumont-Thomas numeration system (see e.g. (Dumont and Thomas 1989, Rauzy 1990, Dumont and Thomas 1993) and Section 9.4.2). More precisely one checks that every finite prefix of ucan be uniquely expanded as  $\sigma^n(p_n)\sigma^{n-1}(p_{n-1})\cdots p_0$ , where  $p_n \neq \varepsilon$ , and  $(p_0, a_0, s_0) \cdots (p_n, a_n, s_n)$  is the sequence of labels of a path in the prefix-suffix automaton  $\mathcal{G}_{\sigma}$  (see (Dumont and Thomas 1989, Theorem 1.5)). Hence, we can expand the non-negative integer N as  $N = |\sigma^n(p_n)| + \cdots +$  $|p_0|$ , where  $u_0 \cdots u_{N-1} = \sigma^n(p_n) \sigma^{n-1}(p_{n-1}) \cdots p_0$ . Let  $\beta$  be the Perron-Frobenius eigenvalue of  $\sigma$ . This numeration system also provides generalised radix expansions of positive real numbers, with digits belonging to a finite subset of the number field  $\mathbb{Q}(\beta)$ . We first define the mapping  $\delta_{\sigma}: A^* \to \mathbb{Q}(\beta), p \mapsto \langle \mathbf{P}(p), \mathbf{w}_{\beta} \rangle$ , where  $\mathbf{w}_{\beta}$  is a left eigenvector of  $\mathbf{M}_{\sigma}$ with positive entries associated with the Perron–Frobenius eigenvalue  $\beta$ . One has  $\delta_{\sigma}(\sigma^n(p)) = \beta^n \delta_{\sigma}(p)$ , for every n and  $p \in A^*$ . We then associate with the combinatorial expansion  $(p_n, a_n, s_n) \dots (p_0, a_0, a_0)$  the real number  $\delta_{\sigma}(p_n)\beta^n + \cdots + \delta_{\sigma}(p_0) \in \mathbb{Q}(\beta)$ . To recover the  $\beta$ -numeration in the particular case where  $\sigma$  is a  $\beta$ -substitution,  $\mathbf{w}_{\beta}$  has to be normalised so that its first coordinate is equal to 1: the coordinates of  $\mathbf{w}_{\beta}$  are then of the form  $T^i_{\beta}(1)$ , for  $0 \le i \le n-1$ , with  $T_{\beta} \colon x \mapsto \{\beta x\}$ . We have chosen to work in the present chapter with an eigenvector  $\mathbf{v}_{\beta}$  normalised so that its coordinates belong to  $\mathbb{Z}[\beta]$ . This choice of normalisation plays a role in particular in the
proof of Lemma 5.7.2. One checks that if  $\sigma$  is a unit Pisot irreducible  $\beta$ substitution, the theory and the results of this chapter also hold by working with  $\mathbf{w}_{\beta}$  normalised in a way that its first coordinate is equal to 1, instead of working with  $\mathbf{v}_{\beta}$ . In particular, the proof of Lemma 5.7.2 can easily be adapted, by noticing that there exists a positive integer D > 0 such that the coordinates of  $\mathbf{w}_{\beta}$  all belong to  $\frac{1}{D}\mathbb{Z}[\beta]$ .

### Section 5.3

GIFS substitutions, introduced in (Arnoux and Ito 2001), were inspired by the geometric formalism of (Ito and Ohtsuki 1993), whose aim was to provide explicit Markov partitions for hyperbolic automorphisms of the torus associated with particular substitutions produced by Brun's continued fraction algorithm. GIFS substitutions have already proved their efficiency for Diophantine approximation (Ito, Fujii, Higashino, et al. 2003), in word combinatorics (Arnoux, Berthé, and Siegel 2004), and in discrete geometry (Arnoux, Berthé, and Ito 2002), (Fernique 2006) and (Fernique 2009).

There is a second multiple tiling defined on  $\mathbb{H}_c$  that plays an important role in the study of the substitutive symbolic dynamical system  $(X_{\sigma}, S)$ . It is obtained by projecting points in  $\mathbb{Z}^n$  that lie on the hyperplane with equation  $\langle \mathbf{x}, (1, \ldots, 1) \rangle = 0$  by  $\pi_c$ . The corresponding translation set, called the lattice translation set, is thus defined as  $\{[\gamma, i]^* \in \pi_c(\mathbb{Z}^n) \times A \mid \gamma \in \sum_{k=2}^n \mathbb{Z}(\pi_c(\mathbf{e}_k) - \pi_c(\mathbf{e}_1))\}$ . It is clearly periodic. According to (Canterini and Siegel 2001b), if  $\sigma$  is a unit Pisot irreducible substitution that satisfies the combinatorial strong coincidence condition, the lattice translation set is a Delone set that also provides a multiple tiling for the subtiles of the central tile. This multiple tiling is called the *lattice multiple tiling*. Rauzy introduced in the seminal paper (Rauzy 1982) the notion of central tile with respect to this tiling. According to (Ito and Rao 2006) (see also (Barge and Kwapisz 2006, Remark 18.5)), we know that the lattice multiple tiling is a tiling if, and only if, the self-replicating multiple tiling is a tiling, if  $\sigma$  is assumed to be a unit Pisot irreducible substitution.

There are two dynamical systems that can be associated in a natural way with a unit Pisot substitution, namely the substitutive dynamical system  $(X_{\sigma}, S)$  (with its natural Z-action by the shift), and the one-dimensional tiling space associated with the self-similar tiling of the expanding line (described in terms of an R-action by translations). The lattice multiple tiling is intimately connected to the spectral properties of the substitutive dynamical system  $(X_{\sigma}, S)$  (see (Queffélec 1987) and (Pytheas Fogg 2002, Chapter 7)), whereas the self-replicating multiple tiling is connected to the spectral properties of the one-dimensional tiling space associated with the tiling of the expanding line (see (Barge and Kwapisz 2006)). Note that there exist unit Pisot reducible substitutions for which  $(X_{\sigma}, S)$  does not have discrete spectrum, as shown by (Baker, Barge, and Kwapisz 2006, Example 5.3). See also (Clark and Sadun 2006) for the study of the spectral impact of deformations of the lengths of tiles for the tiling spaces associated with a substitution. More generally, see (Sadun 2008) for a topological study of tiling spaces with aperiodic order.

## Section 5.4

The geometric finiteness property is intimately related to the so-called (F) property (introduced in Section 2.3.2.2 in the beta-numeration framework). It is expressed in (Berthé and Siegel 2005) in terms of the Dumont–Thomas numeration. It also appears in (Fuchs and Tijdeman 2006) in a related context. Note that we can use the vast literature on the (F) property in the beta-numeration framework to exhibit classes of beta-substitutions that satisfy the geometric finiteness property (see *e.g.* (Baker, Barge, and Kwapisz 2006) and (Barge and Kwapisz 2005)).

The so-called (W) or *weak finiteness* property was first introduced in (Hollander 1996). He has proved that the (W) property implies the pure discreteness of the spectrum of the irreducible beta-shift. The (W) property can be stated for a Pisot number  $\beta$  as follows:

 $\forall z \in \mathbb{Z}[\beta^{-1}] \cap [0,1), \ \forall \varepsilon > 0, \ \exists x, y \in \operatorname{Fin}(\beta) \text{ such that } z = x - y \text{ and } y < \varepsilon.$ 

The (W) property has been proved in (Akiyama 2002) to be equivalent with the tiling property. An algorithm which can tell whether a given Pisot number  $\beta$  has (F) or (W) property is described in (Akiyama, Rao, and Steiner 2004). The condition of Theorem 5.4.14 is related to the (W) property.

### Section 5.5

Similar graphs have appeared in several restricted contexts with different names (see *e.g.* the references in (Akiyama and Thuswaldner 2004) for contact graphs for tiles related to matrix number systems). They are used either to describe beta-expansions for 0 (Akiyama 2002), to describe multiple expansions (Messaoudi 1998, Messaoudi 2000, Durand and Messaoudi 2009), to compute the Hausdorff dimension of the boundary of central tiles (Messaoudi 2000, Feng, Furukado, Ito, et al. 2006, Thuswaldner 2006), or to obtain pure discrete spectrum conditions for substitutive dynamical systems (by referring to the lattice translation set) (Siegel 2004). The knowledge on intersections between tiles also yields criteria for topological properties of central tiles (connectivity, disklikeness, non-trivial fundamental group) (see (Messaoudi 2000), (Siegel 2004), (Siegel and Thuswaldner 2010)).

Contact graphs are inspired by the contact matrix defined in (Gröchenig and Haas 1994) for self-similar lattice tilings. They have been defined in (Thuswaldner 2006) in the framework of substitutions. The properties of the contact graph are based on the polyhedral tiling generated by the geometric tips. This polyhedral tiling property is very specific to the Pisot irreducible case. This is the main reason why the contact graph can only be defined in the Pisot irreducible case, whereas the two-piece ancestor and the boundary graphs can be defined in the Pisot reducible case with slight modifications (see (Siegel and Thuswaldner 2010)). In the Pisot reducible case, some examples of substitutions have been studied in (Ei and Ito 2005) by using the *m*th polyhedral approximations of Definition 5.5.9. Unfortunately, no generic algorithm based on this approach exists so far.

#### Section 5.6

The notion of strand, introduced in (Barge and Diamond 2002), has been very fruitfully developed in the form of the strand space model for one-dimensional substitution tiling spaces in (Barge and Kwapisz 2006), see also (Barge and Kwapisz 2005), (Barge and Diamond 2007) and (Barge, Diamond, and Swanson 2009).

### Section 5.7

Lemma 5.7.10 is strongly related to the notion of algebraic coincidence (see (Lee 2007)).

Much more than Lemma 5.7.11 can be said. In fact, the  $\mathbb{Z}$ -module generated by the lengths  $\pi_e \circ \mathbf{P}(i)$ , for  $i \in A$ , is equal to  $\Xi(\mathcal{E}_u)$ . For a proof, see (Sing 2006, Lemma 6.34) and (Barge and Kwapisz 2006, Section 12). Furthermore, for examples of substitutions for which we have the strict inclusion  $\Xi(\mathcal{E}_u) \subset \mathbb{Z}[\beta]$ , see (Sing 2006, Remark 6.36).

In (Sirvent and Solomyak 2002) the spectrum of the two dynamical systems associated with a substitution of Pisot type (*i.e.*, the substitutive symbolic space  $(X_{\sigma}, S)$  and the one-dimensional tiling space), is studied by comparing the balanced pair algorithm (for the Z-action) and the overlap

# V. Berthé, A. Siegel, J. Thuswaldner

algorithm (for the  $\mathbb{R}$ -action). It is proved in (Clark and Sadun 2006) and in (Barge and Kwapisz 2006) that for a unit Pisot irreducible substitution, the tiling space has discrete spectrum if, and only if, the substitutive symbolic dynamical system has discrete spectrum.

# Section 5.8

In Section 5.8 we have dealt exclusively with Pisot irreducible substitutions. Recently, (Martensen 2004) has generalised the balanced pair algorithm to the Pisot reducible case. In this case one has to identify certain patterns in order to get a proper behaviour of the algorithm, *i.e.*, to show its termination to be equivalent to the fact that the dynamical system associated with the substitution in question is purely discrete.

#### References

- [Adamczewski, Frougny, Siegel, et al. 2010] Adamczewski, B., Frougny, C., Siegel, A., and Steiner, W. Rational numbers with purely periodic beta-expansion. J. London Math. Soc. To appear.
- [Akiyama 1999] Akiyama, S. Self affine tiling and Pisot numeration system. In S. Kanemitsu and K. Györy, eds., Number Theory and Its Applications, pp. 7–17. Kluwer, 1999.
- [Akiyama 2000] Akiyama, S. Cubic Pisot units with finite beta expansions. In F. Halter-Koch and R. F. Tichy, eds., Algebraic Number Theory and Diophantine Analysis, pp. 11–26. Walter de Gruyter, 2000.
- [Akiyama 2002] Akiyama, S. On the boundary of self affine tilings generated by Pisot numbers. J. Math. Soc Japan 54(2), (2002) 283–308.
- [Akiyama, Barat, Berthé, et al. 2008] Akiyama, S., Barat, G., Berthé, and Siegel, A. Boundary of central tiles associated with Pisot beta-numeration and purely periodic expansions. *Monatsh. Math.* **155**, (2008) 377–419.
- [Akiyama, Rao, and Steiner 2004] Akiyama, S., Rao, H., and Steiner, W. A certain finiteness property of Pisot number systems. J. Number Theory 107, (2004) 135–160.
- [Akiyama and Thuswaldner 2004] Akiyama, S. and Thuswaldner, J. M. A survey on topological properties of tiles related to number systems. *Geometriae Dedicata* 109, (2004) 89–105.
- [Arnoux, Berthé, and Ito 2002] Arnoux, P., Berthé, V., and Ito, S. Discrete planes, Z<sup>2</sup>-actions, Jacobi-Perron algorithm and substitutions. Ann. Inst. Fourier (Grenoble) 52(2), (2002) 305–349.
- [Arnoux, Berthé, and Siegel 2004] Arnoux, P., Berthé, V., and Siegel, A. Twodimensional iterated morphisms and discrete planes. *Theoret. Comput. Sci.* 319, (2004) 145–176.
- [Arnoux and Ito 2001] Arnoux, P. and Ito, S. Pisot substitutions and Rauzy fractals. Bull. Belg. Math. Soc. 8, (2001) 181–207.
- [Baker, Barge, and Kwapisz 2006] Baker, V., Barge, M., and Kwapisz, J. Geometric realization and coincidence for reducible non-unimodular Pisot tiling spaces with an application to beta-shifts. Ann. Inst. Fourier (Grenoble) 56(7), (2006) 2213–2248.
- [Barat, Berthé, Liardet, et al. 2006] Barat, G., Berthé, V., Liardet, P., and Thuswaldner, J. Dynamical directions in numeration. Ann. Inst. Fourier (Grenoble) 56, (2006) 1987–2092.
- [Barge and Diamond 2002] Barge, M. and Diamond, B. Coincidence for substitutions of Pisot type. Bull. Soc. Math. France 130, (2002) 619–626.
- [Barge and Diamond 2007] Barge, M. and Diamond, B. Proximality in Pisot tiling spaces. Fundamenta Math. 194(3), (2007) 191–238.
- [Barge, Diamond, and Swanson 2009] Barge, M., Diamond, B., and Swanson, R. The branch locus for one-dimensional Pisot tiling spaces. *Fundamenta Math.* 204(3), (2009) 215–240.
- [Barge and Kwapisz 2005] Barge, M. and Kwapisz, J. Elements of the theory of unimodular Pisot substitutions with an application to  $\beta$ -shifts. In Algebraic and topological dynamics, vol. 385 of Contemporary Mathematics, pp. 89–99. Amer. Math. Soc., 2005.
- [Barge and Kwapisz 2006] Barge, M. and Kwapisz, J. Geometric theory of unimodular Pisot substitution. Amer. J. Math. 128, (2006) 1219–1282.
- [Berthé and Rigo 2007] Berthé, V. and Rigo, M. Abstract numeration systems and tilings. In Proc. 32nd Symposium, Mathematical Foundations of Computer Science 2007, vol. 3618 of Lecture Notes in Computer Science, pp. 131–143.
  - 79

Springer-Verlag, 2007.

- [Berthé and Siegel 2005] Berthé, V. and Siegel, A. Tilings associated with beta-numeration and substitutions. *Integers* 5 (2005), #A02 (electronic), http://www.integers-ejcnt.org/vol5.html
- [Berthé, Siegel, Steiner, et al. 2009] Berthé, V., Siegel, A., Steiner, W., Surer, P., and Thuswaldner, J. Fractal tiles associated with shift radix systems, 2009. Preprint.
- [Berthé and Vuillon 2000] Berthé, V. and Vuillon, L. Tilings and rotations on the torus: a two-dimensional generalization of Sturmian sequences. *Discrete* Math. 223, (2000) 27–53.
- [Canterini and Siegel 2001a] Canterini, V. and Siegel, A. Automate des préfixessuffixes associé à une substitution primitive. J. Théorie Nombres Bordeaux 13, (2001) 353–369.
- [Canterini and Siegel 2001b] Canterini, V. and Siegel, A. Geometric representation of substitutions of Pisot type. Trans. Amer. Math. Soc. 353, (2001) 5121– 5144.
- [Chekhova, Hubert, and Messaoudi 2001] Chekhova, N., Hubert, P., and Messaoudi, A. Propriétés combinatoires, ergodiques et arithmétiques de la substitution de Tribonacci. J. Théorie Nombres Bordeaux 13, (2001) 371–394.
- [Clark and Sadun 2006] Clark, A. and Sadun, L. When shape matters: deformations of tiling spaces. Ergod. Th. & Dynam. Sys. 26, (2006) 69–86.
- [Dekking 1978] Dekking, F. M. The spectrum of dynamical systems arising from substitutions of constant length. Z. Wahrscheinlichkeitstheorie und verw. Gebiete 41, (1978) 221–239.
- [Dumont and Thomas 1989] Dumont, J.-M. and Thomas, A. Systèmes de numération et fonctions fractales relatifs aux substitutions. *Theoret. Comput. Sci.* 65, (1989) 153–169.
- [Dumont and Thomas 1993] Dumont, J.-M. and Thomas, A. Digital sum moments and substitutions. Acta Arith. 64, (1993) 205–225.
- [Durand and Messaoudi 2009] Durand, F. and Messaoudi, A. Boundary of the Rauzy fractal sets in  $\mathbb{R} \times \mathbb{C}$  generated by  $P(x) = x^4 x^3 x^2 x 1$ . Osaka J. Math. To appear.
- [Ei and Ito 2005] Ei, H. and Ito, S. Tilings from some non-irreducible Pisot substitutions. Discrete Math. & Theoret. Comput. Sci. 7, (2005) 81–122.
- [Ei, Ito, and Rao 2006] Ei, H., Ito, S., and Rao, H. Atomic surfaces, tilings and coincidences II. Reducible case. Ann. Inst. Fourier (Grenoble) 56, (2006) 2285–2313.
- [Einsiedler and Schmidt 2002] Einsiedler, M. and Schmidt, K. Irreducibility, homoclinic points and adjoint actions of algebraic Z<sup>d</sup>-actions of rank one. In Dynamics and randomness (Santiago, 2000), vol. 7 of Nonlinear Phenom. Complex Systems, pp. 95–124. Kluwer, Dordrecht, 2002.
- [Fabre 1995] Fabre, S. Substitutions et β-systèmes de numération. Theoret. Comput. Sci. 137, (1995) 219–236.
- [Feng, Furukado, Ito, et al. 2006] Feng, D.-J., Furukado, M., Ito, S., and Wu, J. Pisot substitutions and the Hausdorff dimension of boundaries of atomic surfaces. *Tsukuba J. Math.* **30**, (2006) 195–223.
- [Fernique 2006] Fernique, T. Multidimensional Sturmian sequences and generalized substitutions. Int. J. Found. Comput. Sci. 17, (2006) 575–600.
- [Fernique 2009] Fernique, T. Generation and recognition of digital planes using multi-dimensional continued fractions. *Pattern Recognition* 42, (2009) 2229– 2238.
- [Fretlöh and Sing 2007] Fretlöh, D. and Sing, B. Computing modular coincidences

for substitution tilings and point sets. Discrete and Computational Geometry **37**, (2007) 381–401.

- [Fuchs and Tijdeman 2006] Fuchs, C. and Tijdeman, R. Substitutions, abstract number systems and the space filling property. Ann. Inst. Fourier (Grenoble) 56(7), (2006) 2345–2389. Numération, pavages, substitutions.
- [Gröchenig and Haas 1994] Gröchenig, K. and Haas, A. Self-similar Lattice Tilings. J. Fourier Anal. Appl. 1, (1994) 131–170.
- [Hama and Imahashi 1997] Hama, M. and Imahashi, T. Periodic β-expansions for certain classes of Pisot numbers. Comment. Math. Univ. St. Paul. 46(2), (1997) 103–116.
- [Hardy and Wright 1985] Hardy, G. H. and Wright, E. M. An Introduction to the Theory of Numbers. Oxford University Press, 5th edn., 1985.
- [Hollander 1996] Hollander, M. Linear Numeration systems, Finite Beta Expansions, and Discrete Spectrum of Substitution Dynamical Systems. Ph.D. thesis, University of Washington, 1996.
- [Hollander and Solomyak 2003] Hollander, M. and Solomyak, B. Two-symbol Pisot substitutions have pure discrete spectrum. *Ergod. Th. & Dynam. Sys.* **23**(2), (2003) 533–540.
- [Hubert and Messaoudi 2006] Hubert, P. and Messaoudi, A. Best simultaneous Diophantine approximations of Pisot numbers and Rauzy fractals. Acta Arith. 124(1), (2006) 1–15.
- [Ito, Fujii, Higashino, et al. 2003] Ito, S., Fujii, J., Higashino, H., and Yasutomi, S.-I. On simultaneous approximation to  $(\alpha, \alpha^2)$  with  $\alpha^3 + k\alpha 1 = 0$ . J. Number Theory **99**(2), (2003) 255–283.
- [Ito and Ohtsuki 1993] Ito, S. and Ohtsuki, M. Modified Jacobi-Perron algorithm and generating Markov partitions for special hyperbolic toral automorphisms. *Tokyo J. Math.* 16(2), (1993) 441–472.
- [Ito and Rao 2006] Ito, S. and Rao, H. Atomic surfaces, tilings and coincidences I. Irreducible case. Israel J. Math. 153, (2006) 129–156.
- [Kalle and Steiner 2009] Kalle, C. and Steiner, W. Beta-expansions, natural extensions and multiple tilings, 2009. Preprint.
- [Kenyon and Vershik 1998] Kenyon, R. and Vershik, A. Arithmetic construction of sofic partitions of hyperbolic toral automorphisms. *Ergod. Th. & Dynam.* Sys. 18(2), (1998) 357–372.
- [Knuth 1998] Knuth, D. E. The Art of Computer Programming. Volume 2: Seminumerical Algorithms. Addison-Wesley, 1998. 3rd edition.
- [Lagarias and Pleasants 2002] Lagarias, J. C. and Pleasants, P. A. B. Local complexity of Delone sets and crystallinity. *Canad. Math. Bull.* 45(4), (2002) 634–652.
- [Lagarias and Pleasants 2003] Lagarias, J. C. and Pleasants, P. A. B. Repetitive Delone sets and quasicrystals. *Ergod. Th. & Dynam. Sys.* 23(3), (2003) 831– 867.
- [Lee 2007] Lee, J.-Y. Substitution Delone sets with pure point spectrum are intermodel sets. J. Geom. Phys. 57(11), (2007) 2263–2285.
- [Lee, Moody, and Solomyak 2003] Lee, J.-Y., Moody, R. V., and Solomyak, B. Consequences of pure point diffraction spectra for multiset substitution systems. Discrete and Computational Geometry 29, (2003) 525–560.
- [Lindenstrauss and Schmidt 2005] Lindenstrauss, E. and Schmidt, K. Symbolic representations of nonexpansive group automorphisms. Israel J. Math. 149, (2005) 227–266.
- [Livshits 1987] Livshits, A. N. On the spectra of adic transformations of Markov compact sets. Uspekhi. Mat. Nauk 42(3(255)), (1987) 189–190. In Russian. English translation in Russian Math. Surveys 42 (1987), 222-223.

#### References

- [Lothaire 2002] Lothaire, M. Algebraic Combinatorics on Words, vol. 90 of Encyclopedia of Mathematics and Its Applications. Cambridge University Press, 2002.
- [Martensen 2004] Martensen, B. F. Generalized balanced pair algorithm. Topology Proc. 28(1), (2004) 163–178. Spring Topology and Dynamical Systems Conference.
- [Mauldin and Williams 1988] Mauldin, R. D. and Williams, S. C. Hausdorff dimension in graph directed constructions. *Trans. Amer. Math. Soc.* 309(2), (1988) 811–829.
- [Messaoudi 1998] Messaoudi, A. Propriétés arithmétiques et dynamiques du fractal de Rauzy. J. Théorie Nombres Bordeaux 10, (1998) 135–162.
- [Messaoudi 2000] Messaoudi, A. Frontière du fractal de Rauzy et systèmes de numération complexes. Acta Arith. 95, (2000) 195–224.
- [Michel 1978] Michel, P. Coincidence values and spectra of substitutions. Z. Wahrscheinlichkeitstheorie und verw. Gebiete **42**(3), (1978) 205–227.
- [Moody 1997] Moody, R. V. Meyer sets and their duals. In R. V. Moody, ed., The Mathematics of Long-Range Aperiodic Order, vol. 489 of NATO ASI Ser., Ser. C., Math. Phys. Sci., pp. 403–441. Kluwer, 1997.
- [Praggastis 1999] Praggastis, B. Numeration systems and Markov partitions from self-similar tilings. Trans. Amer. Math. Soc. 351, (1999) 3315–3349.
- [Pytheas Fogg 2002] Pytheas Fogg, N. Substitutions in Dynamics, Arithmetics and Combinatorics, vol. 1794 of Lecture Notes in Mathematics. Springer-Verlag, 2002. Ed. by V. Berthé and S. Ferenczi and C. Mauduit and A. Siegel.
- [Queffélec 1987] Queffélec, M. Substitution Dynamical Systems Spectral Analysis, vol. 1294 of Lecture Notes in Mathematics. Springer-Verlag, 1987.
- [Rauzy 1982] Rauzy, G. Nombres algébriques et substitutions. Bull. Soc. Math. France 110, (1982) 147–178.
- [Rauzy 1990] Rauzy, G. Sequences defined by iterated morphisms. In R. M. Capocelli, ed., Sequences: Combinatorics, Compression, Security, and Transmission, pp. 275–287. Springer-Verlag, 1990.
- [Reveillès 1991] Reveillès, J.-P. Géométrie discrète, calcul en nombres entiers et algorithmique. Thèse de Doctorat, Université Louis Pasteur, Strasbourg, 1991.
- [Robinson 2004] Robinson, E. A., Jr. Symbolic dynamics and tilings of  $\mathbb{R}^d$ . In Symbolic dynamics and its applications, vol. 60 of Proc. Sympos. Appl. Math., Amer. Math. Soc. Providence, RI, pp. 81–119. 2004.
- [Sadahiro 2006] Sadahiro, T. Multiple points of tilings associated with Pisot numeration systems. *Theoret. Comput. Sci.* 359(1-3), (2006) 133–147.
- [Sadun 2008] Sadun, L. Topology of tiling spaces, vol. 46 of University Lecture Series. Amer. Math. Soc., Providence, RI, 2008.
- [Sano, Arnoux, and Ito 2001] Sano, Y., Arnoux, P., and Ito, S. Higher dimensional extensions of substitutions and their dual maps. J. Anal. Math. 83, (2001) 183–206.
- [Schmidt 2000] Schmidt, K. Algebraic coding of expansive group automorphisms and two-sided beta-shifts. *Monatsh. Math.* **129**(1), (2000) 37–61.
- [Sidorov 2003] Sidorov, N. Arithmetic dynamics. In S. B. et al., ed., Topics in dynamics and ergodic theory, vol. 310 of London Math. Soc. Lect. Note Ser., pp. 145–189. Cambridge University Press, 2003.
- [Siegel 2004] Siegel, A. Pure discrete spectrum dynamical system and periodic tiling associated with a substitution. AIFG **54**(2), (2004) 288–299.
- [Siegel and Thuswaldner 2010] Siegel, A. and Thuswaldner, J. Topological properties of Rauzy fractals. *Mémoire de la SMF*. To appear.

- [Sing 2006] Sing, B. *Pisot substitutions and beyond*. Ph.D. thesis, Universität Bielefeld, 2006.
- [Sirvent and Solomyak 2002] Sirvent, V. F. and Solomyak, B. Pure discrete spectrum for one-dimensional substitution systems of Pisot type. *Canad. Math. Bull.* 45(4), (2002) 697–710. Dedicated to Robert V. Moody.
- [Sirvent and Wang 2002] Sirvent, V. F. and Wang, Y. Self-affine tiling via substitution dynamical systems and Rauzy fractals. *Pacific J. Math.* 206(2), (2002) 465–485.
- [Solomyak 1997] Solomyak, B. Dynamics of self-similar tilings. Ergod. Th. & Dynam. Sys. 17, (1997) 695–738.
- [Thurston 1989] Thurston, W. Groups, Tilings and Finite State Automata, 1989. AMS Colloquium Lecture Notes.
- [Thuswaldner 2006] Thuswaldner, J. M. Unimodular Pisot substitutions and their associated tiles. J. Théorie Nombres Bordeaux 18(2), (2006) 487–536.
- [Vershik and Livshits 1992] Vershik, A. M. and Livshits, A. N. Adic models of ergodic transformations, spectral theory, substitutions, and related topics. In A. M. Vershik, ed., *Representation Theory and Dynamical Systems*, vol. 9 of *Advances in Soviet Mathematics*, pp. 185–204. Amer. Math. Soc., 1992.